

pubs.acs.org/CR

Quantum Information and Algorithms for Correlated Quantum Matter

Kade Head-Marsden,^{*,§} Johannes Flick,^{*,§} Christopher J. Ciccarino, and Prineha Narang*

Cite This: https://dx.doi.org/10.1021/acs.chemrev.0c00620		Read Online		
ACCESS	III Metrics & More		E Article Recommendations	

ABSTRACT: Discoveries in quantum materials, which are characterized by the strongly quantum-mechanical nature of electrons and atoms, have revealed exotic properties that arise from correlations. It is the promise of quantum materials for quantum information science superimposed with the potential of new computational quantum algorithms to discover new quantum materials that inspires this Review. We anticipate that quantum materials to be discovered and developed in the next years will transform the areas of quantum information processing including communication, storage, and computing. Simultaneously, efforts toward developing new quantum algorithmic approaches for quantum simulation and advanced calculation methods for many-body quantum systems enable major advances toward functional quantum materials and their deployment. The advent of quantum computing brings new possibilities for eliminating the exponential complexity that has stymied simulation of correlated quantum systems on high-performance classical computers. Here, we review new algorithms and computational approaches to predict and understand the behavior of correlated quantum matter. The strongly



interdisciplinary nature of the topics covered necessitates a common language to integrate ideas from these fields. We aim to provide this common language while weaving together fields across electronic structure theory, quantum electrodynamics, algorithm design, and open quantum systems. Our Review is timely in presenting the state-of-the-art in the field toward algorithms with nonexponential complexity for correlated quantum matter with applications in grand-challenge problems. Looking to the future, at the intersection of quantum information science and algorithms for correlated quantum matter, we envision seminal advances in predicting many-body quantum states and describing excitonic quantum matter and large-scale entangled states, a better understanding of high-temperature superconductivity, and quantifying open quantum system dynamics.

CONTENTS

1. Introduction	В
1.1. Layout and Sections	D
2. Wave Function-Based Electronic Structure Meth-	
ods for Quantum Matter	E
2.1. Introduction to Electronic Correlation Effects	E
2.2. Quantum Chemistry Approaches to Corre-	
lated Quantum Matter	F
2.2.1. Predicting Molecular Qubits	Н
3. Density-Based Electronic Structure to Predict	
Quantum Matter	I
3.1. Reduced Density Matrix Methods for Corre-	
lated Quantum Matter	I
3.2. Density-Functional Theory Approaches to	
Compute Quantum Materials	L
3.2.1. Predicting Artificial Atom Qubits in	
Solids	M
3.3. Time-Dependent Density-Functional Theory	
and Quasiparticle Methods for Correlated	

3.4. Light–Matter Control of Correlations in	
Quantum Materials	Р
3.4.1. New Directions in QED Control of	
Correlated Quantum Materials	S
3.5. Emerging Directions in Descriptions of	
Correlated Quantum Matter	S
4. Quantum Algorithms for Quantum Matter	S
4.1. Early Algorithms and Quantum Computa-	
tion of Matter	Т
4.2. Hybrid Quantum–Classical Algorithms	U
4.2.1. General Concepts of the Variational	
Quantum Eigensolver Algorithm	U
4.2.2. Extensions of VQE	V

Special Issue: Quantum Materials

Received: June 16, 2020



Quantum Matter

Α

0

4.2.3. Beyond VQE: Other Hybrid Quantum-	
Classical Algorithms	Х
4.2.4. Quantum Convolutional Neural Net-	
works for Quantum Phase Classification	Х
4.3. Error Mitigation to Enable Practical, Near-	
Term Quantum Algorithms for Strongly	
Correlated Quantum Matter	Y
5. Decoherence and Noise in Quantum Systems	Z
5.1. Markovianity: Definition and Measures in	4.0
Quantum Systems	AB
5.2. Methods in Non-Markovian Dynamics of	10
Quantum Systems	AC
5.2.1. Perturbative Approaches to Open Quan-	٨٢
5.2.2 Numerical Approaches to Open Ouan	AC
s.z.z. Numerical Approaches to Open Quan-	۵D
5.3 Emerging Directions in Environmentally	ΑU
Coupled Quantum System Dynamics	AF
6. Conclusions, Outlook, and Future Directions	AG
6.1. Outlook on Correlated Quantum Matter for	
Ouantum Information Science	AG
6.2. Predicting Correlated Quantum Matter with	
New Quantum Algorithms	AH
6.3. Open Quantum Systems and Quantum	
Transport on NISQ Devices	AI
Author Information	AI
Corresponding Authors	AI
Author	AI
Author Contributions	AI
Notes	AI
Biographies	AI
Acknowledgments	AJ
Glossary	AJ
Keterences	AK

1. INTRODUCTION

Accurate and efficient prediction of molecules and materials is one of the most important outstanding problems in science and engineering. Discoveries in quantum materials, which are characterized by the strongly quantum-mechanical nature of electrons and atoms, have revealed exotic properties that arise from correlations, including unconventional superconductors, topological insulators, spin liquids, and new phases of matter induced via strong light-matter interactions. These systems exhibit remarkable effects including quantum interference, tunneling, fluctuations, entanglement, and topological quantum states. However, the physics of quantum materials has conventionally been inaccessible to numerical modeling, so many of their interesting phenomena remain poorly understood, and technological applications are elusive. For example, electronic structure simulations of molecules with long-range interactions are beyond the capabilities of the best supercomputers and are extremely challenging when collective quantum effects become important, as in strongly correlated electron systems. Further, real quantum materials have heterogeneity and disorder that are challenging to model.

We anticipate that quantum materials to be discovered and developed in the near future will transform the areas of quantum information processing, which encompasses communication, storage, computing, sensing, and metrology. Simultaneously, efforts toward developing new quantum algorithmic approaches for quantum simulation and advanced calculation methods for many-body quantum systems will enable major advances toward functional quantum materials and their deployment. It is the promise of quantum materials for quantum information science at the same time as the potential of new quantum algorithms to discover entirely unexplored quantum matter that has inspired this Review. Over the past few years, there has been substantial activity in establishing the prediction of molecules as the primary near-term application of quantum computation. There are similarly impressive discoveries to come in quantum computation of quantum materials. Our Review is therefore timely and presents the state-of-the-art in the field toward algorithms with nonexponential complexity for strongly correlated quantum systems, with applications in excitonic quantum matter and many-body quantum states, large-scale entangled states, and high-temperature superconductivity.

The challenge for classical approaches is the treatment of the intrinsically quantum mechanical nature of the problem, in particular for highly entangled or correlated states. The advent of quantum computing raises new possibilities for eliminating the exponential complexity that has stymied simulation of strongly correlated quantum systems on high-performance classical computers. Despite this promise, the current challenge for quantum approaches is the limited size and coherence of available quantum hardware. While the promise of quantum computing for quantum materials is large, the current state of the field is best described as "noisy intermediate-scale quantum" (NISQ). Heuristically, in this regime quantum computers are too large to be directly simulated on classical computers but remain too small to perform useful error correction. Utilizing various hardware realizations such as superconducting qubits, trapped ions, cold atoms, or photonic systems, academic groups and commercial ventures have demonstrated various levels of operational control and entanglement on tens of qubits. Here, we aim to review new algorithms and computational approaches to predict and understand the behavior of correlated quantum matter, utilizing both quantum and classical resources.

In 2020, the field of quantum information processing is on the cusp of a critical moment: special-purpose NISQ devices have become capable of solving certain problems that classical computers cannot solve, and quantum computers are poised to show practical "quantum advantage". This is in part due to the breakthrough announced in 2019, showing that a quantum device can perform calculations that are classically intractable, benchmarked on the large supercomputer Summit.¹ Researchers at Google and NASA presented heuristic benchmarking showing that their 54-qubit superconducting (SC) circuit quantum computer performs certain sampling algorithms much faster than classical computers,² though these algorithms have no known practical application. However, Google's SC quantum computer is still firmly in the regime of NISQ devices, as its components are so noisy that the chances of completing a task vanish exponentially with task size. That this "quantum advantage" achievement would be first realized by a SC architecture is not surprising, as staggering progress in recent decades has enabled high-fidelity and high-speed qubit initialization and logic gates. This result is an important step toward incorporating quantum information as a general purpose tool for scientific problems. Researchers at Google proceeded a step further and utilized their superconducting processor to determine molecular energies using experimentally imple-mented Hartree–Fock theory,^{3–5} illuminating the pathway toward practical quantum advantage for molecular and material problems.



Figure 1. Scalable quantum systems have been the focus for many different fields including quantum optics, quantum chemistry, materials science, and computer science. For instance, method development in quantum electrodynamics, correlated electronic structure, open quantum systems, computational algorithms, and the interplay between these fields have provided a better understanding of quantum systems. Their progress has enabled physical realizations of novel quantum information technologies including quantum networks, quantum computers, and quantum sensors.

In parallel, over the past few years, the ability to connect quantum information processors by quantum communication channels, toward a future "quantum internet", has enabled a wide variety of multiparty computation and decision making protocols that go beyond classical secure communication. In 2020, we saw the first demonstration of a quantum repeater, an important step toward enabling large-scale quantum networks. This would allow secure communication of information over long distances using matter and photonic based quantum information systems.⁶ Quantum technologies for precision measurement already form the basis for the world's most accurate clocks and sensors. We expect that the next generation of sensing technologies will use large-scale quantum coherence and entanglement to increase accuracy and precision by many orders of magnitude. There is inherent synergy between a better understanding of quantum many-body dynamics and developing new quantum information technologies. Therefore, in this Review, we emphasize quantum materials for quantum information science, enabling the development of physical systems that can coherently store and manipulate quantum states well enough for error correction.

In the NISQ-era, quantum devices with increasing complexity are emerging rapidly. Multiple different hardware platforms have become available recently that are intrinsically capable of simulating quantum systems due to their quantum nature. Ideas in quantum simulation have been realized in a wide range of systems, including superconducting qubits, molecular qubits, defect systems, photonic platforms, and topological systems. With a plethora of these novel platforms, there is now an opportunity for the realization of functional quantum resources: quantum devices that offer practical quantum advantage in simulations.

A significant focus of the community has been creating robust and resilient quantum devices, many based on new materials. Therefore, an accurate description of the electronic structure of molecules and quantum materials is critical for predicting the properties of these quantum devices. The electronic structure gives rise to quantum mechanical states, which can then be entangled and used as a "quantum resource" or more directly lead to novel applications, such as the optimization of molecular energy storage. The electronic structure of such systems is a good candidate for quantum computation since the inherent quantum nature of quantum devices provides natural benefits for predicting quantum behavior. However, the realization of quantum algorithms on quantum devices faces the major challenge of susceptibility to noise, leading to sizable error effects. These error rates are due to multiple factors: qubits can only stay in a mixed state for a certain period of time, the qubit operations are susceptible to accuracy errors, and subtle changes in the environment can affect accuracy. To consider quantum matter beyond small-scale model Hamiltonians, reducing the error rates is essential for maximizing the reliability of the results produced by current hardware.

This discussion shows that the description, prediction, and optimization of these quantum devices requires the intersections of methodologies from different fields including but not limited to quantum electrodynamics, electronic structure theory, open quantum systems, and algorithm design, as depicted across the top row of Figure 1. Quantum electrodynamics focuses on the accurate description of quantum systems under light-matter interaction, their optical properties, and experimental realization. Electronic structure theory, including methods such as Hartree-Fock, density-functional, many-body perturbation and coupled cluster theories, aims to accurately predict electronic configurations within general molecular and material systems. Open quantum systems focus on systems whose behavior and dynamics are dictated by system-environment interactions, including environmental noise. Computational algorithm design draws ideas from classical computing and computer science while exploiting the quantum properties of the hardware for universal quantum computation. These fields are wellestablished and have produced accurate and general methods

Chemical Reviews

for predicting behavior in quantum systems. Combining recent breakthroughs from these fields with ideas from computer science to optimally design quantum algorithms has led to physical realizations and novel application of a wide range of quantum technologies including quantum networks, quantum computers, and quantum sensors. Beyond the field of quantum information, these platforms promise a new perspective on some of the longstanding problems in chemistry such as improvements in the design of molecules for catalysis, photosynthetic light-harvesting, and energy storage.

1.1. Layout and Sections

In this Review, part of a Thematic Issue on Quantum Materials, we provide a fresh look at classical and quantum algorithms to approach novel quantum systems. With increased understanding and remarkable experimental imaging of quantum behavior of molecules and materials, these systems are becoming technologically relevant. Researchers have made tremendous progress toward the accurate treatment of the electronic structure of quantum mechanical systems, tailoring and controlling correlations, thereby allowing the design of novel molecules and materials.⁷⁻¹¹ In section 2, we discuss classical methods in electronic structure theory that use the wave function as the primary variable of interest to accurately treat quantum systems where electronic correlation is critical. In section 2.1, we discuss the definition and importance of electronic correlation in molecules and materials. In section 2.2, we discuss classical electronic structure theory methods including Hartree-Fock, configuration interaction, and coupled cluster theories. We then highlight molecular qubit candidates as both an independent and exciting active field of experimental research and as a potential application of these classical methods.

An alternative approach from using the wave function as the primary variable of interest is to consider a density perspective, which we discuss in section 3. In section 3.1, we highlight reduced density matrix techniques, with an emphasis on the variational method to capture correlation in molecular and material compounds. We then discuss physically motivated density matrix constraints that can be used as error mitigation on quantum devices. We briefly discuss reduced density matrix functional theory, leading into the remainder of this section, which focuses on density functional approaches.

Density functional based methods have been extensively developed and utilized for describing aspects of molecular and solid-state quantum systems. Research advances have extended DFT methods to be time-dependent and, importantly, have allowed for direct inclusion of correlations with electrons, phonons, and photons. There have also been method developments such as the quasiparticle GW approach, which builds upon DFT. Altogether these methods have been timely, as experimental work has expanded to include novel materials and chemical systems, along with the inclusion of light-mediated and cavity-driven interactions. Light-mediated manipulation of quantum matter is particularly attractive, as it enables control at fundamental time scales and access to novel nonequilibrium states of matter. Ultrafast and quantum electrodynamical methods have been recently used as a new tuning knob to induce insulator-to-metal transitions, topological phases, and ferroelectricity, as well as transient superconductivity in copper oxides and organic crystals. While this provides a tantalizing opportunity to explore new quantum phases, particularly in the case of transient superconductivity, key questions remain on the underlying excitation mechanisms and computational approaches to describe such systems.

With these as motivating factors, we discuss DFT methods in section 3.2. In particular, we highlight in section 3.2.1 the theoretical approaches for describing atomic point defects in crystals. These defects, which can behave as "artificial atom" qubits or "color center" qubits, serve also to highlight opportunities for algorithmic advances to capture correlated electron—nuclear dynamics in these and other quantum matter systems. Going beyond conventional DFT, section 3.3 covers time-dependent density functional theory and quasiparticle methods for correlated quantum matter. In section 3.4, we introduce algorithmic advances toward describing "driven quantum matter". In particular, we highlight *ab initio* quantum electrodynamical methods that can model inherently correlated quantum interactions, where photon, nuclear, and electron degrees of freedom are treated at the same level of quantization.

We then review, in section 4, the recent progress in quantum algorithms and how the realization of quantum devices has allowed for an alternative pathway for reducing the scaling of the *N*-body problem of interacting quantum mechanical particles. In section 4.1, we consider the inception of quantum algorithms and early progress in their application to chemical and material systems. In section 4.2, we discuss an important class of quantum-classical approaches where a limited number of atoms or orbitals are mapped on the quantum computer, while the remaining part of the material system is treated using the strength of classical supercomputing platforms. In section 4.3, we present error mitigation ideas to aid in successful simulation of large quantum systems as NISQ platforms of appropriate qubit size and coherence times become available. The diversity of hardware including gates, connectivity, control schemes, and noise characteristics are both a challenge and an opportunity; as much as one algorithmic approach may not succeed on one platform, it might be adjusted to succeed on another, as the community has recently shown. Validation and verification of the quantum hardware is important as we establish a new computational paradigm, especially with the high error rates and initially unknown nature of errors of NISQ hardware.

In section 5, we will discuss the field of open quantum systems and how this perspective can be used as a lens to think about quantum systems of interest, from atoms and molecules to quantum networks. The control of quantum systems requires coupling diverse subsystems together while simultaneously driving them and minimizing the detrimental effects of the environment. The inherent contradiction in using external classical fields for control while limiting the deleterious effect of the environment requires a fundamental understanding of a quantum system that is open to its environment. Modeling and controlling quantum systems that operate with some degree of decoherence is an interdisciplinary problem, as we will illustrate. In the simplest case where the system is well isolated and the coupling to the environment is weak, methods such as the Lindblad-Kossakowski formalism can be used to capture decoherence and loss due to the environment. While this alone is challenging, certain regimes, such as strong systembath coupling, are much more complex. Strides toward defining and distinguishing between different regimes and capturing the resulting dynamical properties are discussed in sections 5.1 and 5.2, respectively. Further, in section 5.3, we will present advances in the treatment of open quantum systems through a myriad of applications from correlated materials physics to descriptions of hybrid quantum systems.



Figure 2. Electronic structure methods are a computational backbone for building quantum matter. From atoms to bulk material, correlated electronic structure methods are used to investigate properties including orbital densities, potential-energy surfaces, level diagrams and optical transitions, band structures, and phase diagrams.

We acknowledge that the strongly interdisciplinary nature of the topics covered is unprecedented and there is a need for a common language to integrate ideas from these fields. In this Review, we aim to provide this common language while weaving these fields together. Taken together with the other reviews in this Thematic Issue on Quantum Materials, including those by Zunger et al.,¹² Felser et al.,¹³ and Nguyen and Cava,¹⁴ we present a comprehensive foundation for those entering the fields of correlated quantum matter and quantum information science.

We conclude the Review in section 6 with open questions in quantum simulations and algorithms for correlated quantum matter. In particular, we highlight directions beyond variational quantum eigensolvers and discuss recently developed algorithms to capture correlations, entanglement, and excited states in quantum materials. Further, we present forward-looking ideas in predicting and controlling driven quantum matter.

2. WAVE FUNCTION-BASED ELECTRONIC STRUCTURE METHODS FOR QUANTUM MATTER

Ever-increasing computational power, theoretical advances, and improvements in algorithm design have opened new frontiers in 21st century quantum chemistry and quantum materials science. Treating atoms, molecules, and materials using computational methods involves solving a many-body problem of interacting electrons and nuclei and possibly also coupled electromagnetic fields. Due to the vast configuration space, this problem is exponentially complex and can only be solved exactly for very small systems. Therefore, all practical methods for real systems rely on approximations, and balancing the computational costs of these methods with the accuracy of the prediction is a constant challenge. The simplest methods to treat interacting multicomponent systems are adiabatic and semiclassical approximations. Examples include the Born-Oppenheimer approximation, which simplifies electron-nuclei problems, or a semiclassical Schrödinger-Maxwell treatment, which simplifies the problem of the electromagnetic field interacting with electronic systems. In each method, the many-body problem is reduced to a problem of interacting electrons, which are treated as quantized, and a separate problem of nuclei or the electromagnetic field governed by Maxwell's equations. The different time scales of nuclear, electronic, and photonic motion provide justification for these approximations. When considering the electronic degrees of freedom, relatively accurate treatment for electronic correlation is often critical for obtaining important molecular and material properties such as those depicted in Figure 2. In this section, we will focus on the development of electronic structure methods and how these theories can enable future experiments in quantum materials and quantum information.

2.1. Introduction to Electronic Correlation Effects

The correlation energy of a general molecular electronic system can be given by

$$E_{\rm corr} = E_{\rm exact} - E_{\rm HF} \tag{1}$$

where $E_{\rm exact}$ is the exact total electronic energy and $E_{\rm HF}$ is the Hartree-Fock or mean-field energy. Generally the correlation energy is the energy associated with the interaction between electrons beyond a single-particle picture or the part of the electronic energy that the Hartree-Fock method fails to capture.¹⁵ For quantum matter applications, electronic correlation can be divided into two categories, static and dynamic.¹⁶ Static or multireference electronic correlation arises when the low-energy configurations are degenerate or nearly degenerate with the reference Slater determinant, as in stretched molecular configurations.¹⁷ In other words, a single molecularorbital diagram is insufficient to describe the wave function and more determinants must be included.¹⁸ Dynamic correlation arises from the repulsion due to the relative motion of electrons and often requires the need to mix higher-order excited configurations with the reference Slater determinant. $^{16}\ \mathrm{As}$ an example of these energies, the total energy, Hartree-Fock energy, and correlation energy of hydrogen fluoride at varying bond distances are shown in Figure 3.



Figure 3. Hartree–Fock (teal) and full-configuration interaction (gray) energies versus atomic distance *r* for the dissociation of hydrogen fluoride in the STO-6G basis set¹⁹ calculated using PySCF.²⁰ The separation dependent correlation energy is shown as $E_{\rm corr}(r)$ as the difference between $E_{\rm HF}$ and $E_{\rm FCI}$.

There are several popular methods of approximating the correlation energy, including Møller-Plesset perturbation theory (MPT), configuration interaction (CI) theory, and coupled cluster (CC) theory. When using any approximate or perturbative method, there are a few important properties to preserve while considering the electronic structure of a system, including size consistency and size extensivity. Size consistency refers to the energy of a molecular system being equivalent to the sum of its individual parts at infinite separation, whereas size extensivity refers to the energy scaling with the system size.²¹ Due to the physical nature of size consistency and size extensivity, methods that satisfy these properties are generally preferable. Accurate calculation of strong electronic correlation in molecular and extended material systems is critical for a variety of applications. Predicting and controlling the electronic entanglement is essential to designing large molecular systems for use as qubits or to store information or energy.²²⁻³² Traditional formulation of the electronic correlation problem leads to exponential scaling in system size; however, current progress in the field is producing novel methods for treating electron correlation with improved scaling.

In section 2.2, we will discuss the traditional quantum chemistry methods as relevant to predictions of correlated quantum matter. In section 3, we will then discuss other routes such as density functional and time-dependent density functional theory and its extensions, quasi-particle and reduced density matrix based methods, and other recent techniques including machine learning-based approaches in computational condensed matter physics.

2.2. Quantum Chemistry Approaches to Correlated Quantum Matter

The simplest approximation to intuitively consider electronic behavior is to ignore explicit treatment of electron–electron correlation and instead consider the energy and configuration of a single electron in the average field of other electrons. This class of approaches is referred to as mean-field methods and the Hartree–Fock method is the most commonly used in quantum chemistry. In the Hartree–Fock method, the solution of the interacting many-body problem of *N* electrons is approximated by a single Slater determinant,

$$|\Psi_0\rangle = |\chi_1\chi_2\dots\chi_N\rangle \tag{2}$$

where χ_i are single-particle wave functions. The variational principle can be used to determine the optimal wave function $|\Psi_0\rangle$ that leads to the lowest energy, E_0 :

$$E_0 = \langle \Psi_0 | \hat{H} | \Psi_0 \rangle \tag{3}$$

where \hat{H} denotes the Hamiltonian of the system.³³

pubs.acs.org/CR

The Hartree–Fock approximation leads to a set of singleparticle equations, known as the Hartree–Fock eigenvalue problem. The ground state is composed out of the *N* lowest occupied orbitals, while the other orbitals are unoccupied or virtual orbitals. By choosing a single Slater determinant and therefore only considering occupied orbitals and by ignoring the correlated nature of electron–electron interactions, Hartree– Fock scales as N_{orb}^{3} , where N_{orb} is the basis set size, while capturing 99% of the electronic energy for a wide range of atoms, molecules, and materials.³³ As a mean-field approximation, the Hartree–Fock method fails to capture the electronic structure of strongly correlated systems. This critical failure has led to the development of many methods to include electronic correlation, commonly referred to as post-Hartree–Fock methods.

A standard approach for improving upon the Hartree–Fock method is by including an increasing number of Slater determinants, consequently increasing the size of the Hilbert space. Many of these extensions use active spaces as a method of decreasing the space of chemical interest and therefore decreasing the computational cost, allowing for the treatment of larger quantum systems. Using an active space involves selecting a subset of orbitals to be treated with a high level of theory, while the remainder of the orbitals are treated at a mean field level of theory.³⁴ Generally the valence orbitals near the HOMO–LUMO gap are selected as the active space, while the lower energy occupied orbitals (core orbitals) and the high energy unoccupied orbitals (virtual orbitals) are treated at a lower level of theory. Active spaces are incorporated into many of the post-Hartree–Fock methods discussed below.

An important class of post-Hartree-Fock methods is the multiconfigurational self-consistent field (MC-SCF) methods.^{35–37} In MC-SCF, the wave function is written as a linear combination of Slater determinants with coefficients that are determined variationally through minimizing the energy of the MC-SCF wave function. MC-SCF methods have been used to accurately treat static correlation in a variety of molecular systems, including transition metal and actinide chemistry.³⁸ An important subset of MC-SCF methods are the complete active space self-consistent field (CASSCF) methods. These involve generating a full configuration interaction wave function within the active space and then variationally optimizing the orbitals and configuration coefficients.³⁴ A more recently developed subset of MC-SCF methods is the generalized active space (GAS) method where multiple active spaces are chosen and controlled such that insignificant configurations are neglected.41-43 Restricted active space (RAS) and localized active space (LAS) methods are other directions that have been explored,^{44,45} the latter being introduced recently to improve the use of active space methods in the density matrix embedding framework.^{46,45}

There are a variety of perturbative techniques for adding electronic correlation to the Hartree–Fock energy in molecular

systems. Møller-Plesset perturbation theory is a systematic way of adding correlation energy to the total electronic energy of a system.⁴⁸ This method is generally used to second order in correlation potential, denoted as MP2, and while this method originated in the 1930s, much progress has been made since then, which has allowed it to become a popular method for capturing electronic correlation.⁴⁹ Methods such as the resolution of identity (RI) or density fitting,^{50–55} Cholesky decomposition,^{56–60} local correlation,^{61,62} and combinations therein have reduced the complexity of the MP2 method and introduced efficient computational schemes allowing for its use to treat molecular systems with hundreds of atoms.⁶³ The advantages of MP2 include polynomial scaling, from $O(N_{\rm orb}^{5})$ down to $O(N_{\text{orb}})$ with the correct approximations,⁶⁴⁻⁶⁹ sizeconsistency, and the ability to capture dynamic correlation. However, the treatment of electronic correlation is incomplete and therefore inaccuracies arise when treating systems that include stacking or hydrogen bonding.⁷⁰⁻⁷² Many method extensions based on MP2 have been developed to overcome such drawbacks, notably spin component scaled MP2, which has been shown to improve the ground-state energies in noncovalent interactions, $^{73-78}$ orbital-optimized MP2, $^{79-82}$ spinnetwork-scaled MP2, which relies on the inclusion of neural nets to weight interaction energy terms,⁸³ the use of natural orbital functionals to include static correlation,⁸⁴ and many more.^{85–87} Other perturbative techniques include the complete active space perturbative approach to second order (CASPT2)^{88,89} and the restricted active space perturbative approach to second order (RASPT2).90 Both of these perturbative methods improve the CASSCF energy by including dynamical correlation through the use of perturbation theory. These methods have been successful in the description of excited states,⁹¹ dissociation energies,⁹² and magnetic properties.⁹³ A specific example of the use of CASSCF and CASPT2 was for the investigation of the spin density on the metal centers of a variety of molecular qubit candidates.³¹ Alongside experimental investigation, the ability to theoretically investigate molecular properties of these novel qubit candidates allows for improved design and efficiency.

Another possible avenue for including electronic correlation is through using the configuration interaction method. In the full configuration interaction (FCI) method, all determinants contribute to the wave function, and in the limit of an infinite basis set, the exact energies could be calculated.³³ The wave function can be written as

$$|\Psi\rangle = (I + \hat{T})|\Psi_0\rangle \tag{4}$$

with

$$\hat{T} = \sum_{i} \hat{T}_{i} \tag{5}$$

where \hat{T}_i is the operator that creates all particle excitations of order *i* and $|\Psi_0\rangle$ denotes a reference state, which is commonly the Hartree–Fock wave function. The prominent drawback of this method is that the scaling is exponential in terms of system size. Due to the exponential scaling, even with small basis sets the FCI method can only capture the electronic configuration for cases in which no more than around 18 orbitals are necessary.³³ Although the scaling of this method is computationally unfavorable, multiple flavors of configuration interaction have been explored. The simplest approximation is to truncate the level of excitations, starting with including all determinants that represent a single excitation (CIS), then single

pubs.acs.org/CR

and double excitations (CISD), then single, double, and triple (CISDT), and so on until the desired accuracy is obtained.⁹⁴ Another method is considering the seniority number of the wave functions or determinants to include. Notably, the doubly occupied configuration interaction (DOCI) method restricts the FCI space to the double occupied space or to the seniority zero space, where only determinants that represent paired configurations are included.⁹⁵ While this produces a reduced scaling, it is still exponential and therefore limited in a similar way to the FCI. Numerous different seniority schemes have been studied and applied successfully to treat small molecular systems.⁹⁶ Recently, machine learning techniques have been implemented using artificial neural networks to select which configurations are important contributions to the electronic wave function.^{97,98}

A major drawback of the CI method is that when the level of excitations is truncated, it is no longer size-extensive.⁹⁹ Using an exponential excitation operator ansatz to include higher order excitation determinants to the wave function allows for size extensivity to be recovered, in a method referred to as coupled cluster. The wave function in coupled cluster theory is often written as

$$|\Psi\rangle = e^T |\Psi_0\rangle \tag{6}$$

with

$$\hat{T} = \sum_{i=1}^{N} \hat{T}_i \tag{7}$$

where $|\Psi_0\rangle$ denotes a references state and \hat{T}_i is the operator that creates all particle excitations of order *i*. For instance, \hat{T}_1 is the operator that creates all possible single particle excitations and \hat{T}_2 double excitations. The highest possible excitation operator is \hat{T}_N , where *N* is the number of electrons in the system. Typically the Hartree–Fock wave function is used as the reference state; however other options also exist.

Analogous to the CI expansions, coupled cluster methods fall into classes according to excitation order by including single and double excitations (CCSD), including triple excitations (CCSDT), and so on.¹⁰⁰ Using a single reference state, the Baker–Campbell–Hausdorff (BCH) formula can be used to determine the ground-state energy. Due to the commutation properties of the excitation operators, the similarity-transformed Hamiltonian expansion is truncated at fourth order, thus allowing for efficient energy calculations.^{99,101} While this allows for low computational cost, the BCH formula only holds for single reference states, and in this sense can not capture strong static correlation. In addition, the coupled cluster ansatz is not variational and is therefore similar to a perturbation theory.

A further reformulation of the CC ansatz involves ensuring that the excitation operator is unitary such that

$$\Psi\rangle = e^{\hat{T} - \hat{T}'} |\Psi_0\rangle \tag{8}$$

where \hat{T}^{\dagger} is the conjugate transpose of the excitation operator \hat{T} . The unitary coupled cluster (UCC) formulation has the benefit of being variational; however, the calculation proves to be challenging as the BCH formula no longer truncates.^{102,103} Method development^{104,105} and computational improvements¹⁰⁶ have allowed coupled cluster methods to accurately predict dynamic electronic correlation¹⁰¹ in molecular and solidstate systems.¹⁰⁷ Moreover, there has been a renewed interest in UCC as it can be used in conjunction with a classical optimization strategy in variational quantum eigensolver (VQE) algorithms for quantum devices, a class of methods that will be discussed in section 4.

In the following, we want to briefly comment on the validity of the different approximations in the context of correlated quantum matter. To this end, the scaling and performance in capturing the dissociation of hydrogen fluoride for a sampling of the above methods including Hartree–Fock, MP2, CCSD, FCI, and DOCI is shown in Figure 4a,b, respectively. The scaling and



Figure 4. (a) Approximate computational scaling of a variety of electronic structure methods including Hartree–Fock (HF), density functional theory (DFT), Møller–Plesset perturbation theory to second order (MP2), variational 2-electron reduced density matrix (v2RDM), coupled cluster single doubles (CCSD), and full and double occupied configuration interaction (FCI and DOCI, respectively) where *N* is the basis set size. (b) Dissociation curve of hydrogen fluoride, comparing the accuracy of the Hartree–Fock, MP2, CCSD, v2-RDM, DFT, and FCI methods.

performance of density functional^{108,109} and reduced density matrix methods are also included and will be discussed in section 3. It should be noted that a minimal basis of STO-6G was used for these dissociation curves and improved results could be obtained by using a more complete basis.

Many of these methods have been in use for decades, and there continues to be an abundance of research dedicated to extensions, improvements, and applications in prediction of new quantum materials and phases of quantum matter. As discussed above, one such recent development is mapping these methods from classical high-performance computing onto quantum devices. While this will be discussed in detail in section 4, methods such as Hartree–Fock,^{1,3} full configuration interaction, and coupled cluster theory have all been used as ansätze in hybrid quantum algorithms on quantum devices. A considerable amount of research is currently being dedicated to adapting classical electronic structure theory methods to be amenable to the framework of gate-based quantum computing.

Another approach to overcome the computational challenges in the methods presented above is dividing the system into smaller computationally cheaper subsystems and then combining these subsystems. When one or some subsystems and the surrounding environment are treated at different levels of theory, this is often referred to as quantum embedding.¹¹⁰ Embedding theories aim to predict properties of interest of the subsystem or impurity without having to perform a costly calculation on the whole quantum system. A plethora of embedding models exist, each focusing on different methods for dividing up the system and utilizing different computational techniques for treating each part.¹¹⁰⁻¹¹³ For example, in chemical quantum systems it is often natural to focus on geometry-based approaches, carefully avoiding breaking chemical bonds that contribute to the strong correlation of the system.¹¹⁰ The computational techniques differ based on the primary variable of interest. There are three main options leading to three main types of embedding theories:^{110,114} Green's function embedding,¹¹⁵ density functional embedding,^{116–118} and density matrix embedding.¹¹⁹ Excellent and comprehensive reviews of embedding theories are presented in refs 114 and 110.

One of the most well-known embedding theories for the treatment of strong correlation is dynamical mean-field theory (DMFT) in which the infinite bulk system is mapped to an impurity model that consists of a subsystem embedded in a noninteracting environment.¹²⁰⁻¹²⁴ The correlated frequencydependent Green's function of the impurity is determined selfconsistently, often resulting in predictions that approximate the bulk limit. Since the calculation of Green's functions is a more complex task than the calculation of stationary states, DMFT is more computationally costly than necessary for ground-state calculations. An alternative embedding approach referred to as density matrix embedding theory (DMET) utilizes the 1-body density matrix as the variable of interest instead of the 1-body Green's function.¹¹⁹ In the noninteracting limit, this theory is formally exact.^{110,119} Through the DMET correlation potential, DMET self-consistently optimizes the impurity embedded in an approximate many-body Schmidt basis for the environment.^{119,125} To treat the environment, often low-level theories such as Hartree–Fock^{119,126} are utilized. The impurity or subsystem is treated at a high-level of theory using methods such as coupled cluster¹²⁷ or complete active space self-consistent field.¹

DMET has been successfully applied to a variety of spin models, including 1D^{119,127,129,130} and 2D Hubbard models,^{119,131–134} and in chemical systems. A few chemical examples include small molecular configurations such as hydrogen rings, hydrogen sheets, and beryllium rings^{126,135} and extended molecular systems including boron-nitride sheets, polymers, and diamond structures.¹²⁷ Moreover, DMET has been successful at capturing and predicting chemical processes including hydrogen chain dissociation¹³⁶ and organic substitution reactions.¹³⁵ While DMET has shown to be an efficient technique for solving ground-state properties, successful extensions have also been developed for calculating excited-state properties through response theory formulations.^{131,137,138}

2.2.1. Predicting Molecular Qubits. A promising application for these classical wave function electronic structure methods, extensions, and the density methods, which will be discussed in section 3, is in the investigation of molecular qubit candidates. While many different types of qubits are being experimentally prepared, verified, and utilized, spin-based molecular qubits show great promise due to recent advances led by experimental groups.²⁹ The design of molecular spin qubits is gaining interest due to their chemical properties, which allow for ease of initialization into a well-defined quantum state,

enhanced quantum coherence, natural implementation as quantum logic gates, ease of integration into devices, and potential for creation of quantum networks.^{139,140} Their atomic precision, magnetic properties, and synthetic tunability make them excellent candidates for quantum computing applications, quantum sensors, and environmental probes.^{31,141–143} Moreover, through synthetic design and theoretical insight, molecular qubits can be relatively easily produced, tuned, and controlled.¹⁴²

When considering molecular spin based qubits, nuclear or electronic spins can be considered. While nuclear spins tend to have longer coherence times, electronic spins offer a few unique advantages, including precise spatial control due to their confinement to molecular structures and ease of tunability due to improved manipulation technology such pulsed electron paramagnetic resonance.^{144,145} Electronic spins can be found in radical, diradical, or coordination complexes spanning the periodic table,¹⁴⁵ including transition metal²⁷ and lanthanoid complexes.^{139,146} A few examples of such molecular qubit compounds are shown in Figure 5.²⁷



Figure 5. Crystal structures of a variety of the most relevant spin qubits made from transition metal ions where purple, orange, pink, light blue, green, maroon, light green, blue, yellow, red, and gray spheres represent iron, copper, chromium, nickel, vanadium, bromine, fluorine, nitrogen, sulfur, oxygen, and carbon atoms, respectively. Reprinted with permission from ref 27. Copyright 2017 American Chemical Society.

Like many other platforms for quantum computation, one of the biggest challenges that molecular qubits face is in extending and controlling coherence times.¹⁴² The coupling to environmental degrees of freedom can impact decoherence;¹⁴⁷ however, due to the quality of current synthetic techniques and the flexibility of molecular compounds, the structure of molecular qubits can be tuned to mitigate these detrimental effects.^{25,148,149} Other challenges in the field consist of improving the control of molecular excited states and entanglement processes.¹⁴²

The potential advantages of molecules as qubit candidates would be in the versatility of molecular structures and bonding. Utilizing different chemical types could allow for the tuning of properties critical for their performance and application in quantum information science. For example, organic molecules offer advantages such as weak spin–orbit coupling,¹⁴⁹ while transition metal complexes offer the advantage of additional flexibility through the choice of both metal center and ligands, and additional vibronic degrees of freedom.^{27,142} While the chemical components of individual molecular qubit candidates are important, another key advantage of these specimens lies in

their intermolecular interactions. These bonds can be exploited to construct molecular arrays and therefore qubit arrays.¹⁴²

Going beyond single molecules, one study set out to verify a theory that predicted that long coherence times should exist in defects in conjugated carbon materials.^{30,150} In these works, the molecular bonding structure was used as a form of qubit engineering to create unpaired electronic spins in molecular graphene nanostructures. Another experimental study extended the concept of molecular qubits through use of metal–organic framework chemistry to create an array of qubit candidates and investigate their properties. This work demonstrates considerable progress toward the creation of a quantum network based on molecular qubits that could have wide reaching applications in quantum information sciences.³²

The majority of the progress in spin molecular qubits has been driven by experiments; however, there is great potential for theoretically and computationally guided search for the ideal molecular qubits. One such example utilized CASPT2 to predict the pulse electron paramagnetic resonance spectrum of the spin density on the metal centers in molecular qubit candidates.³¹ Other work has utilized a combination of electronic structure methods to investigate the spin-vibrational coupling in lanthanide complexes.¹⁵² As this field progresses, theory has the potential to aid in the experimental efforts to design, tune, and control interesting molecular properties in these spin qubit candidates.

While classical electronic structure methods can be used to investigate, predict, and optimize the chemical properties of small- to medium-sized molecular compounds, they face challenges in terms of computational cost when trying to predict properties in larger material structures and molecular networks. We recognize that molecular qubits are a nascent platform, yet they represent a good model system for many of the methods in correlated quantum matter discussed in this Review. The remainder of this Review discusses alternative methods to predict properties of such systems, including density based approaches as discussed in section 3 for systems such as molecular graphene nanostructures, algorithm developments on quantum devices as discussed in section 4, and open quantum system based approaches as discussed in section 5 for systems such as molecular networks.

3. DENSITY-BASED ELECTRONIC STRUCTURE TO PREDICT QUANTUM MATTER

In the previous section, we presented solutions to quantum many-body problems in terms of a ground-state or excited-state many-body wave function Ψ . While wave functions provide a natural path to obtain observables, it is usually not feasible to construct many-body wave functions due to the exponential scaling of the parameter space. Other approaches to describe the electronic ground and excited states of a quantum system can be based on reduced quantities, such as the density matrix, the electron density, or the Greens function, which will be discussed in the following.

3.1. Reduced Density Matrix Methods for Correlated Quantum Matter

An alternate approach for considering the electronic state of a system is to consider the density matrix,

$$^{N}D = |\Psi\rangle\langle\Psi| \tag{9}$$



Figure 6. Applications of different flavors of the variational 2-RDM method and *N*-representability conditions. (a) Electron density contour plot of the highest occupied molecular orbital in vanadium(III) oxo determined from a [42,40] active space variational 2-RDM method showing the additional electron distributed over the ligands. Reprinted with permission from ref 7. Copyright 2016 American Chemical Society. (b) Electronic pictures for the vanadium(IV) oxo complex (left) and vanadium(III) oxo complex (right) with the upper panel showing the results from the [42,40] active space variational 2-RDM calculation and the lower panel showing the traditional results from ligand field theory. Reprinted with permission from ref 7. Copyright 2016 American Chemical Society. (c) Vertical band gap for the acene chains of length one to five comparing CASCI, CIS, TDDFT, and the ES-2RDM method using a π orbital active space. Reprinted with permission from ref 215. Copyright 2018 American Chemical Society. (d) Dissociation curve of the NO⁺ molecule comparing the doubly occupied configuration interaction method with the variational pair 2-RDM method using *DQG* and various *DQGT* constraints. Reproduced with permission from ref 216. Copyright 2018 AIP Publishing.

where ${}^{N}D$ is the N-particle density matrix and Ψ is the wave function. The N-particle density matrix is the same size as the wave function, so it would provide few computational advantages over using the wave function directly; however, since electrons interact pairwise, a natural object to consider is the 2-electron reduced density matrix (2-RDM),¹⁵³

$${}^{2}D(1, 2; 1', 2') = \int \Psi(1, 2, ..., N) \Psi^{*}(1', 2', ..., N) \, \mathrm{d}3 \, ... \, dN$$
(10)

where the elements are given by

$${}^{2}D_{kl}^{ij} = \langle \Psi | \hat{a}_{i}^{\dagger} \hat{a}_{i}^{\dagger} \hat{a}_{l} \hat{a}_{k} | \Psi \rangle \tag{11}$$

where second quantization notation is used and \hat{a}^{\dagger} and \hat{a} are the fermionic creation and annihilation operators, respectively. The 2-RDM represents the probability of two electrons interacting in a field of N - 2 electrons. While one can obtain a 2-RDM from any wave function by contraction or integration, not every general two-electron density matrix corresponds to an Nrepresentable wave function. To ensure that a 2-RDM represents a physical N-particle system, the 2-RDM must satisfy additional constraints known as N-representability conditions.¹⁵⁴⁻¹⁶⁰ Advances in semidefinite programming (SDP) and the derivation of a formal hierarchy of ensemble Nrepresentability conditions on the 2-RDM have led to recent applications of 2-RDM theory to important problems in the study of strongly correlated molecules. Progress has also been made in the study of pure N-representability conditions of the 1and 2-RDMs.^{161,161-1}

For practical applications, an approximate set of necessary constraints are the 2-positivity *DQG* constraints, ^{157,171}

$$^{2}D \ge 0 \tag{12}$$

$$^{2}Q \ge 0 \tag{13}$$

$$^{2}G \succeq 0 \tag{14}$$

where ${}^{2}Q$ is the two-hole RDM, ${}^{2}G$ is the particle-hole RDM, and $M \geq 0$ implies that the matrix M has non-negative

eigenvalues or that the matrix must be positive-semidefinite. These constraints are physically motivated as the probability of finding two particles, two holes, or one particle and one hole must always be non-negative. It should be noted that while the DQG-positivity constraints are often sufficient for relatively accurate electronic structure calculations, they are a subset of the exact *N*-representability conditions.^{160,171}

Several methods utilize the 2-RDM to determine ground and excited electronic states, including variational and parametric^{172–186} 2-RDM methods as well the solution to the anti-Hermitian contracted Schrödinger equation.^{187–197} Here, we will focus on the variational 2-RDM method, which is particularly useful in capturing strong correlation in systems important to molecular chemistry and condensed-matter physics.^{15,198–205}

The variational 2-RDM method consists of variationally minimizing the energy of a system as a linear functional of the 2-RDM,

$$E = \mathrm{Tr}(^{2}K^{2}D) \tag{15}$$

where ${}^{2}K$ is the two-body Hamiltonian under *N*-representability constraints. ${}^{155-160,206-211}$ The energy produced from this constrained variational minimization is a lower bound on the ground-state energy. 161 While the *DQG* constraints in eqs 12–14 are generally sufficient for relatively accurate electronic structure calculations, more complete sets of constraints have also been explored, including the three-index T_2 constraint which requires that $T_2 \ge 0$. 158,160,212 Constraining these matrices to be positive-semidefinite requires the use of a form of convex optimization known as semidefinite programming. 159,213,214 Beyond this application to electronic structure theory, SDPs are important in several applications, including the max-cut problem, economics, and combinatorics. 203 In the context of variational 2-RDM methods, recent advances in SDP algorithms 203 yield a scaling of N_{orb}^{6} and N_{orb}^{9} , for *DQG* and *DQG* with T_2 conditions respectively, providing a substantial improvement over wave function approaches to electronic structure calculations with exponential scaling.

The variational 2-RDM method has also been implemented as a replacement for the wave function calculation in active space methods such as CASSCF^{34,217,218} and DOCI.²¹⁹⁻²²¹ In the CASSCF framework, partitioning the orbitals into core, active, and valence allows for the variational algorithm to consider only a subset of the total electronic space and consequently treat significantly larger systems. There are several examples of this in recent literature, including applications to quantum dots,^{201,222} organometallic compounds,^{7,223,224} conductivity and transport in molecular junctions,^{225–228} exciton condensation,²²⁹ and molecular periodic systems.²³⁰ One specific example is the investigation and explanation of experimental phenomena such as the reduction of a vanadium oxo complex.⁷ An active space consisting of 42 electrons in 40 orbitals, which approximates 10²¹ variables in the wave function framework, reveals a ligandcentered reduction produced through electron entanglement. The highest occupied molecular orbital (HOMO) for the vanadium(III) oxo complexes, shown in Figure 6a, demonstrates that the electron density is delocalized across the pyridine ligands. This result contradicts both ligand field theory and smaller active space calculations, which predicted metalcentered reduction as shown in the level diagrams of Figure 6b where the vanadium(IV) complex is shown on the left and the vanadium(III) complex on the right. This result demonstrates the capacity of the active-space variational 2-RDM method to predict electron transfer in chemical systems accurately when multireference effects are important in the ground-state wave function.

For many transformations in matter including photoexcitation processes, ground-state calculations are insufficient and capturing excited-state phenomena is critical. While some early work in the RDM community studied excited-state spectra for small molecular systems, $^{231-234}$ recently these methods have been improved to accurately capture optical band gaps in larger strongly correlated molecules. Based on previous theories that use the ground-state 2-RDM along with the Hermitian operator method to calculate excited-state energies, 231-233,235-238 a recent study provides an improvement through the use of a Hamiltonian-shifted regularization algorithm.²¹⁵ The excitedspectra 2-RDM method (ES-2RDM) was used to calculate the band gaps of acene chains of varying lengths, which are in closer agreement to the CASCI gaps than those from the timedependent density functional theory (TDDFT) and configuration interaction singles (CIS), as shown in Figure 6c. Moreover, this method allows for the determination of excitation energies for larger molecules, such as optical dyes, which are important for optically sensing voltage in neurons² with greater accuracy than comparable excited-state methods.²¹⁵ ES-2RDM has since been applied to the band gap trend in graphene nanoribbons,²⁴⁰ and shows promise for investigation into excited-state phenomena in large molecular and quantum material systems.

In addition the variational 2-RDM method has been used in the pairing framework, also known as DOCI or seniority zero, where only doubly occupied configurations are considered.^{95,216,219–221,241–244} While in the configuration interaction context, restriction to the doubly occupied space maintains exponential scaling, a similar restriction under the variational framework reduces the scaling from N_{orb}^{6} to $N_{orb}^{3,219,220}$ The computational advantage comes from increased simplicity of the structure of the positivity constraints under the pair restriction.²²⁰ Early work was dedicated to considering benchmark small molecules to test the limitations of such a stringent approximation, showing that despite the importance of orbital selection and potential need for orbital optimization or rotation, the pair approximation in conjunction with the variational 2-RDM has the potential to be a powerful method.²¹⁹ There are two areas on which current research is focusing to improve the pair variational 2-RDM method. The first area of focus is determining the effects of utilizing N-representability constraints beyond the DQG conditions. Extensions from the twoindex *N*-representability constraints to the three- and four-index N-representability constraints have been considered, the latter being used to consider the Heisenberg XXZ spin model of quantum magnetism.^{216,220,245,246} An example of the effect of increasing the order of the N-representability constraints on the variational 2-RDM method in the seniority zero space can be seen in the dissociation curve for the NO⁺ molecule in Figure 6d.²¹⁶ As the molecule dissociates, all presented DOCI methods capture the correlation; however, it can be seen that with increasing N-representability constraints, the energy becomes increasingly similar to the full DOCI solution. This shows that increasing the N-representability constraints, and hence the computational cost, can systematically increase the accuracy of this method.²¹⁶ The second area of focus is orbital selection. Early work focused on orbital optimization;²¹⁹ however, this adds cost to the calculation, negating some of the speedup due to the pair approximation. One study replaced the orbital optimization procedure with an orbital localization procedure to treat hydrogen chains, acene chains, and cadmium telluride chains.^{220'} The localization of the molecular orbitals allowed the pair variational 2-RDM method to be approximately size extensive, a trait that is lost when using the canonical molecular orbitals.

From extensions to CASSCF and DOCI, variational 2-RDM approaches leverage their polynomial scaling to reveal electronic structure, which can aid in the design and understanding of quantum correlated matter for applications in energy storage, qubits, and catalysis, among others. An exciting direction for the variational 2-RDM method is its recent adaptation for use on a quantum simulator.²⁴⁷ While there has been plenty of focus on the development of variational quantum eigensolver (VQE) algorithms, which will be discussed further in section 4.2, the majority of methods focus on the variational principle for wave functions. The analogous RDM variational principle requires Nrepresentability constraints, which were shown to reduce the number of measurements required in VQE calculations²⁴⁸ and act as an error mitigation scheme.²⁴⁷ The *N*-representability constraints are imposed on the 1-RDM and the 2-RDM is then reconstructed. The pure N-representability constraints for the 1-RDM are sometimes referred to as the generalized Pauli constraints (GPCs), and have been an area of intensive research over the past decade. Coleman's ensemble N-representability conditions for the 1-RDM¹⁵⁴ mimic the Pauli exclusion principle,²⁴⁹

$$\sum_{j=1}^{\infty} n_j = N \tag{16}$$

and

$$0 \le n_j \le 1 \tag{17}$$

where n_j are the eigenvalues and N is the particle number. This means that the eigenvalues, or the natural orbital occupation numbers, must remain between 0 and 1, while their sum must equal the total number of particles in the system. While these are

state *N*-representability

pubs.acs.org/CR

fairly simple constraints, the pure state *N*-representability conditions have proven to be a more challenging topic. The pure state *N*-representability conditions are necessary and sufficient constraints placed upon the 1-RDM such that it represents a pure state, or a state that can be represented by a single wave function. The development of these pure *N*-representability constraints has been a topic of much interest due to their importance in quantum chemistry and quantum information theory. Early work was dedicated to these constraints for specific systems,^{250,251} such as the work of Borland and Dennis, which derived conditions for three fermions in six orbitals.²⁵¹ These constraints are

$$n_5 + n_6 - n_4 \ge 0 \tag{18}$$

with

$$n_1 + n_6 = n_2 + n_5 = n_3 + n_4 = 1 \tag{19}$$

where n_i are the occupation numbers of the 1-RDM from largest n_1 to smallest n_6 . These constraints are an empirical rule that are useful for three electron systems, such as H₃ or the lithium atom,^{16,164} but they do not provide a framework for systematic generalization. More recently, a systematic method for generating these 1-RDM constraints for systems beyond the three fermion case was proposed,^{252,253} as well as for the 2-RDM and general *p*-RDM.¹⁶¹ The geometry and physical relevance of these constraints have been studied,^{163,254,255} as well as the effects of the GPCs on the occupation numbers of atoms and small molecules,^{16,163,168,256,257} spin systems,^{258,259} control of entanglement,^{260,261} and methods such as the variational 2-RDM optimization¹⁶² and reduced-density-matrix functional theory (RDMFT).²⁵⁶ Recently, the three fermion constraints were verified on a quantum device as shown in Figure 7.¹⁷⁰ The



Figure 7. The entire convex polytope represents the orbital occupations of the 1-RDM allowed by the ordinary Pauli constraint, the yellow polytope represents the orbital occupations allowed by the Generalized Pauli Constraints. Each red point represents the three highest eigenvalues of a 1-RDM generated experimentally by a quantum computer. Reproduced with permission from ref 170. Copyright 2019 Springer Nature.

entire convex polytope represents the occupation numbers available for the 1-RDM under the ordinary Pauli constraints, while the yellow polytope represents the possible occupation numbers for the 1-RDM under the generalized Pauli constraints. The three highest eigenvalues of the 1-RDM as calculated on a quantum device are shown to be within the polytope defined by the Borland–Dennis constraints. This work was a novel demonstration that the GPCs on the 1-RDM are obeyed on a quantum device. The investigation into both ensemble and pure *N*-representability conditions is critical for variational 2-RDM methods and reducing the errors of chemical calculations on quantum devices.

The generalized Pauli constraints have also been explored in the context of minimizing the energy functional in reduced density matrix functional theory.²⁵⁶ RDMFT is an alternative approach to electronic structure, which combines density functional theory, discussed in section 3.2 below, and the reduced density matrix methods discussed above.²⁶² In RDMFT, the energy is expressed as a function of the 1-RDM, though this energy functional is often minimized with respect to the eigenvalues, n_i , and eigenfunctions, ϕ_i , of the 1-RDM instead of the 1-RDM as a whole. This subset of RDMFT calculations is often referred to as natural orbital functional theory since the eigenfunctions and eigenvalues of the 1-RDM are the natural orbitals and natural orbital occupations, respectively. An advantage of this method over traditional density functional theory includes the explicit dependence of the kinetic energy term on the 1-RDM, the introduction of fractional occupation numbers,²⁶³ and the storage of more information in the 1-RDM.²⁶⁴ A notable drawback of this method is that instead of an eigenvalue equation, in the RDMFT framework the determination of orbitals is done through a computationally expensive minimization. Many methods have been considered to address the issue of natural orbital selection, including the use of nonlocal potentials to obtain natural orbitals;^{265,266} however the orbital determination is still the bottleneck of the calculation. Similar to traditional DFT, a plethora of research has been dedicated to designing and improving function-als.^{262,264,267-269,269-274} Through use of these improved and novel functionals, RDMFT has successfully been used to capture the dissociation of molecules, ^{269,271,275,276} accurate band gaps, ^{277–280} and phase transitions. ^{277,279} Since its inception, many methods have extended the RDMFT framework²⁸¹ including local RDMFT,^{256,282,283} excited-state RDMFT, and time-dependent RDMFT.^{284–286} While we do not cover this method and its impact in-depth here, the references and reviews cited in this section are a good starting point for the interested reader.

3.2. Density-Functional Theory Approaches to Compute Quantum Materials

Another attractive wave function alternative is the method of density-functional theory. By exploiting the Hohenberg–Kohn theorem²⁸⁷ with the one-to-one correspondence between the ground-state density given by

$$n_0(\mathbf{r}) = \int d\mathbf{r}_1 \, d\mathbf{r}_2 \dots \, d\mathbf{r}_{n_e} \, \Psi^*(\mathbf{r}_1 \mathbf{r}_2 \dots \mathbf{r}_{n_e}) \Psi(\mathbf{r}_1 \mathbf{r}_2 \dots \mathbf{r}_{n_e})$$
(20)

and the static external potential $v_0(\mathbf{r})$, it is possible to avoid the explicit construction of the many-body wave function and directly obtain observables by expressing them in terms of functionals of the electron density. Although formally DFT is an exact reformulation of Schrödinger's equation, in practice these functionals often remain unknown. The most prominent example that has to be approximated in practical calculations is the exchange–correlation (xc) functional. The quality and accuracy of the calculation is therefore determined by the underlying approximation for the xc potential.²⁸⁸

Many different routes have been explored to find approximations for this xc potential; however, we include limited discussion here as pertinent to the scope of this Review, and we refer the reader to refs 289 and 290 for comprehensive overviews of existing approximations. The local-density approximation $(LDA)^{291}$ is the most basic approximation of the xc potential,



Figure 8. Defects in diamond and hBN using density-functional theory. (a) Pb atom shown in black in a split-vacancy configuration within the diamond lattice, consisting of carbon atoms shown in blue. Reprinted with permission from ref 305. Copyright 2019 by the American Physical Society. (b) Predicted thermodynamic charge transition levels for group III defect centers. GaV, InV, and TiV are found to be likely stable in the negative charge state for diamond samples with naturally occurring nitrogen doping, while AlV is more likely to exist in the neutral charge state. Reprinted with permission from ref 306. (c) Defect wave functions for ground and excited states in multilayer hBN for a nitrogen-vacancy defect (V_N) with the corresponding HSE based predictions shown in panel d reveal that complexes are found to offer more optically relevant defect bands within the gap. Reprinted with permission from ref 307. Copyright 2020 Springer Nature. (e) Predicted luminescence spectra for the group III emitters in diamond near room temperature conditions. Reprinted with permission from ref 306. (f) Potential energy surfaces computed for the product Jahn–Teller system in the excited state of group IV neutral defects in diamond. Reprinted with permission from ref 308.

where the electronic structure is considered as homogeneous and the functional dependency of the electron gas is then used. For the homogeneous electron gas, the exchange contribution can be calculated analytically, while the correlation contribution can be obtained to high accuracy by Monte Carlo methods.²⁹²

A natural extension to the LDA are approximations in which spatial derivatives or gradients of the density are also considered. Since the spatial derivatives are included in addition to the density at a particular point, this level of treatment is called "semi-local" and also referred to as a generalized gradient approximation (GGA). Popular functionals here are the PBE²⁹³ approximation including its extension to solids (PBEsol),²⁹⁴ which is currently one of the most widely used functionals in DFT calculations of quantum materials. With the development of the first semilocal functionals, DFT became popular for quantum chemistry applications.²⁹⁵ One way to systematically include more information into the xc functional leads to the inclusion of the kinetic energy component and is known as "meta-GGA", with realizations such as the recent SCAN functional,²⁹⁶ which promises the accuracy of more computationally expansive approximations, such as hybrid-functional calculations, which effectively mix Hartree-Fock and DFT calculations with similar costs to a GGA functional.

Other recent developments with the design of efficient numerical algorithms have pushed the field toward the direct use of hybrid functionals. Well-known functionals include the HSE06²⁹⁷ and the PBE0 functionals^{293,298} for solid-state quantum material predictions, and the B3LYP^{108,109} functional for molecular quantum systems. Further developments pertinent to predicting quantum materials include dispersion functionals,^{299–302} which can account for dispersion or van der Waals interactions, range-separated functionals,³⁰³ and double hybrid functionals,³⁰⁴ which mix together Hartree–Fock

exchange, a GGA functional, and a many-body perturbation correction.

A recent and prominent example for a class of systems where density-functional theory has been successful but also displayed serious limitations, is in the understanding of the microscopic structure and dynamics of quantum defects in wide-bandgap semiconductor materials or "artificial atom qubits." We show in Figure 8, numerical results obtained using DFT for defect emitters in diamond and monolayer hBN.

In this Thematic Issue, reviews by Zunger and Malyi¹² on "Doping Quantum Materials" and by Kagan and Bassett³⁰⁹ on "Quantum Optical Nanostructures" discuss the recent progress in this field in more detail, and we refer the reader to these excellent reviews. Here we will focus on the computational and algorithmic advances and opportunities in this growing field.

3.2.1. Predicting Artificial Atom Qubits in Solids. Point defects in solids, including defect centers in silicon carbide (SiC),^{310–334} point defects and dopants in 2D transition metal dichalcogenides,^{335–352} nitrogen vacancy (NV)^{353–356} and silicon vacancy (SiV)^{357,358} centers in diamond, or single defects in hexagonal boron nitride (hBN),^{307,359–380} among others,^{381–383} have seen rising attention in recent years.^{384–388} These systems have been identified as promising systems for a wide range of quantum information applications, such as sensing and metrology,³⁸⁹ computing,^{390,391} and use as quantum repeaters for long-range quantum networks.³⁹² The negatively charged nitrogen vacancy center in diamond with a spin triplet ground state is probably the most prominent one out of this list and has been long studied and demonstrated in many applications, including refs 354 and 393–419. The NV center possesses both long-lived spin states and high-fidelity spin–photon interfaces, which can enable quantum networking protocols, and has been theoretically studied, for example, in

refs 420 and 421. Despite these successes, there are also some drawbacks connected to the NV center. In particular, the optical applications of the NV center remain a challenge due to low quantum efficiencies, with weak emission into the zero-phonon line (ZPL). These limitations have spurred a theory-driven search for alternative quantum emitters.

In the diamond host lattice, in addition to the NV center, group IV quantum defects have been recently reported sparking experimental and theoretical studies. Both the NV center and the group IV centers replace one of the carbon atoms by a vacancy and an additional carbon atom by the corresponding element. While in the NV center, the nitrogen atom stays at the position of a carbon site, in group IV elements, the substitutional atom moves between those two carbon sites, as shown in Figure 8a for the PbV defect. The inversion-symmetric split-vacancy structure of the group IV elements results from the defect adopting a D_{3d} point-group symmetry. The inversion symmetry means the defect has no permanent electric dipole moment, which makes their optical transition frequencies less sensitive to electric field noise, which is often found for defects near surfaces.⁴²² In addition, they show more efficient emission into the zero-phonon line than the NV. Recent experimental advances have observed the negative charge state of these defect complexes, that is, SiV⁻, GeV⁻, SnV⁻, and PbV⁻,^{305,423–438} which have also been characterized theoret-ically.^{439–441} The neutral charge state of the SiV has also been observed experimentally.^{358,442–444} One fewer electron than the SiV⁻ gives rise to a spin triplet ground state reminiscent of the NV center. A recent study also explores the excitonic excitations of SiV⁰ and SiV^{-.445} While SiV⁰ requires the doping of the diamond host with boron impurities for thermodynamical stability, more recently, group III-related complexes in diamond have been theoretically proposed³⁰⁶ some of which are predicted to be stable in diamond intrinsically without the need for additional doping, as shown in Figure 8b. Single photon emission has also been explored for defects in 2D materials, in particular for hBN.³⁵⁹⁻³⁶¹ We show a representative example, the V_N defect, in Figure 8c,d, where the defect is the result of a missing nitrogen atom.

From a computational perspective, these systems can be described using density-functional theory on the PBE or HSE06 level, using large supercells to characterize their ground-state behavior. With the inclusion of excited-state properties, photoluminescence spectra can be predicted, 446-450 as shown in Figure 8e for the group III defects in diamond. Descriptions of other prominent interactions, for example, electron-phonon coupling, may be necessary for accurate predictions of groundand excited-state properties. In particular, some of these systems show a prominent Jahn-Teller effect, 308, 420, 421, 439, 451-453 demonstrated in Figure 8f. To include these electron-phonon coupling interactions, these systems have been described by model Hamiltonians parametrized using DFT data. To describe the excited-state manifold, the method of constrained DFT (Δ SCF), which consists of promoting an occupied to an unoccupied orbital, can be used but is limited to weak correlations and low-lying excited states.^{308,439}

The Jahn-Teller interaction is one instance of electronphonon coupling and can be extremely important for accurate predictions of solid state defects and periodic structures, as well as molecules. In the diamond defects mentioned, such as NV and SiV, localized electronic orbitals can be energetically degenerate. When such degenerate orbitals are unequally occupied, there exists the possibility that a symmetry-lowering nuclear distortion can push the system to an overall lower energy state, as first outlined by Jahn and Teller.⁴⁵⁴ The potential energy surface depicted in Figure 8f shows one type of Jahn– Teller landscape, where the coordinate axes represent directions along particular nuclear displacements associated with the Jahn–Teller distortion. The basic idea of this potential energy surface is common for all Jahn–Teller-related phenomena: at the high-symmetry configuration (where $Q_x = Q_y = 0$), the total energy is a local maximum, and a low-symmetry configuration is energetically preferable. There can exist distinct energy minima, as shown in Figure 8f; however in general the potential energy landscape resembles a Mexican hat. The distortion requires coupling of associated vibrational modes and overall the interaction is formulated as a case of electron–vibrational (or electron–phonon) coupling.

The correlated nature of the electronic and nuclear degrees of freedom in Jahn–Teller systems can be especially important for correct predictions of not only energy levels but also observables. As described in ref 455, *dynamic* Jahn–Teller distortions, in which case the system effectively exists in a superposition of the low-symmetry minima, can lead to quenching of observables associated with electronic operators. This is due to the strongly correlated nature of the electronic and nuclear degrees of freedom, which strongly couple to form a hybrid vibrational and electronic ("vibronic") system. Electronic observables can be strongly altered when acting on such vibronic states, for example, the quenching of spin–orbit interaction. Accurate modeling of this Jahn–Teller problem has been crucial to predicting level structures and splittings consistent with experiment,^{439,456} and likewise will be important for predicting properties of novel emitters^{306,308} and molecular qubits going forward.

The Jahn–Teller interaction is one example of a variety of important electron–phonon interactions that can be critical for correct prediction and understanding of correlated quantum materials. In these electron–nuclear correlated systems, the nuclear degrees of freedom cannot be ignored. Phenomena that can emerge from this include things like charge density wave (CDW) distortions, which can result in a symmetry-breaking distortion in periodic systems that can alter the electronic properties.^{457–462} Laser-induced phonon excitation can be used to induce novel electronic properties and phenomena in general (see section 3.4). Electron–phonon interactions can also be crucial for understanding superconductivity in quantum materials.^{463–470} These phenomena can also be modulated by doping, which is covered in greater detail in the corresponding Review by Zunger et al.¹²

Importantly, these phonon-related effects are often temperature dependent. Thus, accurate models and understanding of these phenomena are important for understanding possible finite temperature effects associated with these systems. Theoretically describing these types of interactions is challenging, as at the heart of most quantum chemistry and materials calculations is the Born-Oppenheimer approximation⁴⁷¹ that allows the separation of the electron and nuclear degrees of freedom. In situations where nonadiabatic effects are important, these effects can be included in dynamical and time-dependent cases by explicitly including the nuclei in the calculation, for example, with the Ehrenfest or surface hopping approaches or other methods, such as the exact factorization approach⁴⁷² or conditional wave function approach.⁴⁷³ More generally, a variety of theoretical approaches based on first-principles frameworks to describe electron-nuclear interactions exist, such as generalized

density-functional theories, $^{474-479}$ the nuclear-electron orbital (NEO) DFT, 480 coupled cluster, 481 or density-functional perturbation theory for electron-phonon interactions. $^{468,482-484}$

For systems where strong electronic interactions become important, a different approach is essential. One augmented DFT method to treat more strongly electron-correlated materials within a density-functional theory framework is the LDA+U approach. In its standard formulation, the DFT+U method is an empirical method, in which the effective on-site Hubbard U parameter is chosen prior to the calculation. More recently, this approach has been extended, and it became possible to evaluate the values of U and J ab initio and selfconsistently through linear-response techniques^{485–488} and with the ACBN0 functional.^{489,490} Along similar lines, in recent years the idea to combine the method of DFT and DMFT^{122,491} has seen attention to describe strongly correlated materials, in particular the strong Coulomb repulsion for materials that feature partially filled d or f shells. This hybrid method is based on the assumption that for the specific material it is possible to separate the electronic degrees of freedom into a weakly correlated part that can be treated on the DFT level and a correlated part, which has to be treated with a different method. These two methods are then combined in a self-consistent way. Such a treatment can be expected to be important for systems, where these stronger correlations induce a redistribution of the electronic structure.49

Excited and nonequilibrium states play a fundamental role in quantum materials. The following section will focus on classical approaches that allow access to excited-state properties of quantum materials. Later in this Review, we will discuss approaches for computing relevant excited states of quantum materials on quantum computers.

3.3. Time-Dependent Density-Functional Theory and Quasiparticle Methods for Correlated Quantum Matter

In many cases of experimental relevance, it is necessary not only to obtain ground-state properties of the system of interest but also to access excited-state properties. A straightforward calculation of some of the properties of low-lying excited states, such as ionization energies, can be obtained by the Δ SCF or constrained DFT approaches mentioned in section 3.2. In general, a constrained DFT calculation can be used to calculate the lowest state of a specific symmetry. Therefore for ionization energies, Δ SCF is strictly defined. However, for charge neutral excitations, Δ SCF is only strictly defined for the lowest states of different symmetries. Nevertheless, Δ SCF may also be applied to sets of low-lying excited states by promoting occupied orbitals to unoccupied orbitals. In practice, such a procedure can work well for triplet states that are well described by a single Slater determinant, while it may fail for singlet states that show multiconfigurational character.

A more general density-functional approach to access excitedstate properties is time-dependent density-functional theory (TDDFT).^{493,494} Although TDDFT is in principle valid for any out-of-equilibrium system, in practice TDDFT has been particularly successful in the linear-response regime, from which excitation energies and transition dipole elements for many-body systems can be obtained. The solution of these equations can either be obtained in the time-domain by an explicit time-propagation⁴⁹⁵ or in frequency-domain by solving a pseudoeigenvalue equation.⁴⁹⁶ Practically, to calculate the timedependent properties of real systems, we rely on approximations for the exchange-correlation potential $v_{\rm xc}(\mathbf{r}, t)$, and the validity of the calculation depends on these underlying approximations.⁴⁹⁷

In TDDFT, the simplest but also most widely used approximation is the adiabatic approximation.⁴⁹⁵ In the adiabatic approximation, we assume that the system adiabatically follows the external perturbation and uses functionals that have originally been developed for static or ground-state cases. The adiabatic approximation is justified for certain quantum systems, where the time-dependent density does not change too rapidly. TDDFT is most successful in the linear response regime^{496,498} where the change in the electron density, $\delta n(\mathbf{r}t)$, and the external perturbation, $\delta v(\mathbf{r}t)$, are connected via response functions $\chi_R^{(1)}(\mathbf{r}t, \mathbf{r}'t) = [n(\mathbf{r}t), n(\mathbf{r}'t')]$. In linear-response theory, the adiabatic approximation restricts the calculation to the single excitation manifold and practical calculations can be performed using the eigenvalue (Casida) equation. 496,498 These equations have also found application for artificial atom qubits, for example, in diamond⁴⁹⁹ and hBN.⁵⁰⁰ The latter study compares the accuracy of a variety of different exchange– correlation functionals and quantum chemistry methods for defects with the conclusion that simpler GGA functionals are typically not accurate, but hybrid functionals such as HSE06 can accurately capture the energetics of such states, if the system is a triplet state. To accurately describe the dynamics of strongly correlated systems out-of-equilibrium using TDDFT, recently the ACBN0 functional has been extended to the time domain. For example in ref 501, it has been shown that the absorption spectra of transition metal oxides, such as NiO or MnO, are well reproduced by TDDFT+U simulations. DFT and TD-DFT methods have been shown to be effective at describing out-ofequilibrium plasmonic excitations.⁵⁰²⁻⁵⁰

Beyond linear response, there remain different challenges within TDDFT. A key open problem is extending beyond the adiabatic approximation.⁵¹⁰ Although many studies have been devoted to the analysis and construction of the exact timedependent exchange-correlation functional for one-dimensional⁵¹¹ and real-space systems,^{512,513} a simple, intuitive approximation remains unknown. More general DFT approaches are known to fundamentally underestimate the fundamental band gap, also called the band gap problem of DFT.^{514,515} Recent approaches such as the use of hybrid functionals or generalized Kohn-Sham approaches⁵¹⁵ have been successful in improving the accuracy of the band gap description. An additional challenge for DFT approaches is a correct description of excitons, the bound states of electrons and holes. Some studies have shown that in principle those excitations can be described with TDDFT.^{516,517} Recent work has included the development of stochastic approaches⁵¹⁸ to make TDDFT more computationally efficient.

A more rigorous framework to calculate excited-state properties and obtain a more accurate description of the band gap and excitonic effects of molecular systems is the many-body perturbation approach using the interacting Green's function (GW).^{519–527} The Green's function can be defined for a specific number of particles, for instance, the one-particle and two-particle Green's function are sufficient to extract information about the quasiparticle excitations and the optical responses of an interacting system. In practice, the Green's function can be defined in terms of the electron field operators, which leads to the Martin–Schwinger hierarchy. This can be reformulated formally by introducing the electronic self-energy $\Sigma(1,2)$. In a similar spirit as density-functional methods, the self-energy $\Sigma(1,2)$ can be divided into Hartree and exchange–correlation

parts by using the Hartree potential, $\nu_{\rm H}$, and numerically approximating $\Sigma_{\rm xc}$. In many-body perturbation theory the selfenergy is routinely expanded using Feynman diagrams.⁵²⁸ By formulating the vertex correction to second-order, one finds an equation that can be seen as the linear-response formalism of the self-energy to a change in external potential leading to the socalled Bethe–Salpeter equation (BSE).^{521,529,530} These vertex corrections account for exchange–correlation effects between electrons and also includes electron–hole interactions.

Aside from shifts in predicted transition energies⁵³¹⁻⁵³³ relative to DFT, electron-hole interactions can lead to bound collective excitations known as excitons. These may coalesce into more complex subsystems, such as bound pairs of excitons, biexcitons, and singly charged excitons or trions. The relevance of these collective states depends on the strength of the electron-hole interaction. In many two-dimensional systems, such as transition metal dichalcogenides, the importance of excitonic effects is very prominent, as the exciton binding energies can be on the order of electronvolts and both exciton and trion peaks have been observed experimentally.⁵³⁴ These interactions can be important for looking at behavior of the "bulk" electronic states across the 2D flake as well as localized defect-induced states and their excited-state transitions.⁵³⁵ While excitonic effects are particularly prominent in these 2D monolayers, they can also be important for considering optoelectronic properties in other materials and chemical systems.

Another useful aspect of GW with BSE is access to multireference excited states in quantum matter. One particularly interesting and timely application is in the case of defect states with excited-state singlets. The energetic positioning and interaction with the triplet subspace of these singlet states can be critical for understanding optical efficiency as well as achieving controllable spin-readout^{536,537} of defect states. Due to the inherent multireference nature of these states, methods beyond conventional DFT, such as $GW(+BSE)^{538}$ or TDDFT are required to capture them. Altogether, these unique types of quantum interactions accessible within a GWframework makes it an intriguing method for future study and characterization of a variety of quantum materials. We anticipate many contributions in quantum materials for quantum information to emerge from these methods in future.

With this, recent theoretical and computational developments have made GW-based methods more computationally accessible. Conventional GW calculations require that the screened Coulomb interaction W and the noninteracting Green's function G are determined by a perturbative expansion over Kohn-Sham electronic states. The expansion requires explicit treatment of both occupied and unoccupied states, and often the expansion over unoccupied states yields slow convergence, such that hundreds of unoccupied states may need to be considered for converging a particular system.^{539,540} Recent work that achieves convergence without the need for treating so many excited states, done by solving self-consistent linear Sternheimer equations, has been theoretically demonstrated⁵⁴¹ and imple-mented in available codes.^{542,543} In addition, advances have been made using stochastic methods such that the computational cost of a "single-shot" G_0W_0 calculation scales linearly with the number of electrons $(\sim O(N_e))$. ^{544,545} This is a notable improvement over standard approaches, where the scaling is between $O(N_e^3)$ and $O(N_e^4)$. With this improvement, efficient G_0W_0 calculations have been demonstrated for large systems

consisting of 10⁴ electrons.⁵⁴⁶ The use of stochastic orbitals has also been applied to solving the Bethe–Salpeter equation,⁵⁴⁷ where again explicit calculations have been demonstrated with systems consisting of thousands of electrons.

In addition to bulk-like structures, GW and BSE approaches have also been developed and applied for finite molecular systems, including refs 548-554. Similar to periodic systems, these techniques generally improve excited-state quasiparticle energy predictions compared to conventional DFT and TD-DFT methods and can also reveal the importance of electronhole correlations.⁵⁵⁵ GW-based methods can also be useful in understanding electronic interactions with other perturbations in a given system, such as phonons. The quasiparticle self-energy due to external perturbations can shift the eigenenergy of the state via the real part of Σ and can also contribute to the line width of the electronic state via the imaginary part of Σ . In all practical calculations of these self-energy contributions, a manybody perturbation theory approach is needed to describe the interaction. However, often the interaction itself is evaluated using semilocal, DFT-based techniques, 468,556 which is not strictly translatable to a many-body description or quasiparticle self-energies. On the other hand, GW methods are naturally written in terms of self-energies and include nonlocal electronic effects. Recent work has outlined methods that incorporate phonon interactions into a GW framework in both periodic systems⁵⁵⁷ (denoted as *GWPT*) and molecules.⁵⁵⁸ The periodic case demonstrates that for the correlated superconductor Ba_{1-x}K_xBiO₃, typical DFT-based approaches cannot capture the significant (>50%) enhancement of the electron-phonon interaction strength resulting from many-electron correlations.557

In general, the prediction of novel bulk quantum materials, point defects, or molecules requires methods that give an unbiased physical description of the system. The GW-based techniques can offer a powerful toolkit for accessing the optoelectronic response of these systems, including prediction of band gaps and effects of electron—hole correlations. In particularly correlated systems, the use of GW techniques extended to look at interactions with other perturbations such as phonons or electric fields may be critical, as possible correlations might have important feedback on the associated interaction strengths, which would not be otherwise captured in DFT-based approaches.

3.4. Light-Matter Control of Correlations in Quantum Materials

Driven by new experimental and theoretical advances, recently there has been a push to explore the strong coupling regime of light and matter, that is, a regime where quantum chemical systems and quantum matter are strongly coupled to the electromagnetic field. This regime of strong light-matter coupling opens many new directions by creating novel polaritonic quasiparticles and new states of matter, with promises such as the on-demand control of quantum systems.^{559–561} For a detailed discussion of opportunities in strong light-matter coupling from a photonics and optics perspective, we refer the reader to the reviews in refs 8 and 562–570. Our discussion is centered on theoretical and algorithmic advances in treatment of strong light-quantum matter interactions and opportunities to use such coupling in quantum information science.

Driven by experimental demonstrations in optical cavities⁵⁶² and nanophotonic resonators,⁵⁷¹ the strong coupling regime of

Matter-Photon Interaction

pubs.acs.org/CR

Figure 9. Schematics of prototypical configurations that reach strong light—matter coupling with an optical cavity on the left and a nanogap strongly coupled to nanophotonic excitations on the right. Strong light—matter coupling has be achieved experimentally for single molecules, collective coupling in ensembles of molecules, and two-dimensional and three-dimensional quantum materials as indicated in the middle.

many emitters has been explored, leading to a collective enhancement of the light-matter coupling and strong coupling down to the level of single emitters (Figure 9).⁵⁷² Different experiments have shown that this regime can be realized for a wide range of systems of chemical, physical, or biological interest. Examples of relevant recent studies include the demonstration of changes in chemical reactivity under strong light-matter coupling, such as the chemical reactivity for ground-state reactions⁵⁷³ and excited-state photochemical reactions⁵⁷⁴ and the suppression of photo-oxidation processes.⁵⁷⁵ Other examples besides reactivity examples include the realizations of hybrid organic-inorganic polariton LEDs, 57 single-molecule tautomerization,⁵⁷⁷ the inversion of singlet and triplet excited states,^{578,579} femtosecond transient absorption spectra under ultrastrong coupling,⁵⁸⁰ the deep strong coupling limit in plasmonic nanoparticle crystals,⁵⁸¹ and intermolecular vibrational energy transfer.582

Chemical Reviews

Recent theoretical studies in this field include the study of the effects of anharmonicity of vibrational modes on strong lightmatter coupling,⁵⁸³ effects beyond the dipole approximation,⁵⁸⁴ surface-enhanced Raman spectroscopy,^{585,586} or the origin of asymmetric emission.⁵⁸⁷ Other effects studied are ensembleinduced strong light-matter coupling of single emitters,⁵⁸⁸ polariton-assisted remote energy transfer,⁵⁸⁹ electron transfer,⁵⁹⁰ or excitation transfer,⁵⁹¹ among others. The broad spectrum of effects seen in such light-matter control of quantum systems necessitates novel theoretical and computational approaches.

Previously many of these experimental results were described by using effective Hamiltonians inspired by work in the established quantum optics community. More recently, the field has seen a push toward the use of *ab initio* methods in quantum electrodynamics to describe electron—photon interactions. While traditional *ab initio* methods, such as those described in the prior sections, are capable of accurately describing the electronic structure, the electromagnetic field is typically not accounted for. In cases of strong light-matter coupling, the treatments of electronic and photonic degrees of freedom become equally important, making it essential to go beyond conventional electronic *ab initio* methods.

Review

The nonrelativistic electronic Hamiltonian can be formulated to account for transverse electron–photon interactions by changing to the length gauge and in dipole approximation as follows: ^{592,593}

$$\hat{H} = \hat{H}_{0} + \frac{1}{2} \sum_{\alpha} \hat{p}_{\alpha}^{2} + \frac{1}{2} \sum_{\alpha} (\omega_{\alpha} \hat{q}_{\alpha} - \lambda_{\alpha} \cdot \mathbf{R})^{2} + \frac{j_{\text{ext}}^{(\alpha)}(t)}{\omega_{\alpha}} \hat{q}_{\alpha}$$
(21)

where \hat{H}_0 represents the electronic Hamiltonian and **R** represents the electronic dipole operator with $\mathbf{R} = \int d\mathbf{r} \ n(\mathbf{r})\mathbf{r}$. In the length gauge, the conjugated variable to the magnetic field is the electric displacement field that is given by $\hat{\mathbf{D}} = \sum_{\alpha} \sqrt{4} \pi \omega_{\alpha} \hat{q}_{\alpha}$. The electronic displacement field is coupled here in the dipole approximation where the bosonic operators $\hat{q}_{\alpha} = \sqrt{\frac{1}{2\omega_{\alpha}}} (\hat{a}^{\dagger} + \hat{a})$ and $\hat{p}_{\alpha} = i \sqrt{\frac{\omega_{\alpha}}{2}} (\hat{a}^{\dagger} - \hat{a})$ are connected to the electric field at the center of charge by $\hat{\mathbf{E}} = \sum_{\alpha} \sqrt{4\pi} (\omega_{\alpha} \hat{q}_{\alpha} - \lambda_{\alpha} \cdot \mathbf{R})$. The operators \hat{a}^{\dagger} (\hat{a}) create (destroy) one photon in mode α .

Starting from the Hamiltonian in eq 21, one can define a density-functional theory for electron—photon coupled problems as follows: Analogous to the electronic problem, a basic set of variables has to be chosen. In this case, these are the electron density, $n(\mathbf{r})$, and the electric displacement field, q_{α} . These basic variables then can be used to formulate a one-to-one correspondence between the basic (internal) variables and the



Figure 10. (a) Experimental setup for SrTiO₃ coherently excited with tunable wavelength pulses. Adapted from ref 625. Reprinted with permission from AAAS. (b) Dynamical ferroelectricity in SrTiO₃ phase diagram. While bulk SrTiO₃ is paraelectric, small amounts of strain can induce a ferroelectric transition. Alternatively, this transition can also be induced dynamically through vibrational excitation. Adapted from ref 625. Reprinted with permission from AAAS. (c) Cavity-mediated resonance in nonlinear phononics with the schematic setup of a sample material in a terahertz cavity. Adapted with permission from ref 626. (d) (left) Frequency mismatch between an IR-active and two Raman-active phonon modes hinders the transfer of energy between them and (right) formation of two phonon–polariton modes with frequencies Ω_+ and Ω_- due to the hybridization between an IR-active phonon mode leads to more efficient excitation transfer.⁶²⁶ Adapted with permission from ref 626. (e) Light–matter phase diagram as a function of the electronic interaction strength *U* and the light–matter coupling strength *g* with the novel superradiant excitonic insulator phase.⁶²⁷ Reprinted with permission from ref 627. Copyright 2019 by the American Physical Society.

external variables, $v_{\text{ext}}(\mathbf{r})$ and $j_{\text{ext}}^{(a)}$, that allows for the reformulation of expectation values as functionals of the internal variables. This density-functional theory is also called quantum-electrodynamical density-functional theory (QEDFT)^{592–595} and has been applied to electronic strong coupling in molecular systems for ground-state^{596–598} and excited-state problems,^{599,600} as well as vibrational strong coupling⁹ and cavity losses.⁶⁰¹

Following this work in QEDFT, other electronic firstprinciples methods were generalized to account for electronphoton interactions. Key examples include the Maxwell-Hartree–Fock wave function approach,⁶⁰² the multiconfigura-tion time-dependent Hartree method,⁶⁰³ the polaritonic coupled cluster approach,^{604,605} the exact factorization approach606,607 for electron-photon systems, a conditional approach,⁶⁰⁸ extensions of the Born–Oppenheimer approx-imation,^{609–611} and efficient Maxwell–Schrödinger propagation schemes.^{612,613} First-principles descriptions of electron–photon problems are still nascent. We expect that in the next few years, with joint experimental and theoretical advances, more accurate ab initio models will explore the realm of strong light-matter interactions for quantum chemical systems. There are still many open questions in this field, including what is the role of a realistic description of the optical cavity, including its imperfections, in an *ab initio* description, and questions about the precise change of the transition state in vibrational strong coupling experiments, the effect of solvents in these experiments, and effects beyond the dipole approximation.

New avenues to explore light-matter coupling can be expected if the rich possibilities of different electromagnetic environments are considered. More specifically, can we achieve strong coupling for regimes that are systematically different from a more simple one-mode coupling as is the case for optical cavities? These ideas can be explored by using the framework of macroscopic QED,⁶¹⁴⁻⁶¹⁶ which provides a consistent quantiza-

tion procedure for absorbing and dispersive media and allows at the same time for an accurate description of systems and their response if they are embedded in complex photonic structures. Using this scheme, the fundamental material variables are the microscopic charge currents in the medium and the quantized field operators, both of which have the correct commutation relations leading to Maxwell's equations in the classical limit. In this description, the losses of the system are already intrinsically included in the field operators and do not have to be introduced as parameters. The same is true for electron—photon coupling strength and the broadening and line widths of electromagnetic modes and excitations.

Recent theoretical proposals^{617,618} and experimental demonstrations⁶¹⁹ have shown that strong light—matter coupling can be used to access ionizing transitions. Typically, transitions above the ionization threshold strongly hybridize with the continuum, leading to Fano line shapes. Strong light—matter coupling can now be used to hybridize with this transition, leading to a lower polariton state below the ionization threshold, effectively decoupling it from the continuum. In this way, new states can be made accessible. It will remain to be seen how those states can be explored for chemical applications such as novel reaction pathways in photochemistry.

One of the key open questions concerns the limit of lightmatter coupling strength. Recent experimental progress has demonstrated the so-called ultra and deep strong coupling limit^{581,620} for plasmonic systems, going far beyond the early demonstrations of this limit in circuit QED,⁶²¹ thus pushing the experimental limit of accessible systems and consequently their theoretical description. Since changes in the electronic ground state have been shown to be rather small in regular strong coupling,⁶²² ultra and deep strong coupling are a very different case due to the scaling with coupling strength. In particular, probing these experimental regimes would benefit from a firstprinciples description.⁵⁹⁷ Recent proposals have used coupled systems with high transition dipole moments⁶²³ to couple directly to excitations with weak transition dipole moments. It has been shown that single-molecule strong coupling can be achieved through coupling to an additional ensemble,⁵⁸⁸ which motivates engineering of the environment to effectively enhance the light–matter coupling.

Besides the use of optical cavities to control material properties, we can also expect novel applications for strong light-matter coupling in emerging quantum information science applications⁶²⁴ and in probing quantum matter. Quantum sensing of quantum matter in regimes of ultrastrong coupling is an exciting new area. There are two closely related concepts here: harnessing ultrastrong coupling (enabled by optical or superconducting circuitry) for the detection of intrinsic quantum material properties and correlations, and creating new states of quantum matter prepared via ultrastrong coupling to the cavity.

Next we turn our attention to the manipulation of quantum materials using strong light-matter interactions. This is a particularly attractive direction in engineering quantum matter for quantum technologies, as it enables the control at fundamental time scales and access to novel nonequilibrium states of matter.^{559,628,629} Ultrafast optical methods^{505,508} have been recently used as a new tuning knob to induce insulator-tometal transitions,⁶³⁰ topological phases,^{631,632} ferroelectricity,^{625,633} and transient superconductivity in copper oxides⁶³⁴ and organic crystals.⁶³⁵ While this provides a tantalizing opportunity to explore new quantum phases, particularly in the case of transient superconductivity, key questions remain on the underlying excitation mechanisms, as well as on how to optimize and stabilize such short-lived electronic phases. Many of these studies explore the nonlinear interactions between phonon modes that govern the behavior of vibrationally highly excited quantum matter. Two of the most prominent examples of nonlinear phonon induced states include superconductivity far above the equilibrium critical temperature^{629,636} and lightinduced ferroelectricity and ferroelectric switching.^{625,633} One example of how optical cavities can be used to control the redistribution of energy from a highly excited coherent infraredactive phonon state into the other vibrational degrees of freedom using nonlinear phononic interactions has been proposed theoretically in ref 626. In this work, the hybridization of the infrared-active phonon mode with the fundamental mode of the cavity induces a polaritonic splitting that can be used to tune the nonlinear interactions with other vibrational modes in and out of resonance, as demonstrated in Figure 10c. In a typical material, the IR-active phonon mode with frequency $\Omega_{\rm IR}$ is out of resonance with two Raman-active phonon modes with frequencies $\Omega_{\rm R}$, preventing efficient coupling between them. By splitting the IR-active phonon mode in a terahertz-frequency cavity, one of the polariton branches is shifted into resonance with the Raman-active phonon modes, enabling efficient energy redistribution. This work broadens the range of materials in which resonant nonlinear phononic processes can be exploited to yield nonequilibrium states of matter. Concepts in cavity control of nonlinear processes enable a new pathway for quantum optical engineering of new states of matter. The analysis presented here is applicable to resonant coupling mechanisms between IR-active phonon modes and other fundamental excitations in solids and molecules.

One of the hallmarks of quantum materials is that their electronic ground state can be extremely susceptible to small perturbations. This susceptibility presents an opportunity to engineer these quantum materials, for example, by small tweaks to their chemical composition, lattice structure, or symmetry, leading to a dramatically altered ground state and uncovering interfacially correlated quantum effects with dramatic changes to conductivity, superconductivity, and magnetism.

3.4.1. New Directions in QED Control of Correlated Quantum Materials. Research in this direction has been motivated by theoretical proposals and experimental results that suggest that placing materials in a cavity or other environment with a strong dielectric response can also substantially affect the many-body physics. This includes exciton-polariton condensates⁶³⁷ or superfluidity⁶³⁸ formed by hybrid quasiparticles composed of photons and excitons in semiconductors,639 driving a material through an excitonic insulator transi-tion,^{627,640} polaritons in excitonic insulators,⁶⁴¹ the manipu-lation of the ferroelectric phase-transitions,⁶⁴² and nonlinear phononics.⁶²⁶ Recently, various studies have suggested an influence of strong light-matter coupling with different mechanisms on superconductivity.⁶⁴³⁻⁶⁴⁷ It has also been studied that an unconventional dielectric environment can enhance the transition temperature of superconductivity. The underlying idea of cavity-enhanced superconductivity is that by embedding a material in an appropriate dielectric environment, key modes may be changed in a way that is advantageous for stabilizing the superconducting phase. Further, the cavity modes can couple to carriers, serving as a new "boson pairing glue" and possibly leading to new kinds of superconducting states.⁶⁴⁵ This so-called cavity-mediated superconductivity is particularly appealing in nonequilibrium regimes, which may enable the field to attain nonthermal distributions. However, cavityactivated phenomena in superconductors remains an experimentally largely unexplored field with only very recent first experimental realizations.⁶²³ Inspired by recent developments on nonlinear phononics in optical cavities,⁶²⁶ other possible examples for light-matter altered effects can be expected not only in chemical reactivity, but also in other molecular relaxation mechanisms, such as intramolecular vibrational energy redistribution (IVR)⁶⁴⁸ in ground-state potential energy surfaces. Similar effects have been shown for excited-state relaxation processes⁶⁴⁹ and anharmonicities.⁵⁸³

3.5. Emerging Directions in Descriptions of Correlated Quantum Matter

Much like other areas in the physical sciences, ideas from machine learning and data science, in particular neural networks, are seeing more applications in capturing correlations in quantum chemistry and quantum material science. Ideas that have been already successfully realized include using neural networks as a representation for quantum states in electronic structure calculations⁶⁵⁰ for problems such as the Hubbard and Heisenberg models, and now even in chemistry for small molecules reaching chemical accuracy.⁶⁵¹ Molecular-orbital-based machine learning methods have also been applied to accurately predict post-Hartree–Fock energies in molecular systems.^{652–654} Along similar lines, deep neural networks^{655,656} have been explored as representations for quantum wave functions.^{652–654}

4. QUANTUM ALGORITHMS FOR QUANTUM MATTER

The focus in both wave function and density based electronic structure has been on reducing the exponential scaling of the *N*-body problem. While classical electronic structure theory has

many methods to address this problem as discussed in sections 2 and 3, all of these methods face the challenge of describing the intrinsically quantum mechanical nature of the problem using a classical representation, which can become impractical for highly entangled or correlated states. Because of this, many open problems are still far beyond the current classical computing capabilities. An alternative pathway toward a solution for quantum systems is the use of quantum devices, which leverage their inherently quantum-mechanical nature for computation.

The first quantum algorithms suited particularly for quantum system predictions appeared as early as the late 1990s with various attempts at simulating fermionic Hamiltonians. Recent developments and improvements of functional quantum hardware devices have re-energized the field, leading to both research and commercial interests in exploiting the inherent quantum properties of the machinery to achieve Feynman's original vision dating back to 1982.658 There has been much anticipation over the potential for quantum computers to be able to solve classically intractable chemical, physical, or combinatorial problems, increasingly so since the recent demonstration of quantum advantage using a superconducting device by colleagues at Google and NASA.¹ As they highlight and demonstrate,³⁻⁵ a co-design approach to the hardware and algorithms is critical to utilizing the advantages offered by smallscale noisy quantum devices. We note that there has also been compelling progress in cold-atom and ultracold Rydberg quantum "emulators"⁶⁵⁹ that are beyond the scope of this Review.

There is a considerable push in the quantum computing community to achieve algorithms with nonexponential complexity for strongly correlated systems with applications in excitonic quantum matter, many-body quantum states, large-scale entangled states, and high-temperature superconductivity. Each of these has remained a "holy grail" problem in condensed matter physics and quantum chemistry. Remarkably, despite decades of intense effort, we still lack a full theoretical understanding of the workings of high-temperature superconductors, thereby precluding rapid progress in raising the critical temperature from less than half of room temperature⁶⁶⁰ under ambient pressure. Notable recent developments under high pressure have demonstrated critical temperature up to 250 K.⁶⁶¹

This search for a theoretical description and an understanding of the phase diagram has animated large-scale research in quantum simulations of the Hubbard (or Fermi-Hubbard) model, presenting a high-reward application of quantum algorithms in correlated quantum matter. Tractable and accurate quantum algorithms to address these applications would be transformative not only for the quantum materials community but across the physical sciences.

Practical applications on current quantum devices are challenged by the susceptibility of the devices to various noise mechanisms, effectively limiting the number of operations that can be performed. To this end, research has been focused on hybrid quantum—classical algorithms that combine classical resources, which are much more established and error-tolerant, with quantum computation. For chemical systems, the resource estimates to use quantum computers to elucidate the reaction mechanism in complex chemical systems have been suggested to be within reach of near-term quantum devices.⁶⁶²

Recently, there have been several excellent reviews on the topic of quantum algorithms for quantum chemistry and materials science applications, including refs 663–666, to

which we refer the interested reader for an in-depth introduction to quantum algorithms from various perspectives. In section 4.1, we primarily focus on introducing the concepts of quantum computation for applications in quantum matter simulations. We then give an overview of recent progress on algorithm development for modeling physical systems using quantum and hybrid classical—quantum hardware suitable for the NISQ area in section 4.2. In section 4.3, we present a brief outlook on the field, including error mitigation schemes and protocols for scalable quantum networks. We recognize that this field is vast and fast-paced with important papers appearing daily; our Review captures key contributions to the field, constrained by our original motivation to highlight the potential of new quantum algorithms to discover entirely unexplored regimes of quantum matter.

4.1. Early Algorithms and Quantum Computation of Matter

Early work on quantum algorithms such as Grover's⁶⁶⁷ and Shor's⁶⁶⁸ algorithms brought to the forefront the potential for computational speed-ups due to quantum resources. A few years later the concept of quantum speed-ups was transferred over to physical and chemical systems such as approximation of partition functions for Ising spin glasses,⁶⁶⁹ calculation of thermal rate constants,⁶⁷⁰ and simulation of fermionic systems.⁶⁷¹ Around the same time, the quantum phase estimation algorithms (QPEAs) emerged.⁶⁷² However, these original algorithms faced the challenge of requiring very deep circuits. To successfully perform the necessary operations, error mitigation schemes on current quantum devices that are inherently noisy (NISQ devices) are insufficient, and faulttolerant quantum computers would be essential.^{664,673} Despite these challenges, the advent of the QPEA triggered a plethora of algorithms applied to chemical and physical systems,⁶ including an algorithm for simulating many-body Fermi systems,⁶⁷⁵ followed by the first algorithm that allowed for the calculation of eigenvalues and eigenvectors of a local Hamiltonian.⁶⁷⁶ This novel approach allowed for the first calculation of static electronic structure properties of a system on a quantum simulator.⁶⁷⁶

An impressive push forward for the field of electronic structure was presented in 2005 when Aspuru-Guzik et al. proposed and validated an algorithm for calculating molecular energies on a quantum device.⁶⁷⁷ Using the FCI Hamiltonian on the quantum device and using the Hartree-Fock wave function as a reference, correlation effects were taken into account for small molecules.⁶⁷⁷ This algorithm proved to require only polynomial scaling in terms of system size, suggesting great potential for quantum speed up while treating quantum chemical systems. Further work was done by replacing the Hartree-Fock wave function with a wave function from MC-SCF to obtain an energy spectrum for molecular systems that included excited-state energies.⁶⁷⁸ These early quantum algorithms showed great potential toward more efficient modeling of electronic structure. However, their requirements of many qubits and large gate depth renders them impractical on existing NISQ devices. This obstacle inspired investigation toward hybrid quantum-classical algorithms, which can be computationally efficient by dividing the work between the robust classical devices and more noise-susceptible quantum devices. The theory, advances, and application of these algorithms are discussed in the next section.



Figure 11. Generic hybrid quantum-classical algorithm where classical resources are used for optimization and quantum resources for state preparation and measurement. As the algorithm iterates, the molecular energy is decreased and the predicted molecular orbital density becomes more accurate.

4.2. Hybrid Quantum-Classical Algorithms

Currently, the majority of quantum algorithms relevant to physics and chemistry applications are hybrid algorithms. In these methods, part of the computation is performed on a classical device, typically a parameter optimization step, while the remaining part is done on a quantum device. Both parts of the calculation can interact or are iterated, when necessary. A general schematic of a generic hybrid algorithm is shown in Figure 11. In the figure, the rail diagram in the center shows a calculation using a classical device (teal squares), which generally corresponds to a classical initialization of the parameters of interest. Information is then passed to a quantum device (purple circles), which performs an additional calculation and is then measured. The classical device can read in the outcome of the quantum measurement and in the majority of cases, many such cycles are performed iteratively until a convergence threshold is obtained. Above the rail diagram, we show an energy diagram to demonstrate the iterative procedure, here the decrease of the system energy during the calculation for an arbitrary quantum mechanical system, while on the bottom of the figure an iterative improvement on molecular orbitals can be seen.

4.2.1. General Concepts of the Variational Quantum Eigensolver Algorithm. Early work in hybrid algorithms includes many-body simulation algorithms⁶⁷⁹ and calculations on the Hubbard model.⁶⁸⁰ More recently, the most widely used hybrid algorithm has been the variational quantum eigensolver (VQE), which is designed to variationally minimize the expectation value of the full molecular Hamiltonian based on the variational principle. This minimization strategy leads to solutions of eigenvalue problems by using classical resources for parameter optimization in an outer loop and quantum resources for quantum state preparation and measurement in an inner loop.^{681–684} Within this scheme, the VQE algorithm requires the definition of the system Hamiltonian and the specific parameters that can be optimized. One way to define these parameters is based on the classical unitary coupled cluster

(UCC) method^{102,103,685,686} discussed in section 2.2. The number of terms in the many-body Hamiltonian scales polynomially, and the number of possible excitations used to define the UCC wave function also scales polynomially. Since the Baker-Campbell-Hausdorff series does not terminate^{102,103} for UCC, its complexity is exponential on a classical device, and no general efficient implementation of UCC is known on classical computers. In contrast, this state can be prepared and measured efficiently on a quantum device. After initial state preparation and measurement of the expectation value on the quantum device for a specific initial guess, the classical computer can then be used via an optimization algorithm to provide a new set of parameters leading to a lower total energy. This procedure is iterated until a convergence criterion is met, resulting in the lowest eigenvalue of the Hamiltonian.^{681,682}

In practice, to map the electronic operators onto qubits, fermion-to-spin mappings,⁶⁸⁷ such as the Jordan–Wigner transformation,⁶⁷¹ parity transform,⁶⁸⁸ or Bravyi–Kitaev⁶⁸⁹ transformations, can be used. Other mappings, such as those based on ternary trees have been put forward recently.⁶⁹⁰ Once the electronic operators are mapped onto qubits, the UCC state can be constructed. To construct this state on the quantum device, the Trotter expansion is used,⁶⁹¹

$$e^{A+B} = \lim_{n \to \infty} (e^{A/n} e^{B/n})^n$$
(22)

where *A* and *B* are one- or two- particle operators contributing to the UCC operator. This expression is only exact if the operators *A* and *B* commute or in the limit of $n \rightarrow \infty$, and it therefore in general requires truncation. In practice it has been shown that even truncation to first order can often be sufficient to obtain accurate results, since the variational nature of the algorithm can mask the truncation error.^{692,693} However, in general first order Trotterization still requires a high gate complexity, which is one of the drawbacks of this approach. Multiple studies have been dedicated to investigating other VQE ansätze and generalpubs.acs.org/CR



Figure 12. (a) Schematic depiction of the ADAPT-VQE algorithm where a collection of operators are defined in an "operator pool" to be used to construct the ansatz. The wave function is initialized, operators are selected from the pool, a trial state is prepared, and then the gradient is measured. If the convergence criteria is not met then the operator with the largest gradient is added to the ansatz for a new VQE calculation and the process is repeated. (b) Potential energy as a function of nuclear coordinate of BeH₂ in units of hartrees. (c) Error between the ADAPT-VQE energy and the FCI energy in kcal/mol for BeH₂. (d) Number of variational parameters required for the ansatz used in the energy calculation in panel b. Reproduced with permission from ref 693. Copyright 2019 Springer Nature.

izations,^{686,694} such as a VQE ansatz based on the particle preserving exchange gate to achieve excitations⁶⁹⁵ and constrained VQE schemes.⁶⁹⁶

In comparison with the early quantum algorithms discussed in section 4.1, a strength of the VQE algorithms lies in their use of shorter circuits with more measurements,⁶⁹³ allowing for adaptability to a wide variety of quantum hardware. Since its inception on photonic qubits, VQE has also been used on superconducting qubit^{683,697} and trapped ion devices.^{698,699}

Early experimental work on superconducting (SC) NISQ devices at IBM Research successfully produced potential energy surfaces using the VQE ansatz for small molecules, including H_2 and BeH₂.⁶⁸³ This was a major achievement for the field at the time; however, imperfections in the potential energy surfaces arose from the quantum calculations. While experimental setups and hardware can be responsible for some of these imperfections, this study inspired further theoretical method development to improve upon these methods. Many studies have focused on extending the VQE to increase accuracy, including state and ansatz preparation improvements, ^{693,700,701} which will be the focus of the next section.

4.2.2. Extensions of VQE. A recent adaption of the VQE scheme has used a transcorrelated Hamiltonian that is connected to the original Hamiltonian by a similarity transformation, showing increased accuracy without the need for extra quantum resources.⁷⁰² Another such study aimed to increase the accuracy specifically for strongly correlated systems by allowing the system to determine a quasi-optimal ansatz. This method is referred to as the Adaptive Derivative-Assembled Pseudo-Trotter variation, or the ADAPT-VQE, and is outlined in Figure 12a.⁶⁹³ In this schematic, the first step is performed classically and therefore omitted from the diagram. However, it is the standard first step for VQE and involves computing the 1and 2-electron integrals then transforming the fermionic Hamiltonian into a qubit representation. The ADAPT-VQE was demonstrated on several molecular examples, one being the BeH₂ molecule. The potential energy of dissociation, the error as compared to FCI, and the number of variational parameters required for the ansatz are shown in Figure 12b,c,d respectively. To date this method has been tested on a simulator and has shown substantial improvement toward the accuracy of the VQE algorithm for molecular dissociation energies.

VQE has also found applications beyond the established chemical systems, including strongly correlated quantum matter systems. In such systems, hybrid quantum–classical DMFT and DMET algorithms suitable for NISQ devices have been proposed and realized by using VQE as the impurity solver.^{703–706} Similar ideas have been applied using the constrained random-phase approximation approach⁷⁰² to investigate color centers with defects in wide-band gap materials, also discussed in section 3.2. While research continues to be dedicated to the improvement of the VQE method in terms of accuracy and efficiency, there also exist many extensions of the VQE algorithm to broaden the scope of applicability to different quantum system regimes including excited states, spin states, and energy derivatives.

Excited states play a fundamental role in quantum chemistry and condensed matter physics, including in predictions of reaction rates and optical spectra. Many extensions of the VQE algorithm exist to treat excited-state phenom-ena.^{681,682,697,707-709} One of the first such techniques was the folded spectrum method, which was proposed in the original VQE work some time ago.⁶⁸¹ While this method has the ability to successfully capture excited-state properties, it requires a quadratic increase in the number of terms in the effective Hamiltonian and consequently, a significant increase in the number of required measurements.⁷¹⁰ There are Lagrangianbased methods that require variationally minimizing the Lagrangian instead of the original Hamiltonian⁶⁸² to obtain an approximation for the excited state. The quantum subspace expansion (QSE) is yet another approach that has been popular in the last few years and is based on linear response theory.⁶ Essentially, after obtaining a wave function from a quantum channel (defined by a set of Kraus operators), a subspace of lowlevel excited states is approximated. The key drawback of the quantum subspace expansion is that the quality of the excited states (and corresponding spectra) obtained depends heavily on errors introduced by the linear-response expansions and on the chosen ansatz. However, an advantage of the QSE method is that it can be used as an error mitigation scheme, a topic that will be discussed in detail in section 4.3. Yet another excited-state extension of VQE is referred to as the witness-assisted variational eigenspectra solver,⁷¹⁰ which augments the objective function to include the energy and an approximation for the entropy. A control ancilla qubit is considered along with the trial state, with the control qubit acting as an "eigenstate witness," where its entropy measurement nears zero if the optimized trial state is arbitrarily close to an eigenstate of the Hamiltonian. A tunable parameter is used to bias toward excited states; this parameter is set to zero for the ground state as we might expect and then is tuned for each successive iteration such that the resulting states correspond to approximate excited states of the systems. Using either an iterative phase estimation algorithm or a Hamiltonian averaging approach,⁷⁰⁷ the corresponding energies can be extracted. There are certainly other methods out there and this is a rapidly developing area at the vibrant intersection of quantum chemistry and quantum information science.

There has also been relevant work in extensions of VQE to be able to capture electronic properties of ions and different spin states as well as excitation spectra.^{696,707,708} One such example, referred to as the multistate contracted VQE, was able to successfully capture the transition energies and oscillator strengths in an 18-chromophore light-harvesting complex.⁷¹¹ Another interesting extension to the VQE algorithm is for the treatment of nonequilibrium steady states,⁷¹² which has the potential to lead to the treatment of open quantum systems using quantum resources.

While energies are important properties for any quantum chemical or material system, the derivatives and gradients of these energies also provide valuable physical insight including optimal molecular geometries, partition functions, and vibrational frequencies.⁷¹³ By derivation of analytical formalisms and quantum circuits, the VQE algorithm has been extended to measure energy derivatives.^{714,715} A recent study has investigated thermodynamic properties of molecules by using VQE on a NISQ device to construct potential-energy surfaces.⁷ These surfaces are used to construct a Morse potential to mitigate noise and provide vibrational energy levels on a classical computer. Using these parameters and the potential, the partition function can be computed as a function of temperature leading to thermodynamic observables of the molecule. These extensions broaden the scope of VQE by allowing access to these quantities more directly, which are critical to both electronic structure and chemical reactivity.⁷¹⁵ Moreover, these parameters can be obtained using similar quantum architecture and circuits as those used in the original variational calculation.⁷¹⁴

As previously mentioned a strength of the VQE framework is in its adaptability to a wide variety of quantum hardware. For successful adaptation across different platforms, optimization of an algorithm is crucial since parameters such as the number of qubits and circuit depth is currently limited. Moreover, the circuit structure and gate set availability need to be taken into account when implementing these algorithms on real devices. For variational quantum algorithms, quantum circuit structures often break the symmetry of the Hamiltonian.⁷¹⁷ To overcome this challenge, an adapted-variational scheme was proposed to restore the spatial symmetry through postprocessing classical application of a projection operator.⁷¹⁷ This algorithm showed increased fidelity of the ground state while adding versatility to the hardware by using the same ground-state circuit structure to approximate low-lying excited states.⁷¹⁷

One of the biggest challenges that quantum algorithms face is that increasing system size is generally associated with an increase in the required number of qubits and the circuit depth. Inspired by classical active space methods as discussed in section 2, a recent study approximated the core and virtual orbitals using a classical computer while utilizing a quantum device for the active space calculations.⁷¹⁸ Strategically chosen additional measurements make up for the accuracy lost due to the active space approximation. Through this method, the accuracy of a 20-qubit representation was matched with a 4-qubit quantum computer for the hydrogen atom.⁷¹⁸

When developing novel algorithms, validation and verification is important in part due to the high error rates and initially unknown nature of errors of NISQ hardware. This has led to interesting research questions in how to systematically benchmark results of quantum computations. Toward this goal, a twoelectron ansatz was developed that uses the *N*-representability constraints of a two-electron system, discussed in section 3.1 to efficiently partition local and nonlocal degrees of freedom on the classical and quantum computer for a basis set of arbitrary size. The ansatz can be used for benchmarking small molecular systems and was used to evaluate 4- and 6-qubit simulations of H_2 and H_3^+ , respectively, on superconducting quantum devices.⁷¹⁹ The ease of classical simulation for the electron pair and related pair theories could serve as verifiable targets for benchmarking molecular simulations as quantum devices looking

continue to expand.719 4.2.3. Beyond VQE: Other Hybrid Quantum–Classical Algorithms. A major drawback of the VQE algorithm is that it often requires high-dimensional optimization over a nonoptimal surface thus requiring rapidly increasing computational cost with increasing system size. While improvements continue to be made to the VOE algorithm to overcome this obstacle, a variety of alternative algorithms are emerging. One recently presented alternative approach is based on the classical anti-Hermitian contracted Schrödinger equation (ACSE), mentioned briefly in section 3.1. The ACSE is a contraction of the Schrödinger equation onto the 2-particle space and has been successfully used to capture strong correlation in atoms and molecules.¹⁸⁷⁻¹⁹⁷ Classically, this method is limited because it requires reconstruction of the 3-RDM; however, in the quantum analogue this dependency can be eliminated through the preparation of an appropriate state.⁷²⁰ This novel study introduces a quantum eigenvalue solver which solves a contracted eigenvalue equation for efficient, scalable molecular simulations on quantum computers that does not rely on derivative-free optimization. This method is benchmarked on IBM's quantum devices and a quantum simulator with the ground-state dissociations of H₂ and H₃, respectively. Due to its potentially short circuit depths and exponential speed-up over its classical counterpart, the quantum ACSE algorithm shows great promise for capturing strong correlation in molecules using quantum devices.⁷²⁰

The search for near-term quantum advantage has been highly focused on variational algorithms, one of the most promising being the quantum approximate optimization algorithm (QAOA). This algorithm leverages quantum computers to solve combinatorial optimization problems by encoding a cost function as an operator in the Hilbert space. QAOA uses ideas similar to VQE for classical optimization problems and has been shown to work for a variety of optimization problems, especially those in the NP-Complete regime such as the Max-Cut problem, or its extensions. The Max-Cut problem involves grouping of nodes in a graph into two subgroups by cutting their edges. These cuts are optimized in such a way that the added weights of the edges are maximized. This problem has been shown to be NP-complete and has many applications in network science and in statistical physics.⁷²¹ Some avenues of potential application in correlated matter include spin models, molecular structure problems, and other optimization frameworks relevant in condensed matter physics. The first quantum algorithm to tackle the Max-Cut problem was the QAOA algorithm.⁷²² Since the Max-Cut problem can be mapped onto a classical Ising Hamiltonian, it can also be solved on a quantum device similarly to the VQE method.^{723,724} Research on these quantum problems has in turn inspired developments of improved classical algorithms by exploiting parallelization.⁷²⁵ The variational nature of this algorithm means it includes generalized parameters that must be trained to match the problem instance, so repeated calls to a quantum device are necessary to traverse the parameter space and find global optima. Additionally, it includes an iteration variable that decides how many full applications of the operator are to be executed. This iteration variable increases with the size of the constraint problem, and thus the length of the quantum circuit does as well. Largely, research in this area has focused on a small number of iterations, mainly one or two due to the ability to find exact best parameters. Recent developments have been made when

looking at higher numbers of iterations and on how best to train parameters at given iteration levels and quantification of results versus other methods. One strategy for parameter training that shows promise is doing most or all training on a scaled-down version of the full problem in an attempt to save quantum resources while maintaining high-fidelity results.

In contrast to classical problems, defining a quantum advantage for problems in correlated quantum matter is not as straightforward and remains an open question. Many of the classical optimization problems can be mapped onto the Isingspin Hamiltonian, which is diagonal in the underlying computational basis. One advantage that a quantum algorithm may offer over classical algorithms is the potential to avoid and escape local minima. However, we note that a quantum advantage for these problems is under heavy debate in the community.

While the primary focus of this Review is on correlated systems, since the ideas of VQE are general, that is, to solve an eigenvalue problem, the principles and similar ideas have found applications beyond electronic structure theory. A few examples include quantum algorithms solving sets of linear equations such as of the form Ax = b. It has been shown that quantum algorithms can solve such problems exponentially faster than classical algorithms.^{726,727} These ideas have also been extended to problems in data science with the least-squares fit⁷²⁸ and also extended to nonlinear partial differential equations.⁷²⁹

Besides the developments on universal quantum computers, there have been various efforts to realize adiabatic quantum computing,⁷³⁰ with realizations up to 1800 qubits based on annealing-based quantum processors. These devices can also be useful for quantum chemistry applications⁷³¹ in particular if other mapping schemes are used.⁷³² So far practical simulations of chemical systems remain limited for these devices.

4.2.4. Quantum Convolutional Neural Networks for Quantum Phase Classification. The emergence of NISQ devices has co-occurred with increased interest and progress in classical artificial intelligence and machine learning techniques. Neural network-based algorithms in particular have garnered considerable attention for their ability to learn complex patterns from very high dimensional data sets. Physicists have also sought out pure scientific applications of various machine learning algorithms, particularly in condensed matter physics, where algorithms have been trained to reproduce or even discover phase diagrams by probing complicated long-range patterns of entanglement and classical correlations in many-body quantum systems. Since the dimension of the Hilbert space of a quantum system grows exponentially with the number of gubits, a classical computer will be unable to perform computations on, or even store the state of a quantum system with a large $(N \gtrsim 40)$ number of sites. Currently, there are NISQ devices containing more high fidelity qubits than can be simulated on a classical computer. We expect that the community will be able to extend or improve upon existing machine learning algorithms, as well as develop new algorithms, by taking advantage of these devices. This could enable learning tasks to be performed on large quantum and very large classical data sets. The quantum convolutional neural network (QCNN) is a hybrid quantumclassical algorithm with a structure similar to that of classical convolutional neural networks.733 QCNNs can learn the features of a quantum data set by performing a binary classification task on a nontrivial phase of quantum matter, and we expect such approaches to be realized on near term trapped-ion NISQ devices. An important reason to use a

Х

Chemical Reviews

pubs.acs.org/CR

trapped-ion device here is to leverage intermediate measurements, that is, to measure a fraction of the active qubits and condition subsequent operations on their measurement outcomes, needed for QCNNs to perform such classification tasks. Looking ahead, successful demonstration of a QCNN for classification on a quantum device will pave the way for development and implementation of novel hybrid quantum machine learning algorithms, which will likely take further advantage of intermediate measurement capabilities of trappedion devices, and other architectures.

4.3. Error Mitigation to Enable Practical, Near-Term Quantum Algorithms for Strongly Correlated Quantum Matter

One of the biggest challenges that quantum algorithms face is the transition from running on ideal quantum simulators to realization of these algorithms on actual "noisy" devices. While devices are continuously improving, they are prone to errors, decoherence, and noise. Ideally, scalable error correction techniques could be used to catch and eliminate errors; however, the computational cost of such a task is resourceintensive and currently impractical. A low-cost alternative is error mitigation or error resiliency schemes being built into algorithm design, leveraging algorithm-hardware co-design approaches. These improvements are critical toward the goal of using these devices to accomplish more complex tasks in computational quantum materials science. General error mitigation schemes include active error mitigation, where the impact of errors are artificially boosted and then the zero-error case is extrapolated.^{684,734-736} Many schemes have been proposed specifically to mitigate error while using hybrid variational algorithms to consider the properties of physical systems.

Several of these error mitigation schemes are motivated by the physical nature of the system of interest. One such method relies on the reduced density matrix approach²⁴⁷ as mentioned in section 3.1. This work recognized that the quantum computer measures the 2-RDM and can be assisted by physically motivated constraints placed on the system density matrix, referred to as the N-representability constraints as discussed in section 3.1. In this work, at every iteration of the VQE on the classical device, the occupation numbers of the density matrix are confined to obey the Borland-Dennis constraints prior to feeding an improved set of parameters back into the quantum device. The result of this procedure is a physically motivated error-mitigation scheme, and it has produced a state-of-the-art dissociation curve of the H₃ molecule, accurately predicting the Mott phase transition.²⁴⁷ A second physically motivated error mitigation scheme was presented in recent work that exploited the inherent symmetries found in physical systems.⁷³⁷ A symmetry of a system is an operator that commutes with the system Hamiltonian. Due to this commutation relation, the Hamiltonian can be block diagonalized within the eigenspaces of its symmetry. The system's eigenstates can therefore be considered by performing the investigation within a single target eigenspace of the symmetry. Since the resulting eigenstates should remain within the target eigenspace, errors due to certain types of noise can be detected and discarded based on symmetry verification during or after the calculation. Symmetry verifications have been performed using multiple protocols and have been shown to successfully reduce the error due to noise, as benchmarked on the ground-state dissociation curve for a hydrogen molecule.⁷³⁷

A widely used error mitigation scheme is the quantum subspace expansion (QSE) technique, which was developed to explore the low energy excited subspace of a system and consequently mitigates the effects of decoherence.^{697,718,738,739} In this protocol, an estimate for the ground-state wave function is obtained from the VQE algorithm. A quantum subspace expansion is applied by measuring additional correlation operators to form an approximate matrix representation within an extended subspace. From this approximate matrix, low-lying excited-state energies and improved energies for the ground state can be obtained, while minimizing the errors that could arise due to the quantum channel.⁷⁰⁷ The QSE technique has also been used in conjunction with a superconducting-qubit-based processor to consider the ground and excited states of a hydrogen molecule.⁶⁹⁷ This protocol is outlined in Figure 13a,



Figure 13. (a) Schematic of the QSE-VQE algorithm with classical and quantum resources shown in blue and yellow, respectively. (b) Typical qubit preparation and measurement sequence consisting of a projective heralding measurement, single-qubit and two-qubit pulses, tomography pulses, and finally a projective readout. Reproduced with permission from ref 697. Copyright 2018 American Physical Society.

where classical and quantum operations are shown in blue and yellow, respectively, and the QSE technique is used on the resulting converged state. The qubit preparation and measurement pulse sequence corresponding to the algorithm are shown in Figure 13b. This study showed successful error mitigation for incoherent errors, which further suggests potential for larger-scale quantum calculations.⁶⁹⁷

Small-scale partial quantum error correction schemes with near term applications, as well as ideas for full-scale errorcorrection, have been proposed. Many of these can be achieved by measuring syndromes of corresponding parity check operators and accordingly applying recovery operations, which are computed by a classical decoding algorithm. One class of examples are topological quantum codes, such as surface codes,^{740,741} which are implemented on a two-dimensional grid of qubits with local check operators. These codes offer great promise as they result in high error thresholds by introducing the concept of stabilizer qubits, yet they are currently still impractical due to their large overhead.⁷⁴²

Next we present two recent and exciting examples of practical quantum algorithms to capture quantum dynamics while leveraging error mitigation strategies. The first is predicting exciton condensates on NISQs. An interesting practical application of quantum computation in capturing correlated quantum matter was recently presented in predicting the creation and characterization of exciton condensates.⁷⁴³ Exciton condensation occurs when excitons, or particle-hole pairs, produce a superfluid single quantum state. Understanding superfluidity in exciton condensates has the potential to aid in a variety of technological advances from wire design to efficient room-temperature energy transport.^{229,744} A recent study prepares a highly entangled state on 3-53-qubit IBM quantum computers, which represent 3-53 particles, respectively.⁷⁴ Using the largest eigenvalue of a modified particle-hole density matrix as a quantum signature of exciton condensation, the presence and the extent of exciton condensation was evaluated. These results reveal the formation of an exciton condensate of photons, illuminating a new avenue for the creation and characterization of exciton condensates. This study also highlights the potential for using different quantum computing architectures to facilitate preparation of different exciton condensates.⁷⁴³ The second example involves exploiting the natural mapping between molecular vibrations and photonic waveguides. Boson sampling was one of the first experimentally accessible systems able to challenge the computational power of classical computers with early proposals including the calculation of Franck-Condon profiles (FCPs) with quantum optical networks, using the connection between molecular vibronic spectra and boson sampling.⁷⁴⁵ As an example, refs 746 and 747 have simulated the vibrational dynamics energy transport and relaxation of small molecules including harmonic and anharmonic effects using a versatile photonic chip.

Fully leveraging quantum computers for scientific discovery presents numerous challenges due to the experimental and heterogeneous nature of quantum hardware, and the gaps in essential software abstractions and controls needed to program this hardware in the near term. Software for quantum computing is still being developed, and therefore the development of executable code for quantum hardware using current strategies is arduous. In quantum computers, efforts to realize a set of abstractions analogous to classical computing and create "layers" of the quantum software stack are underway. Many of these have been large open-source projects co-designed with quantum hardware to ensure that the approaches are scalable with larger and more complex circuits. Currently, different types of hardware have different instruction sets, therefore both the basic computer operations and the programming language are still being defined. In addition, reducing the error rates is essential for maximizing the reliability of the results produced by current hardware. To overcome these challenges, recent developments have introduced efficient schemes that allocate the quantum program, consisting of an allocation of logical qubits, and the sequence of circuits onto physical qubits, incorporating the device specific constraints.748-753 The traditional approach to compiling and executing quantum circuits is restricted to using a highly limited set of "native" gates and does not easily allow for the ability to optimize and generalize quantum operations at the pulse level, though some hardware providers (like IBM) have given users pulse-level access to overcome this issue. While the native gates are converted into pulse sequences designed for optimal fidelity of individual single- and two-qubit operations for execution on a

universal quantum computer, they lack diversity and flexibility for optimal performance in specific tasks. More recently, quantum hardware providers have started offering open-source access designed to enable quantum instructions at the level of individual analog pulses. Initial studies have demonstrated that such basic pulse-level optimizations can yield remarkable gate and circuit fidelity improvements. A default implementation of native quantum gates uses fixed pulse shapes and duration found through prior calibration to maximize fidelity of individual operations. This approach, while offering the advantage of simplicity in implementation, suffers from suboptimal overall performance when applied universally to all quantum circuits, particularly in the context of SC NISQ devices. Given one of the primary constraints of NISQ devices, the limited quantum circuit depth, reducing the duration of microwave pulses that implement quantum operations directly increases the number of gates that can be executed within the limited coherence time of a qubit. Unfortunately, reduced pulse duration inevitably leads to an increased level of errors that arise from higher pulse amplitudes and wider spectral components of the corresponding pulses. Using quantum gates that are always realized by pulses of fixed amplitude and duration, therefore, aims to achieve optimal balance between pulse duration and gate fidelity. We anticipate important advances from the field in improved overall fidelity of quantum computation enabled through pulse-level control, thereby enabling larger calculations of quantum materials.

The efficient control of quantum systems creates a competition between the need to couple the system to the control fields while minimizing the detrimental influence of the environment. Accounting for the environment is thus indispensable to the high-precision control demands of quantum technologies. Thus, a different approach toward improving the fidelity of qubit states and therefore the accuracy and noise-resiliency of quantum algorithms is to borrow tools from the field of open quantum systems and explicitly incorporate effects of the environment, which will be the focus of the next section in this Review.

5. DECOHERENCE AND NOISE IN QUANTUM SYSTEMS

Many electronic structure methods treat quantum systems in a vacuum as closed systems, without taking their environment into account. Realistic quantum systems interact with an environment, and in many cases, this interaction dictates both their properties and dynamics. The dynamics of the system and environment together can be treated as closed and therefore evolved under unitary transformations; however, such a treatment quickly becomes computationally intractable.754,755 A common approach is to trace out the bath degrees of freedom and consider the dynamics of the reduced density matrix as an open quantum system under the influence of the environment. Due to the openness induced by this framework, the dynamics can no longer be mapped to a unitary evolution, and more complex methods are required. Moreover, depending on the choice of system and environment partitioning and the relative time scales of dynamics, interesting physical phenomena, such as memory effects, can arise. Despite these complications, how quantum systems evolve under the presence of an environment is important for the study of most molecular processes. One example is the excitonic energy transfer in reaction centers such as photosynthetic light-harvesting complexes.^{756,757} It is also critical to the development and improvement of novel quantum technologies, since decoherence and noise currently strongly



NOISE PROCESSES

Figure 14. Effects of environmental noise on a qubit as depicted on a Bloch sphere: An ideal qubit (left) would be perfectly isolated and immune to the detrimental effects of environmental noise processes such as longitudinal and transverse relaxation and therefore unwanted effects including damping and dephasing (center), which decrease the desired state fidelity, shown by a spread of possible states (right).

impact NISQ devices, the depth of quantum circuits available for practical use, and the storage of quantum information.⁶⁶³ Quantum information technologies use the principles of quantum superposition and nonlocal entanglement for applications in computing, sensing, and communication. Applying concepts from the field of open quantum systems by explicitly simulating molecular and physical qubit candidate dynamics and their environment has the potential to improve our understanding of decoherence and noise effects. This in turn has the potential to lead to control protocols that can protect qubits from the environment, decreasing the detrimental effects of the noise and therefore increasing quantum fidelity.

A schematic of this fidelity loss to the environment can be seen in Figure 14. The left panel shows a Bloch sphere representation of an arbitrary qubit state in the ideal situation where the qubit is isolated with no unwanted interactions with its surroundings. Environmentally induced processes, including but not limited to longitudinal and transverse relaxation, lead to detrimental noise effects, such as damping and dephasing, as depicted in the center panel. These noise effects can result in the loss of critical properties of the qubit and therefore a loss in fidelity of the qubit state, as demonstrated in the right panel.

A more specific example is depicted by the general point defect in a solid-state lattice, as depicted in Figure 15. Electronic structure methods can predict the properties of a single defect as discussed in section 3.2; however, defect-defect interactions and defect-lattice interactions can be treated as system-bath interactions from a master equation or numerical open quantum system perspective at a reduced computational cost. The nuclear spins from the nuclei in the lattice behave as a spin bath, while the spins in surrounding defects also interact with the system of interest, leading to decoherence and shorter spin lifetimes. Several methods have been developed to treat the dynamics of such defect spin systems, including multiple flavors of cluster-expansion (CE) methods, which will be discussed in section 5.2.2. Looking to the future, we anticipate these methods to aid in the prediction of the behavior of a local array of defects in a



Figure 15. General defect in a lattice where the defect is an open quantum system (purple) interacting with both the lattice and other defects (teal).

quantum material, descriptions of local and long-range properties such as interdefect coherence lifetimes, minute adjustments to the spacing or arrangements of defects, and direct engineering of the structure—function relationships that govern specific quantum behaviors needed to enable scalable integration. This integration is critical, especially for the application of solid-state quantum technologies, as additional qubits in quantum repeaters will be needed for error correction, entanglement distillation, and quantum repeater multiplexing. These quantum repeaters constitute the essential nodes of a quantum network, the subject of intense current and future science and engineering endeavors across the globe.

When discussing the treatment of open quantum systems through a reduced density matrix framework, the separation of system and environmental degrees of freedom dictates the nature of the dynamics. If the whole system is partitioned such that the system is weakly coupled to the environment, then Markovian evolution will arise. In more complex cases, such as

Chemical Reviews

when the system is strongly coupled to the environment, the system evolution can deviate from strict Markovianity.⁷⁵⁸ The measure of Markovianity and potential applications of the deviation from strict Markovianity are discussed in section 5.1. An overview of methods developed to treat open quantum systems in a variety of regimes is given in section 5.2, and

applications of these methods, along with an outlook of this field with respect to quantum technologies are discussed in section 5.3.

5.1. Markovianity: Definition and Measures in Quantum Systems

Markovian evolution occurs when the dynamics of a system are driven by a weakly coupled reservoir with short-lived memory effects. In this limit, the Born-Markov approximation is valid and has been successfully implemented in a variety of methods for capturing accurate open system dynamics. However, depending on parameters such as coupling strength and relative relaxation and correlation time scales of the system and the environment, this approximation can breakdown. The deviation from Markovianity is often referred to as non-Markovianity and is colloquially defined by the "backflow" of energy or information from the environment into the system. The formal definition and quantification of Markovianity versus non-Markovianity of a system is an active field of research as demonstrated by recent reviews and literature.^{759–762} The importance of distinguishing between these regimes lies in this "backflow" of information; if understood and controlled correctly, it could aid in the preservation of important quantum properties such as coherences and entanglement. In this section, our aim is to summarize recent developments in the characterization of non-Markovianity and its potential applications.

While many measures of non-Markovianity have been developed, $^{758,763-769}$ there are two leading measures that are most commonly used. 762 The first is the RHP measure based on the divisibility of dynamical maps proposed by Rivas, Huelga, and Plenio in 2010. 758 A process is Markovian if the linear, trace-preserving, completely positive map, $\Lambda(t_j, t_i)$, that connects the density matrix at time t_i to its evolved state at time t_j , can be decomposed as

$$\Lambda(t_j, t_i) = \Lambda(t_j, t_k) \Lambda(t_k, t_i)$$
⁽²³⁾

where $t_{ij}t_{ji}$ and t_k are times such that $t_i \leq t_k \leq t_j$. Since the introduction of this measure, many related techniques have been produced either as direct extensions of the RHP measure or through similar means of derivation.^{770–772} The second leading quantifier of non-Markovianity is referred to as the BLP measure, introduced by Breuer, Laine, and Piilo. It quantifies the non-Markovianity as the maximum rate of change of the trace distance between pairs of evolving quantum states D_1 and D_2 for a given evolution time frame⁷⁶⁵

$$\mathcal{N} = \max_{\{D_1(0), D_2(0)\}} \int_0^T \frac{1}{2} \frac{\mathrm{d}}{\mathrm{d}t} \mathrm{Trl} D_1(t) - D_2(t) \mathrm{l} \mathrm{d}t$$
(24)

where |...| is the square-root of the matrix norm, and the integral runs from 0 to the final time of the evolution, $T.^{765}$ For Markovian evolution, the trace distance between two quantum states $D_1(t)$ and $D_2(t)$ monotonically decreases as a function of time.⁷⁶⁵ This implies that no matter how these two states are initialized, they will become less distinguishable as they evolve. Non-Markovian behavior is connected to intervals of increase in this trace distance. Much like the RHP measure, many novel



pubs.acs.org/CR

Review

Figure 16. (a) Breuer–Laine–Piilo (BLP) measure of non-Markovianity as a function of coupling strength in the Jaynes– Cummings model, where the degree of non-Markovianity for two randomly chosen initial states (gray dots) and for two orthogonal initial states (black circles) are shown. Reproduced with permission from ref 765. Copyright 2009 by the American Physical Society. (b) Concurrence of two noninteracting subsystems interacting with a mutual environment versus the logarithm of the measure of non-Markovianity. The entanglement between the two systems is increased with the application of the control field. Adapted with permission from ref 775. Copyright 2019 by the American Physical Society.

measures have been produced either as extensions to the work of BLP or through a similar means of derivation. 773

An example of the BLP measure is shown in Figure 16a, where the degree of non-Markovianity N is shown as a function of coupling strength for the Jaynes–Cummings model. This is a commonly used benchmarking model consisting of a two-level electronic system interacting with one environmental mode, as shown schematically in the upper left corner.^{765,774} With increasing coupling between the two-level system and the environment, an increase in non-Markovianity is observed. The gray dots, which lay in lines of fixed coupling strengths, represent the BLP measure of non-Markovianity resulting from tracking the distinguishability of two randomly chosen initial density matrix states for a fixed interval of time. The black circles represent the BLP measure resulting from tracking the distinguishability of initially orthogonal density matrix states, which produces the maximum amount of non-Markovianity.

Due to their physically motivated derivations and mathematical ease of use, the above two measures have been widely used for the distinguishability between Markovianity and non-Markovianity. This is a rich and active field of research with a variety of other measures being developed, including recent works that use the covariance matrix,⁷⁷⁶ relative entropies,⁷⁷⁷ optimal state pairs,⁷⁷⁸ and semiempirical methods.^{779,780}

While the formal definition and quantification of non-Markovianity is still an active topic of research, the established measures are being used to design and analyze both theoretical and experimental set ups. One recent example is shown in Figure 16b, where several driving schemes were investigated as protocols for entanglement between two subsystems that are not directly coupled but have the potential to interact through a mutual environment.^{775,781} While three different driving schemes are considered in these works, Figure. 16b shows the entanglement in terms of concurrence, C, as a function of non-Markovianity, N, for the case in which only one subsystem is being driven. It should be noted that in this case, the concurrence depends on two elements of the density matrix and can range from 0 for completely separable states to 1 for maximally entangled states. Entanglement is shown to increase with increasing non-Markovianity, with purple and black circles representing the non-Markovianity measure given from randomly and optimally chosen initial density matrix states, respectively. This study demonstrates that entanglement can be generated by exploiting the non-Markovianity in these systems through invoking different driving schemes.⁷⁸¹ This entanglement generation protocol demonstrates how the measure of non-Markovianity can be a guiding parameter when designing optimal properties of open quantum systems for use in quantum technologies including connected quantum sensors and scalable quantum networks.

A critical component of the entanglement protocol scheme presented above is how to model the dynamics of the open quantum system. In the case of small systems, the dynamics can be solved exactly; however, for the majority of systems of interest this is rarely the case. This has spurred the development and improvement of many methods to treat open system dynamics, which will be discussed in the next section.

5.2. Methods in Non-Markovian Dynamics of Quantum Systems

As demonstrated by the comprehensive review of non-Markovian methods for open quantum systems in ref 782 and the recent perspective on the definitions, measures, and quantifiers of non-Markovianity in ref 762, handling non-Markovian effects in open quantum systems is an active field of research spanning mathematics, physics, chemistry, and quantum engineering. While it has been a topic of interest for many years, it has gained recent attention due to the development of hybrid quantum technologies, including transduction schemes between different realizations of qubits and large-scale quantum network architectures. Since these technologies often depend on optimal information transfer fidelity and preserving coherence, the ability to treat quantum systems in terms of their interactions with their surroundings is critical. Here, we discuss the most commonly used methods of treating open quantum systems in both the Markovian and non-Markovian regimes. It should be noted that this section is meant to serve as an overview of methodologies, highlighting the variety of options and progress in the field. The references within this section and the recent review articles⁷⁸² are recommended for the reader looking for more in-depth detail on the methods presented.

5.2.1. Perturbative Approaches to Open Quantum Systems. One of the most common methods of treating open quantum systems is through the use of master equations, which can be solved exactly in some cases and perturbatively in others.⁷⁵⁴ This involves integrating out the environmental

degrees of freedom from the full density matrix to consider the dynamics of the system through the reduced density matrix.⁷⁸³ In the Born–Markov approximation, the Gorini–Kossakow-ski–Sudarshan–Lindblad formalism is a commonly used master equation method for calculating dynamics, as it guarantees the positivity of the system's density matrix.^{784,785} Derived from Kraus operators with a perturbative expansion to second-order, the Lindblad equation is a first-order differential equation and is given by

$$\frac{\mathrm{d}D_{\mathrm{s}}(t)}{\mathrm{d}t} = -i[H, D_{\mathrm{s}}] + \sum_{k} \gamma_{k} \left(C_{k} D_{\mathrm{s}} C_{k}^{\dagger} - \frac{1}{2} \{ C_{k}^{\dagger} C_{k}, D_{\mathrm{s}} \} \right)$$
(25)

where D_s is the system density matrix, H is the system Hamiltonian, C_k are the k Lindbladian matrices defining different dissipation or decoherence channels, γ_k are the positive decay rates, and $\{,\}$ is the anticommutator.^{754,784,785} Due to its computational simplicity and guaranteed positivity, this has been a widely used method of treating open quantum systems weakly coupled to their environments. It should be noted that the Lindbladian matrices generally come from physical intuition, experimental evidence, or theories such as Marcus theory. In recent years, extensions to this approach such as the inclusion of accurate fermionic statistics^{786,787} and non-Markovian effects have been made.^{788,789} One example extension invoked a systematic Keldysh diagrammatic perturbation theory technique to derive Lindblad-like operators from higher order perturbative terms to investigate a broader variety of dissipative processes.⁷⁹⁰ By including fourth-order diagrams that generated correlated dissipation, this method was able to explain experimental gains and losses in driven double quantum dot resonator systems.⁷⁹⁰

A second popular master equation approach under the Born– Markov approximation is Redfield theory.^{791,792} While this master equation does not inherently preserve the positivity of the density matrix, work has been dedicated to making adjustment to include positivity such as a coarse-grained averaging technique.⁷⁹³ Similar to Lindblad's theory, the Redfield theory has also been modified, extended, and widely used in applications such as energy transfer in photosynthetic light-harvesting complexes.^{794–796}

These two methods and their extensions are common and straightforward choices for considering Markovian dynamics. However, considering system—environment interactions beyond the Markovian regime leads to a plethora of other options.^{754,797} For non-Markovian dynamics, there are several master equation approaches that are extensions of the methods used to treat Markovian dynamics.^{788,789,798–806} Among these extensions are methods referred to as post-Markovian master equations^{807–809} and semi-Markov methods,⁸¹⁰ which are computationally inexpensive but restricted to a somewhat narrow range of interaction regimes.

More generally, non-Markovian master equations tend to be divided into two main categories. There are time nonlocal master equations, which involve the calculation of a memory kernel, and time local equations, which do not explicitly treat the system memory.⁸¹¹ Neglecting the inhomogeneity term and assuming an initially factorized state of the density matrix, $D_{S+E}(0) = D_S(0) \otimes D_E(0)$, the simplified Nakajima–Zwanzig equation, or generalized master equation, is a starting point for many time nonlocal methods and is given by^{754,797,812–814}

$$\frac{\mathrm{d}D(t)}{\mathrm{d}t} = \int_0^t \mathrm{d}\tau \mathcal{K}(t,\tau) D(\tau)$$
(26)

where $\mathcal{K}(t, \tau)$ is the memory kernel. With few exceptions, this method is computationally intractable due to the complexity of the kernel. Often for calculational purposes, the memory kernel needs to be approximated through physical intuition or perturbative expansion.⁸¹⁵ Approximations of the memory kernel lead to a risk of violating the positivity of the system density matrix. An example of this violation can be seen when the generalized master equation was perturbatively expanded to

second order and negative occupation numbers were observed in a simple two-level system.⁷⁵⁴ Several methods have been developed to work around this obstacle through careful mathematical construction of the memory kernel.^{814,816} Other methods have bypassed the perturbative approach entirely and opted for numerical methods for computing the memory kernel,^{817–823} including the use of surface hopping and Ehrenfest dynamics⁸²⁴ and Mori theory.^{825–827}

The Nakajima–Zwanzig method is often compared to a timelocal projection operator technique referred to as the timeconvolutionless or TCL method.^{754,828,829} The TCL master equation is written as,

$$\frac{\mathrm{d}\mathcal{P}D(t)}{\mathrm{d}t} = \mathcal{K}(t)\mathcal{P}D(t) + I(t)QD(0) \tag{27}$$

where \mathcal{K} is the time local generator, \mathcal{I} is the inhomogeneity, and $\mathcal P$ and Q are projection operators that act on the full density matrix to obtain the relevant (system) and irrelevant (environment) components, respectively.⁸²⁸ While this equation is exact, the memory kernel and the inhomogeneity are computationally challenging and eq 27 in its general form is impractical. To overcome this computational challenge, this equation is often solved by perturbatively expanding the time local kernel then truncating at some order. However, the cost of truncating the kernel is that the positivity of the density matrix is no longer guaranteed, similar to the nonlocal form. Despite the risk of losing positivity,⁸³⁰ the time-convolutionless form has been successfully applied to many systems and processes of interest including both linear and nonlinear spectroscopic predictions.⁸³¹ Similar to the Najakima–Zwanzig equation, numerical approaches have been invoked to solve for the memory kernel in this framework.⁸³²

5.2.2. Numerical Approaches to Open Quantum **Systems.** While perturbative approaches offer sets of equations for calculating the density matrix, numerically exact methods can be implemented when an algebraic solution is not required. A variety of methods have been developed that rely on systematic convergence of numerical simulation, including methods based on path integrals^{754,782,833} such as the quasi-adiabatic path integral (QUAPI), $^{834-836}$ path integral Monte Carlo (PIMC), 837,838 and noninteracting blip approximation. The QUAPI method employs the adiabatic reference to determine the short-term system evolution operator and integrate out the degrees of freedom from the bath. By incorporation of nonadiabatic corrections through an influence functional, the exact dynamics of a system along a given adiabatic path can be described. While the method was originally developed to describe low-dimensional systems coupled to a bath of harmonic oscillators, it has been generalized and employed extensively to treat open quantum system dynamics,^{841,842} including those of charge qubits in complex environments.^{843–845} These path integral approaches have also been combined to exploit their respective strengths to further progress numerical simulation of open system dynamics.⁸⁴⁶

A complementary method that stems from Feynman's path integral approach is the hierarchical equations of motion (HEOM) method.^{847–850} Due to its ability to calculate the exact dynamics of a system linearly coupled to its environment, HEOM has been successfully applied to a variety of systems of chemical and physical interest. However, this method is hindered by its requirement that the environment possess a continuous spectral density. In recent years, many improvements and extensions of the HEOM method have been developed and successfully applied to a variety of systems,^{851–857,857} including photosynthetic light-harvesting complexes.^{757,858–861} It should be noted that when the hierarchy is truncated at low order, the HEOM method reduces to the perturbative Nakajima–Zwanzig or time-convolutionless master equation forms presented in the previous section.⁸⁶²

Monte Carlo wave function methods have been an effective alternative to density matrix approaches for numerically simulating the dynamics of Markovian open quantum systems.^{863–865} These methods evolve Monte Carlo wave functions of the system using a non-Hermitian Hamiltonian while incorporating stochastic quantum jumps. In its dissipative form, this method has been widely applied in quantum optics.⁸⁶⁶ The biggest challenge in generalizing this method to treat non-Markovian dynamics is the emergence of negative quantum jump probabilities.⁷⁸² Using a combination of forward and backward jumps enables capturing the backflow of energy or information that is associated with non-Markovian behavior.^{867,868} The positivity of the density matrix in the non-Markovian quantum jump method has also been assessed, noting that violations of positivity are associated with singularities in the jump probabilities and therefore unphysical results are prevented.⁸⁶⁹ This non-Markovian extension has been successfully applied to observe population beatings in a room temperature dimer system and excitonic energy transfer in photosynthetic complexes.^{870,871}

Some numerical methods for open system modeling have been developed with specific systems in mind. One such example is the cluster-expansion (CE) method, which has a close relationship with Feynman diagrams and acts as a systematic truncation of the Bogoliubov-Born-Green-Kirkwood-Yvon hierarchy for interacting systems. Due to its origins being rooted in the numerical treatment of interacting systems, the cluster-expansion method in a variety of forms has been applied to the treatment of open systems interacting with spin baths, including pure dephasing and the qubit decoherence problem.⁸⁷² Different flavors of the CE method have been developed including the density matrix cluster-expansion method,⁸⁷³⁻⁸⁷⁵ the linked cluster-expansion method,⁸⁷⁶ the pair-correlation cluster-expansion method,⁸⁷⁷⁻⁸⁷⁹ and the cluster-correlation expansion method.^{872,880,881} The clustercorrelation expansion (CCE) method factorizes the bath spin evolutions into cluster correlations and has been proven to bear accurate results when converged.^{872,880}

A very different alternative approach for the treatment of open quantum systems is to use machine learning methods, ^{882,883} including the use of restricted Boltzmann machines. ^{650,884,885} Work in this field has focused on Markovian dynamics of open quantum systems, ^{886–889} with a recent extension that uses parametrized quantum circuits to produce an open quantum system algorithm based on Lindbladian dynamics. ⁸⁹⁰ Along these same lines, there has been some progress toward capturing the dynamics of open quantum systems through adapting master equation approaches to be amenable to run on quantum



Figure 17. (a) Structure of a NV center diamond. The longitudinal relaxation process of the NV center is shown in terms of the survival probability of the initial state of electron spin using different orders of CCE with a bath size of 50 under the magnetic field intensity (b) $B_z = 1025.01$ G, (c) $B_z = 1024.99$ G, or (d) $B_z = 1024.97$ G. Adapted with permission from ref 881. Copyright 2020 Elsevier.

devices.^{891–894} The major challenge that these methods face is capturing nonunitary evolution of the system while relying on unitary gates. Dilation methods must be used to incorporate the important environmental degrees of freedom into a new effective system such that the evolution remains unitary. While early work could perform such dilations through use of the Stinespring dilation theory,^{892,895} the computational scaling was not favorable. Recent work used the Sz.-Nagy dilation theorem, which reduces the scaling of the mapping from the dissipative Lindbladian dynamics into a unitary form.⁸⁹⁶⁻⁸⁹⁹ Unitary operations can be decomposed into a sequence of quantum gates,⁹⁰⁰ which allowed for a two qubit calculation of a two-level system in an amplitude damping channel.⁸⁹⁹ This work has recently been extended to treat non-Markovian dynamics.⁹⁰¹ While quantum algorithms for the treatment of open quantum systems is still in its infancy, it shows promise for capturing accurate dynamics in a variety of coupling regimes at a reduced cost.

5.3. Emerging Directions in Environmentally Coupled Quantum System Dynamics

Many of the recently developed open quantum systems methods have proven to be useful in studying a variety of hybrid quantum systems. While the above two sections classified methods as either perturbative or numerical, many methods have been developed at the intersection of these two categories to better treat given systems of interest.^{902,903} In this section, we highlight a few key recent examples.

As previously mentioned, NV centers in diamond or artificial atom qubits in general exhibit complex environmental interactions, and therefore, sources of decoherence are an inherent and critical component of the system. The nitrogen atom and corresponding vacancy together introduce a net electronic spin, which interacts with an environment made up of nitrogen nuclear spins and ¹³C spins as shown in Figure 17a.⁸⁸¹ The CCE method mentioned in section 5.2 was generalized and has been applied to treat a NV center in a diamond crystal. Figure 17b,c,d depict different intensities of magnetic fields ranging from having a large energy gap between the electron and nuclear spins as compared to their hyperfine interaction to the nearly resonant case, respectively.⁸⁸¹ The orders of CCE from 2CCE to 5CCE are also compared, where 2CCE represents including clusters with one bath spin and the NV center while 5CCE represents including clusters with four bath spins and the NV center. Through these three figures, it was shown that the 5CCE provides little advantage over the 4CCE, and therefore the 4CCE is sufficient for accurate calculations. It should be noted in these methods that the spins outside of the cluster of interest are considered to be frozen when considering the cluster contribution. In the cases where the bath spins are as strongly coupled to one another as they are to the system spin, the CCE method is challenging to implement.⁹⁰⁴ However, this recent work shows that the CCE method provides a numerically tractable method of treating open quantum system dynamics in the presence of spin baths, a useful tool for improving the coherence lifetimes in NV centers, which show promise as spin qubit candidates. Moreover, many experimental studies have been committed to utilizing non-Markovian effects for the control of spin qubits such as NV centers.^{905–907} One experimental study showed that preparing different initial superposition states for the nuclear spins allows control of the electron spin dephasing and decoherence dynamics.⁹⁰⁵

Another important framework is the study of transport through molecular junctions since it is an important step toward the improvement of molecular electronics.^{908,909} Recent literature has used a variety of open quantum system methods, including master equations and Monte Carlo, to model both thermal and charge transport in molecular junctions.^{803,910–912} Using these improved methods to capture dynamics, a more accurate depiction of transport through molecular junction can be found, facilitating the path toward improved efficiency. Similar methods have also been used to model energy transfer in molecular or biological systems,⁹¹³ such as photosynthetic lightharvesting complexes,^{822,914–917} as well as manufactured devices such as molecular batteries.⁹¹⁸ Figure 18a shows the 7 site Fenna–Matthews–Olsen complex,⁹¹⁷ a commonly studied photosynthetic light-harvesting complex,^{919,920} while the dynamics of sites 3, 5, and 6 are shown in panel b.⁸²² Here, two methods are compared to the exact solution obtained from the HEOM approach.⁹¹⁴ The first is Ehrenfest mean field theory



Figure 18. (a) Seven site Fenna–Matthews–Olsen complex. Reproduced with permission from ref 917. Copyright 2014 AIP Publishing. (b) Populations of sites 3, 5, and 6 with an initial excitation in the sixth site as it decays using HEOM (dots) and direct mean field theory shown on the left and mean field generalized quantum master equation shown on the right (lines). Adapted with permission from ref 822. Copyright 2019 American Institute of Physics.

(MFT), which shows poor agreement with the exact dynamics. The second is mean field generalized quantum master equation (MF-GQME), which involves combining mean field theory with the generalized master equation and iteratively optimizing the memory kernel. This method is an example of utilizing many different open quantum system techniques, from perturbative to numerical, to obtain an efficient and accurate model of an important system.

A third area of application is in NISQ-based quantum information processing, where controlling the system evolution to minimize the detrimental effect of noise is critical for optimal information communication. While dissipative dynamics exclusively decreases quantum properties such as coherence, the presence of non-Markovian effects can create recurrences of coherences and other quantum properties. With the increased interest in developing methods to treat non-Markovian dynamics, there has also been interest in the exploitation of non-Markovian dynamics for quantum control,^{921–925} entanglement creation,^{775,781,926} and other quantum information tasks.^{927,928} Similarly, work has been dedicated toward controlling the degree of non-Markovianity of a qubit,⁹²⁹ considering the NV center in diamond as a sample system.⁹⁰⁵ Recent work considered a general weakly anharmonic ladder as the system and partitioned the infinite set of oscillators into primary and secondary baths. The primary bath is a set of twolevel systems that are strongly coupled to the system and could be the source of quantum backflow, while the secondary bath is weakly coupled and responsible for relaxation and dephasing. The isolated system allows only for operations in the Lie algebra group, SO(4), while system control through non-Markovian effects enables the realization of all quantum operations in SU(4). The error after optimization of one element in SU(4) in terms of the spin-lattice, or T1, relaxation time is shown in Figure 19a considering both a two-level system and a qudit. Figure 19b in the top panel shows the optimized control amplitudes through use of a ramping scheme shown in blue and a fast oscillating control shown in red. The bottom panel shows the amount of non-Markovianity using the determinant of the volume of reachable system states as the measure. As discussed in section 5.1, non-Markovianity is associated with an increase in this volume. This work shows that the environment can act as a resource for near-unitary quantum control of open quantum systems.923

The above example highlights how the environment can be used as a resource for controlling an arbitrary open quantum system. More generally, the control and application of non-Markovian effects are emerging as promising avenues for quantum information purposes including the development of engineered reservoirs and the improvement of quantum algorithms and protocols. A recent example is the consideration of the fidelity of generalized Pauli channels, which engineered nonlocal noise to maintain prolonged entanglement in the system.930 There also exists the potential to improve physical quantum systems or quantum hardware. Methods such as neural networks or the stochastic estimation of dynamical variables can be used to parametrize experimental noise.^{931,932} Through use of the perturbative methods discussed in previous sections, improved noise models can be integrated into these parameter estimation schemes to provide more realistic models and improved parameters.

Various approaches for using the dissipative dynamics as a computational resource have been explored under the umbrella of "reservoir engineering". During the early years of quantum



Figure 19. (a) Error from the realization of an element of SU(4) on a qudit and on a two-level system. (b) Optimized amplitudes with the control with no ramp (red) and following a fixed ramp of \pm 500 MHz over 2.5 ns at the beginning and end (blue) with the Liouville space determinant of the system evolution. Reproduced with permission from ref 923. Copyright 2015 Springer Nature.

information, a common argument against the feasibility of achieving quantum computational advantage was the notion that noise processes would always conspire to decohere the computational hardware. Reservoir engineering turns that idea around by controlling the coupling between the computational subspace and the dissipative reservoir. The basic approach closely resembles the setting behind adiabatic quantum computing: ensuring that the ground state of a given slowly changing Hamiltonian is gapped from the first excited state prevents noise processes from being able to excite, and hence decohere, the system during a computation. Thus, adiabatic quantum computing enables both computation and passive error protection. Similarly, one can consider the eigenstates of the dissipative superoperator. The "ground" steady-state subspace toward which the nonunitary superoperator forces the evolution of the computational system is naturally "protected". The goal of reservoir engineering is to find a way to make this, usually trivial, subspace into something computa-tionally useful.^{784,785,933-939} For instance, the dissipation superoperator can be engineered such that the ground space is of dimensionality higher than one. In such a subspace the dissipation can protect coherent superpositions. The stronger the engineered dissipation is, the more resilient the memory is against other dissipative effects; thus the engineered bath becomes a passive error correcting mechanism. Moreover, computation can be performed by adiabatically modifying the form of the dissipative operators.

Non-Markovian memory effects have been exploited to improve quantum algorithms and protocols.⁹²⁸ One recent algorithmic development used non-Markovian memory effects to improve the success probability of the refined Deutsch-Josza algorithm implemented on NV center spin qubits.940 The Deutsch-Jozsa algorithm is a quantum algorithm that aims to determine if a function is constant or balanced.⁹⁴¹ More colloquially, this algorithm is used to determine whether a coin is fake, implying the coin will produce either only heads or only tails, or fair, implying there is an equal probability of it landing on heads or tails. Combining non-Markovian memory effects with the dynamical decoupling protection method⁹⁴² results in a substantial improvement in the algorithms' success rate.940 These results demonstrated the ability for non-Markovian memory effects to assist in quantum algorithm improvement. Similar adaptations have been made to utilize non-Markovian effects in quantum protocols, as mentioned briefly in the entanglement generation $protocol^{775,781}$ from section 5.1. Another example is in superdense coding, where a sender can transmit two bits of classical information to a receiver through a single qubit by initially sharing an entangled state.^{674,943} Noise and decoherence decrease the initial entanglement between the sender and receiver, which in turn reduces the effects of the protocol. The experimental application of nonlocal memory effects through the use of non-Markovian noise has allowed improvements to be made in both the efficiency and security of quantum communication.^{926,944}

The measures, methods, and applications discussed here are a small sample of the vast field of open quantum systems. We anticipate that the integration of the field of open quantum systems with quantum chemistry, materials science, and quantum information will aid in improved modeling of correlated quantum matter.

6. CONCLUSIONS, OUTLOOK, AND FUTURE DIRECTIONS

In this Review, we have assembled a comprehensive foundation for those interested in the intersection of correlated quantum matter and quantum information science. In sections 2 and 3, we highlighted the strides that have been made in electronic structure theory, using both wave function and density based methods. Approaches, applications, and opportunities for algorithmic advances to capture the correlated electronic behavior in a variety of systems were discussed with a particular emphasis on molecular and color center qubits and quantum electrodynamical methods. Switching from classical approaches to quantum, in section 4, we highlighted developments in quantum algorithms as an alternative approach for reducing the scaling of the N-body problem, with a focus on hybrid algorithms. In section 5, we discussed the treatment of open quantum systems and how invoking these methodologies with respect to systems such as molecular and color center qubits has the potential to improve their performance for use in quantum technologies. Our Review highlights the interplay between electronic structure, quantum electrodynamics, quantum algorithms, open quantum systems, and novel quantum technologies. Combining breakthroughs from these fields with ideas from quantum computing has led to physical realizations and improvements of a wide range of quantum technologies and materials.

6.1. Outlook on Correlated Quantum Matter for Quantum Information Science

Currently the discovery and development of materials and molecules for quantum information science relies largely on intuition-guided trial and error. Even in the limit of atomic defects in semiconductors, conventional theoretical methods cannot always capture quantum electronic properties and dynamics at finite temperature. In the next few years, we anticipate that the field will overcome this key hurdle in solidstate quantum technologies for quantum networks and quantum sensors by developing theoretical methods to model and predict the properties of quantum defects quantitatively, identifying new protocols to characterize complex and coherently coupled solidstate quantum systems, and deploying them in practical quantum repeater nodes to translate materials-level properties into quantum devices. A success-story would be in transforming the way coherent quantum emitters are identified, developed, and engineered, to one in which numerical modeling and computational screening precede complex and costly intuitionguided experiments. Computational advantages of these methods, tested against precision quantum spectroscopy of emitters, could then be translated to broad classes of materials for other quantum devices and quantum networks. The tight discovery loop availed by the approach would allow the community to, for instance, generate a local array of emitters in a material, predict local and long-range properties such as interemitter coherence lifetimes, make minute adjustments to the spacing or arrangements of emitters, and engineer directly the structure-function relationships that govern specific quantum behaviors needed to enable scalable integration of quantum emitters. Such integration is critical for solid-state quantum technologies as additional qubits in quantum repeaters will be needed for error correction, entanglement distillation, and quantum repeater multiplexing. Further, by driving such many-body systems out of equilibrium, new states of matter can be created as recent results across nanophotonics and condensed matter physics have shown. Nominally prohibited in equilibrium, such novel states of matter feature unexpected properties, including surprisingly long coherence times, allowing for specific tailoring of quantum material and emitter properties. We envision other intersections between predictions of new quantum emitters, understanding their couplings and manybody dynamics, and theoretical control schemes of impurity arrays driven out of equilibrium via strain and phonons with the goal of optimizing them for producing new quantum optical states for quantum sensing, metrology, and quantum networking related applications.

A related promising direction is in theory-guided design of molecules with ideal ground and excited state electronic structure to create optically addressable qubits. The optical addressability in solid-state defect-based spins is dependent on a number of key factors, including an excited state manifold that allows for selective relaxation to a specific spin sublevel, an optically addressable excitation, and the capacity for fluorescence. To create molecules with comparable properties to these defect-based systems, the target molecules need to be designed to have similar attributes. In spin 1 systems, generating the optical excited state manifold requires consideration of the relative energies of the excited state triplet and the excited state singlet. Optical selection rules dictate that allowed optical excitations from a triplet ground state will be to a triplet excited state, which can then relax to a singlet state via intersystem crossings mediated by spin-orbit coupling. Designing molecular qubits with the requisite excited state manifold is achievable and, indeed, well worked out for octahedral and tetrahedral systems in which the Tanabe–Sugano diagrams determine the relationship between ligand field strength and optical excitation energy. Identifying generalizable concepts for arbitrary correlated molecular quantum matter to create modular molecular qubits, such as macrocyclic ligands with varying conjugation and axial ligands, is a rapidly growing area with close feedback between theoretical predictions and experimental investigations.

From a theory perspective, these molecular qubit candidates present an exciting challenge. Popular molecular candidates considered consist of transition metal centers with large ligands, as demonstrated by the examples shown in Figure 5. These molecules tend to exhibit strong correlation, necessitating the use of large active spaces. Systems such as these are what continue to drive electronic structure theory development toward the goal of accurate prediction of strong correlation at a tractable computational cost. For instance, using the reduced density matrix framework discussed in section 3.1, active spaces as large as 42 electron in 40 orbitals⁷ or 120 electrons in 120 orbitals in the seniority zero space²⁴³ have successfully captured strong correlation. Further development of both these and other methods mentioned in sections 2 and 3 will allow for more accurate treatment and therefore prediction of electronic properties of nontrivial molecular complexes, which can in turn be used toward the improvement of quantum technologies.

Theoretical predictions have revealed pathways toward control of microscopic parameters in complex materials, as we have discussed extensively in section 3.4. This includes modeselective enhancement, control of the amplitude and phase of order parameters, and Floquet-driven symmetry protected topological edge states. Utilizing such electromagnetically engineered nonequilibrium phases necessitates a concerted effort spanning theory and computation. Various groups are constructing variational wave functions that extend Gaussian states by including generalized canonical transformations between the matter and light degrees of freedom. The key advantage of such states compared to simple Gaussian states is the presence of nonfactorizable correlations and the possibility of describing states with strong entanglement between electrons and phonons. These wave functions are different from the commonly used canonical transformations, such as the polaron or Lang-Firsov transformations, in that parameters of the transformations are time dependent, which extends their regions of applicability. The main questions that are being addressed from these methods are enhancement of transient superconducting and charge density wave orders by parametric driving of phonons. Further, we envision exciting new work from the community in identifying new states of quantum matter under optical excitation, with a quantitative understanding of microscopic quantum dynamics and collective excitations. This is crucial to realize solid-state quantum information platforms based on nonequilibrium phenomena and has so far been a roadblock to technological exploitation of quantum matter.

6.2. Predicting Correlated Quantum Matter with New Quantum Algorithms

For problems in QED and in particular in the field of strong light—matter coupling, we foresee a wide range of applications for computational quantum algorithms. As the simulation of the Rabi model in strong coupling has demonstrated, ^{945,946} such quantum simulations are now within reach. By going beyond the two-level limitation of the Rabi model, some of the still open

questions for chemical systems in optical cavities can be targeted. Questions such as the changes in transition state of chemical reactions under strong light-matter coupling and cavity-induced superconductivity are still unsolved. These light-matter correlated systems, which include electronic, nuclear, and photonic degrees of freedom, necessitate algorithmic advances on how to efficiently describe the complex interplay of these interactions. Ideas from the electron-phonon mapping⁹⁴⁷ can also be used to map electron-photon problems to NISQ devices.

The last two decades have seen dramatic computational advances in electronic structure theory driven by software development that has exploited massive-parallelization available today on large supercomputers. Most of these highly sophisticated software packages in material science and quantum chemistry still remain poorly interfaced with the frameworks used to access NISO devices. On the forefront of these developments are highly flexible python-based quantum chemistry packages such as PSI4948 and PySCF²⁰ that have already led to successful implementations of hybrid quantumclassical algorithms for molecular systems using VQE. Overcoming the barriers between the optimized and powerful general software packages in material science and general quantum programming languages will drive the development of more advanced hybrid quantum-classical algorithms for applications of extended systems.

We highlight that developing quantum algorithms and useful implementations thereof, which maximize the impact of quantum and classical computing capabilities, requires a diverse set of tools. One such tool is scalable *ab initio* codes that compute the parameters for second-quantized quantum algorithms. Another is quantum compilers that can translate algorithms for different physical realizations of quantum computers. A third tool is quantum device simulators, which can be used to simulate the behavior of algorithms on quantum computers, to complement a physical quantum device, or to simulate the noise sensitivity of a quantum computing technology or algorithm.

So far, the wide-range of nonequilibrium dynamics of quantum systems on quantum devices remains largely unexplored. As has been discussed in this Review, general methods to accurately describe nonequilibrium many-body systems remain limited. While there exists a variety of methods that are in principle capable of describing larger systems, such as TDDFT, in practice these methods often use the adiabatic approximation, and therefore are limited to states close to the ground state. On the other hand, methods based on many-body perturbation theory, such as nonequilibrium GW, are intrinsically computationally complex, and efficient approaches for modeling larger systems have only recently been developed. This wide gap opens many possibilities for quantum devices, due to their scalability and capability to accurately describe the electronic structure. Among the exotic effects that nonequilibrium dynamics of quantum systems promises to reveal is the recently demonstrated light-induced superconductivity. To properly describe these systems, not only is the scaling of current architectures in terms of qubit size and noise reduction necessary, but also ideas of hybrid quantum-classical algorithms have to be transferred to the time domain.

Chemical Reviews

pubs.acs.org/CR

6.3. Open Quantum Systems and Quantum Transport on NISQ Devices

Modeling excitonic energy transport is another example where we anticipate that quantum algorithmic development will make rapid progress toward efficiently using quantum resources. Here, quantum transport, as seen in molecular complexes and correlated condensed matter, can arise from intricate, multibody interactions on partially ordered landscapes. In these cases, energetic disorder tends to localize particles and to inhibit coherent transport, while noise destroys coherence and enhances transport. However, too much noise and decoherence drives the system to a regime in which transport is suppressed resulting in an "optimal spot", where the interplay between static and dynamic disorder optimizes energy transport. One of the current key obstacles in understanding this many-body energy transport dynamics is the difficulty of simulating quantum systems with more than a few dozen particles or sites. NISQ information processors hold the promise of being able to simulate energy transport in quantum systems with 50-100 particles or sites, which should allow the direct simulation of quantum energy transport in catalytic systems. While variational methods are crucial to characterizing the individual energy eigenstates of complex quantum systems in the vicinity of the ground state, simulating quantum energy transport, by definition, requires an approach that goes beyond variational quantum eigensolvers, as one must take into account dynamic processes involving multiple energy eigenstates. In addition, we expect that methods of quantum simulation of complex manybody systems on NISQs will require special attention to the modeling of noise and decoherence. Though environmental noise from the thermal vibrational background can in principle be simulated using conventional open quantum system simulation methods, the NISQ processor will add its own sources of noise, which in general will not match up with the actual effects of noise from the physical environment. Carefully designing the quantum simulation could allow at least part of the hardware noise in the quantum simulator to mimic the effects of environmental noise. This could extend the applicability of noisy quantum simulators to simulating noisy quantum systems for sufficiently long time periods that they can encompass crucial and currently poorly understood aspects of quantum energy transport in multiparticle molecular complexes.

The trend toward increased interest in novel techniques for quantum simulation of open quantum systems in order to better understand and tune the effects of noise and errors is likely to continue. One direction is to directly adapt classical techniques, such as propagation schemes,⁹⁴⁹ to superconducting quantum processors with pulse-level control.950 Current NISQ superconducting quantum information processors consist of qubits coupled via local couplings and via superconducting cavities. The Hamiltonian of the information processor can be controlled by on-chip static fields, and its dynamics is controlled by the application of time-dependent microwave pulses. The conventional method for performing quantum simulations using such devices is to divide up the action of the Hamiltonian to be simulated into small time steps and to simulate each small time step using finite quantum logic operations. We anticipate that the quantum simulation can in principle be made significantly more efficient by using Hamiltonian mapping techniques, in which a carefully designed time-dependent Hamiltonian is used to simulate the quantum dynamics.

Catalyzed by recent discoveries in quantum computation, research has focused on using quantum resources to describe

open quantum systems.^{890,894,899,901} A new quantum algorithm for such open systems, one that scales favorably on quantum devices, would have far reaching impact beyond the quantum computing community. Recently presented approaches invoke the Sz.-Nagy dilation theorem for either a single Lindbladian or an ensemble of trajectories⁹⁰¹ to treat open trajectory⁸⁹ quantum systems in a wide variety of complex environments. With these resulting methods, the time evolution can be simulated on a quantum device with the potential for exponential improvements in simulation time and storage over classical algorithms, offering a more efficient alternative especially with large, correlated systems. We anticipate further algorithmic advances in the treatment of both Markovian and non-Markovian dynamics on quantum computers, as well as extensive application to systems of chemical and material interest.

We hope our timely presentation of breakthroughs, advances, and developments in the direction of quantum information and algorithms for correlated quantum matter resonate with readers interested in this intersection. From our perspective, the field is rapidly approaching algorithms with nonexponential complexity to address the grand challenge of the strongly correlated problem. Looking ahead at the vibrant intersection of quantum information science and algorithms for correlated quantum matter, we envision seminal advances in predicting many-body quantum states, describing excitonic quantum matter and largescale entangled states, a better understanding of high-temperature superconductivity, and quantifying open quantum system dynamics.

AUTHOR INFORMATION

Corresponding Authors

- Kade Head-Marsden John A. Paulson School of Engineering and Applied Sciences, Harvard University, Cambridge, Massachusetts 02138, United States; Email: kheadmarsden@ seas.harvard.edu
- Johannes Flick Center for Computational Quantum Physics, Flatiron Institute, New York 10010, United States; orcid.org/0000-0003-0273-7797; Email: jflick@ flatironinstitute.org
- Prineha Narang John A. Paulson School of Engineering and Applied Sciences, Harvard University, Cambridge, Massachusetts 02138, United States; o orcid.org/0000-0003-3956-4594; Email: prineha@seas.harvard.edu

Author

Christopher J. Ciccarino – John A. Paulson School of Engineering and Applied Sciences and Department of Chemistry and Chemical Biology, Harvard University, Cambridge, Massachusetts 02138, United States

Complete contact information is available at: https://pubs.acs.org/10.1021/acs.chemrev.0c00620

Author Contributions

[§]K.H.-M. and J.F. contributed equally to this work.

Notes

The authors declare no competing financial interest.

Biographies

Kade Head-Marsden is a postdoctoral fellow in the Narang Lab at Harvard University's Paulson School of Engineering and Applied Sciences. She received a Ph.D. in Chemistry from the University of Chicago, after earning a B.Sc. degree in Chemistry and Mathematics from McGill University. Her research has focused on reducing the scaling of electronic structure methods and developing methods for treating open quantum systems.

Johannes Flick is an associate research scientist at the Center of Quantum Physics (CCQ) of the Flatiron Institute in New York, NY, and earned his Ph.D. in theoretical physics from the Fritz Haber Institute and the Humboldt University in Berlin, Germany. Before joining CCQ, Johannes was a postdoctoral researcher at the Max Planck Institute for the Structure and Dynamics of Matter in Hamburg, Germany, and a DFG Postdoctoral Research Fellow in the Narang Lab at Harvard University, Cambridge, MA. His research is focused on the development of new computational methods for light–matter interactions on a microscopic level, as well as their applications in catalysis, spectroscopy, strong light–matter coupling, and quantuminformation processing.

Christopher J. Ciccarino is a graduate student in the Narang Lab at Harvard University's Paulson School of Engineering and Applied Sciences. He received a B.Sc. degree in Physics and Chemistry from Seton Hall University. His research centers on theoretically characterizing properties of point defects in semiconductors and insulators for quantum information processing.

Prineha Narang is an Assistant Professor at the John A. Paulson School of Engineering and Applied Sciences at Harvard University. Prior to joining the faculty, Prineha came to Harvard as a Ziff Fellow and worked as a Research Scholar in Condensed Matter Theory at the MIT Department of Physics. She received an M.S. and Ph.D. in Applied Physics from the California Institute of Technology (Caltech). Prineha's work has been recognized by many awards and special designations, including an NSF CAREER Award in 2020, being named a Moore Inventor Fellow by the Gordon and Betty Moore Foundation for pioneering innovations in quantum science and technology, CIFAR Azrieli Global Scholar by the Canadian Institute for Advanced Research, a Top Innovator by MIT Tech Review (MIT TR35), and a Young Scientist by the World Economic Forum in 2018. In 2017, she was named by Forbes Magazine on their "30under30" list for her work in atom-by-atom quantum engineering. Narang Lab's research focuses on how quantum systems behave, particularly away from equilibrium, and how we can harness these effects. By creating predictive theoretical and computational approaches to study dynamics, decoherence, and correlations in molecules and matter, her work would in future enable technologies that are inherently more powerful than their classical counterparts ranging from scalable quantum information processing to ultrahigh efficiency optoelectronic and energy conversion systems. Prineha is also the CTO and cofounder of a Boston-based startup, Aliro Quantum, towards the commercialization of quantum information processing. Outside of science, she is an avid triathlete and runner.

ACKNOWLEDGMENTS

The authors thank Dr. Stefan Krastanov, Dr. Tomáš Neuman, Gregory D. Cunningham (Harvard University), Scott E. Smart, Dr. Alexandra E. Raeber, Prof. David A Mazziotti (University of Chicago), Olivia Werba (University of Pennsylvania), and Dr. Anthony Schlimgen (Los Alamos National Lab) for the reading of and helpful suggestions on the manuscript. The authors acknowledge support from the Army Research Office MURI (Ab-Initio Solid-State Quantum Materials) grant number W911NF-18-1-0431 (for supporting work on algorithms for correlated quantum materials), the U.S. Department of Energy "Photonics at Thermodynamic Limits" Energy Frontier Research Center under grant number DE-SC0019140 (for supporting work on algorithms and methods for excited-state quantum materials), and DOE Office of Science, Basic Energy Sciences (BES), Materials Sciences and Engineering Division under FWP ERKCK47 "Understanding and Controlling Entangled and Correlated Quantum States in Confined Solidstate Systems Created via Atomic Scale Manipulation" (for supporting algorithms and computational approaches in entangled states and open quantum systems). P.N. is a Moore Inventor Fellow and gratefully acknowledges generous support through Grant GBMF8048 from the Gordon and Betty Moore Foundation. K.H.-M. also acknowledges support from the Harvard Quantum Initiative Seed Grant. The Flatiron Institute is a division of the Simons Foundation.

GLOSSARY

6	* . 11.1
C_k	Lindbladian operator corresponding to kth
	channel
Ε	energy
$E_{\rm HF}$	Hartree–Fock energy
$E_{\rm corr}$	correlation energy
E_{exact}	exact energy
GW	Expansion of electron self-energy in terms of the
	Coulomb operator <i>W</i> and single particle Green's
	function G
Ν	number of electrons
N,	number of orbitals
SO	special orthogonal group
SU	special unitary group
30 Т	partial 2 hady N representability condition
1 ₂	delta self consistent field
	dynamical map
Υ Υ	N-body wave function
$\sum_{(1)}$	electron self-energy
$\chi_{\rm R}$	response function
γ_k	decay rate corresponding to Lindbladian channel
<u>^</u>	C_k
H Â	Hamiltonian
T	excitation operator
a A ⁺	annihilation operator
â	creation operator
D	electric displacement
R	electronic dipole operator
r	position vector
С	concurrence
I(t)	inhomogeneity term
${\mathcal K}$	memory kernel
N	measure of non-Markovianity
${\mathcal P}$ and Q	projection operators
τ	time lag
^{1}D	1-electron reduced density matrix
^{2}D	2-electron reduced density matrix
^{2}G	particle-hole reduced density matrix
^{2}K	2-body Hamiltonian
^{2}Q	2-hole reduced density matrix
^{N}D	N-electron reduced density matrix
i	imaginary number $\sqrt{-1}$
п	electronic density
n_i	occupation number of <i>i</i> th orbital
$v_{\rm o}(\mathbf{r})$	static external potential
$v_{\rm xc}({\bf r})$	exchange-correlation potential
ACBN0	Agapito Curtarolo and Buongiorno Nardelli
	pseudohybrid Hubbard density functional

Chemical Reviews pubs.acs.org/		
ACSE	Antihermitian contracted Schrödinger equation Becke, 3-parameter, Lee–Yang–Parr hybrid	RDM RDMFT
B3LYP	functional	RHP
BCH	Baker–Campbell–Hausdorff	1411
BLP	Breuer-Laine-Piilo measure of non-Marko-	RI
	vianity	RPA
CAS	complete active space	SC
CASPT2	complete active space perturbation theory to	SCAN
0/10/12	second order	00/11
CASPT	complete active space perturbation theory	SDP
CASSCF	complete active space self-consistent field	SiC
CC	coupled cluster	SiV
CCE	cluster correlation expansion	STO-6G
CCS	coupled cluster singles	TCL
CCSD	coupled cluster singles doubles	TDDFT
CCSDT	coupled cluster singles doubles triples	TDRDM
CDW	charge density wave	
CE	cluster expansion	UCC
CI	configuration interaction	v2RDM
CIS	configuration interaction singles	12100111
CISD	configuration interaction singles doubles	VOE
CISDT	configuration interaction singles doubles triples	YAG
DFT	density functional theory	ZPL
DMFT	density matrix embedding theory	
DMET	dynamical mean field theory	
DOCI	doubly occupied configuration interaction	REFEREN
FCI	full configuration interaction	(1) Arute
FCP	Franck-Condon profile	R.: Biswas
CAS	generalized active space	Quantum
GAS CCA	generalized aradiant approximation	Processor.
CME	generalized master equation	(2) Boula
CPC	generalized Bauli constraints	Complexity
bbn	havaganal haran nitrida	Nat. Phys.
HEOM	hierarchical equations of motion	(3) Arute
LE	Hartroa-Each	R.; Boixo, S
HOMO	highest accupied melacular orbital	FOCK OF a
LICE	Howd_Scusoria_Ernzorhof functional	(4) Vuan
IAS	localized active space	Science 202
	local density approximation	(5) Sule
LEN	light_emitting diode	Chemistry.
LED	linear response	(6) Bhasl
	lowest-unoccupied molecular orbital	Nguyen, C
MC-SCE	multiconfigurational self-consistent field	Sukachev,
MET	man field theory	Enhanced
MET-COME	mean field theory generalized quantum master	(7) Schlin
WIT TOQUIL	aquation	Complex
MD2	Maller-Plesset perturbation theory to second	(8) Flick
1011 2	order	Ouantum (
МРТ	Maller_Plesset perturbation theory	1479-150
NISO	noisy intermediate scale quantum	(9) Flick,
NO	noisy interineurate-scale quantum	from First
NV	nitrogen_vacancy	(10) Boyı
005	open quantum systems	Electrons I
PRF	Perdew Burke and Ernzerhof functional	2020 , 11, 4
PIMC	nath integral Monte Carlo	(11) Mon
OCNN	quantum convolutional neural networks	or Yiv. 2000
OFDET	quantum electrodynamical density-functional	01707.
	theory	(12) Zun
OPEA	quantum phase estimation algorithms	Materials.
OUAPI	quasi-adiabatic path integral	(13) Fels
RAS	restricted active space	viewpoint
RASPT2	restricted active space perturbation theory to	rev.0c0073
14101 12	second order	(14) Ngu

RDM	reduced density matrix
RDMFT	reduced density matrix functional theory
RHP	Rivas, Huelga, Plenio measure of non-Marko-
	vianity
RI	resolution of identity
RPA	random-phase approximation
SC	superconducting
SCAN	strongly constrained and appropriately normed
	semilocal density functional
SDP	semidefinite programming
SiC	silicon carbide
SiV	silicon vacancy
STO-6G	Pople Slater-type orbital 6-Gaussian basis set
TCL	time convolutionless master equation
TDDFT	time-dependent density functional theory
TDRDMFT	time-dependent reduced density matrix func-
	tional theory
UCC	unitary coupled cluster
v2RDM	variational 2-electron reduced density matrix
	method
VQE	variational quantum eigensolver
YAG	yttrium ion garnet
ZPL	zero-phonon line

Review

REFERENCES

(1) Arute, F.; Arya, K.; Babbush, R.; Bacon, D.; Bardin, J. C.; Barends, R.; Biswas, R.; Boixo, S.; Brandao, F. G. S. L.; Buell, D. A.; et al. Quantum Supremacy using a Programmable Superconducting Processor. Nature 2019, 574, 505-510.

(2) Bouland, A.; Fefferman, B.; Nirkhe, C.; Vazirani, U. On the Complexity and Verification of Quantum Random Circuit Sampling. Nat. Phys. 2019, 15, 159-163.

(3) Arute, F.; Arya, K.; Babbush, R.; Bacon, D.; Bardin, J. C.; Barends, R.; Boixo, S.; Broughton, M.; Buckley, B. B.; Buell, D. A.; et al. Hartree-Fock on a Superconducting Qubit Quantum Computer. Science 2020, 369, 1084-1089.

(4) Yuan, X. A Quantum-Computing Advantage for Chemistry. Science 2020, 369, 1054-1055.

(5) Suleymanov, Y. Twelve-Qubit Quantum Computing for Chemistry. Science 2020, 369, 1069-1071.

(6) Bhaskar, M. K.; Riedinger, R.; Machielse, B.; Levonian, D. S.; Nguyen, C. T.; Knall, E. N.; Park, H.; Englund, D.; Lončar, M.; Sukachev, D. D.; et al. Experimental Demonstration of Memory-Enhanced Quantum Communication. Nature 2020, 580, 60-64.

(7) Schlimgen, A. W.; Heaps, C. W.; Mazziotti, D. A. Entangled Electrons Foil Synthesis of Elusive Low-Valent Vanadium Oxo Complex. J. Phys. Chem. Lett. 2016, 7, 627-631.

(8) Flick, J.; Rivera, N.; Narang, P. Strong Light-Matter Coupling in Quantum Chemistry and Quantum Photonics. Nanophotonics 2018, 7, 1479-1501.

(9) Flick, J.; Narang, P. Cavity-Correlated Electron-Nuclear Dynamics from First Principles. Phys. Rev. Lett. 2018, 121, 113002.

(10) Boyn, J.-N.; Xie, J.; Anderson, J. S.; Mazziotti, D. A. Entangled Electrons Drive a Non-superexchange Mechanism i. J. Phys. Chem. Lett. 2020, 11, 4584-4590.

(11) Montgomery, J. M.; Alexander, E.; Mazziotti, D. A. Prediction of the Existence of LiCH, a Carbene-like Organometallic Molecule. 2020; arXiv:2008.01707 [physics.chem-ph]; https://arxiv.org/abs/2008. 01707.

(12) Zunger, A.; Malyi, O. I. Understanding Doping of Quantum Materials. Chem. Rev. 2021. DOI: 10.1021/acs.chemrev.0c00608.

(13) Felser, C.; et al. Topological quantum materials from the viewpoint of chemistry. Chem. Rev. 2020, DOI: 10.1021/acs.chemrev.0c00732.

(14) Nguyen, L. T.; Cava, R. J. Hexagonal Perovskites as Quantum Materials. Chem. Rev. 2020, x DOI: 10.1021/acs.chemrev.0c00622.

(15) Löwdin, P.-O. Quantum Theory of Many-Particle Systems. I. Physical Interpretations by Means of Density Matrices, Natural Spin-Orbitals, and Convergence Problems in the Method of Configurational Interaction. *Phys. Rev.* **1955**, *97*, 1474–1489.

(16) Benavides-Riveros, C. L.; Schilling, C. Natural Extension of Hartree-Fock Through Extremal 1-Fermion Information: Overview and Application to the Lithium Atom. *Z. Phys. Chem.* **2016**, *230*, 703–717.

(17) Benavides-Riveros, C. L.; Lathiotakis, N. N.; Marques, M. A. L. Towards a Formal Definition of Static and Dynamic Electronic Correlations. *Phys. Chem. Chem. Phys.* **2017**, *19*, 12655–12664.

(18) Schlimgen, A. W.; Mazziotti, D. A. Static and Dynamic Electron Correlation in the Ligand Noninnocent Oxidation of Nickel Dithiolates. J. Phys. Chem. A 2017, 121, 9377–9384.

(19) Hehre, W. J.; Stewart, R. F.; Pople, J. A. Self-Consistent Molecular-Orbital Methods. I. Use of Gaussian Expansions of Slater-Type Atomic Orbitals. J. Chem. Phys. **1969**, *51*, 2657.

(20) Sun, Q.; Berkelbach, T. C.; Blunt, N. S.; Booth, G. H.; Guo, S.; Li, Z.; Liu, J.; McClain, J. D.; Sayfutyarova, E. R.; Sharma, S.; et al. PySCF: the Python-Based Simulations of Chemistry Framework. *Wiley Interdiscip. Rev.: Comput. Mol. Sci.* **2018**, *8*, No. e1340.

(21) Taylor, P. R. Lecture Notes in Quantum Chemistry II; Springer, 1994; pp 125–202.

(22) Zadrozny, J. M.; Niklas, J.; Poluektov, O. G.; Freedman, D. E. Millisecond Coherence Time in a Tunable Molecular Electronic Spin Qubit. *ACS Cent. Sci.* **2015**, *1*, 488–492.

(23) Fataftah, M. S.; Zadrozny, J. M.; Coste, S. C.; Graham, M. J.; Rogers, D. M.; Freedman, D. E. Employing Forbidden Transitions as Qubits in a Nuclear Spin-Free Chromium Complex. *J. Am. Chem. Soc.* **2016**, *138*, 1344–1348.

(24) Fataftah, M. S.; Coste, S. C.; Vlaisavljevich, B.; Zadrozny, J. M.; Freedman, D. E. Transformation of the Coordination Complex [Co(C3S5)2]2- from a Molecular Magnet to a Potential Qubit. *Chem. Sci.* **2016**, *7*, 6160–6166.

(25) Yu, C.-J.; Graham, M. J.; Zadrozny, J. M.; Niklas, J.; Krzyaniak, M. D.; Wasielewski, M. R.; Poluektov, O. G.; Freedman, D. E. Long Coherence Times in Nuclear Spin-Free Vanadyl Qubits. *J. Am. Chem. Soc.* **2016**, *138*, 14678–14685.

(26) Zadrozny, J. M.; Gallagher, A. T.; Harris, T. D.; Freedman, D. E. A Porous Array of Clock Qubits. *J. Am. Chem. Soc.* **2017**, *139*, 7089–7094.

(27) Graham, M. J.; Zadrozny, J. M.; Fataftah, M. S.; Freedman, D. E. Forging Solid-State Qubit Design Principles in a Molecular Furnace. *Chem. Mater.* **2017**, *29*, 1885–1897.

(28) Pearson, T. J.; Laorenza, D. W.; Krzyaniak, M. D.; Wasielewski, M. R.; Freedman, D. E. Octacyanometallate Qubit Candidates. *Dalton Trans.* **2018**, *47*, 11744–11748.

(29) Fataftah, M. S.; Freedman, D. E. Progress Towards Creating Optically Addressable Molecular Qubits. *Chem. Commun.* **2018**, *54*, 13773–13781.

(30) von Kugelgen, S.; Freedman, D. E. A Chemical Path to Quantum Information. *Science* **2019**, *366*, 1070–1071.

(31) Fataftah, M. S.; Krzyaniak, M. D.; Vlaisavljevich, B.; Wasielewski, M. R.; Zadrozny, J. M.; Freedman, D. E. MetalLigand Covalency Enables Room Temperature Molecular Qubit Candidates. *Chem. Sci.* **2019**, *10*, 6707–6714.

(32) Yu, C.-J.; Krzyaniak, M. D.; Fataftah, M. S.; Wasielewski, M. R.; Freedman, D. E. A Concentrated Array of Copper Porphyrin Candidate Qubits. *Chem. Sci.* **2019**, *10*, 1702–1708.

(33) Szabo, A.; Ostlund, S. *Modern Quantum Chemistry: Introduction to Advanced Electronic Structure Theory*; Dover: New York, 1989.

(34) Roos, B. O.; Taylor, P. R.; Sigbahn, P. E. A Complete Active Space SCF Method (CASSCF) using a Density Matrix Formulated Super-CI Approach. *Chem. Phys.* **1980**, *48*, 157–173.

(35) Hinze, J. MC-SCF. I. The Multi-Configuration Self-Consistent-Field Method. J. Chem. Phys. 1973, 59, 6424–6432.

(36) Dalgaard, E.; Jørgensen, P. Optimization of Orbitals for Multiconfigurational Reference States. J. Chem. Phys. **1978**, 69 (8), 3833-3844. (37) Schmidt, M. W.; Gordon, M. S. The Construction and Interpretation of MCSCF Wavefunctions. *Annu. Rev. Phys. Chem.* **1998**, 49, 233–266.

(38) Roos, B. O.; Malmqvist, P.-A. k.; Gagliardi, L. Exploring the Actinide-Actinide Bond: Theoretical Studies of the Chemical Bond in Ac2, Th2, Pa2, and U2. *J. Am. Chem. Soc.* **2006**, *128*, 17000–17006.

(39) Gagliardi, L.; Roos, B. O. Multiconfigurational Quantum Chemical Methods for Molecular Systems Containing Actinides. *Chem. Soc. Rev.* **2007**, *36*, 893–903.

(40) Radoń, M.; Pierloot, K. Binding of CO, NO, and O2 to Heme by Density Functional and Multireference ab Initio Calculations. *J. Phys. Chem. A* **2008**, *112*, 11824–11832.

(41) Fleig, T.; Olsen, J.; Visscher, L. The Generalized Active Space Concept for the Relativistic Treatment of Electron Correlation. II. Large-Scale Configuration Interaction Implementation Based on Relativistic 2- and 4-Spinors and its Application. J. Chem. Phys. 2003, 119, 2963–2971.

(42) Ma, D.; Li Manni, G.; Gagliardi, L. The Generalized Active Space Concept in Multiconfigurational Self-Consistent Field Methods. *J. Chem. Phys.* **2011**, *135*, No. 044128.

(43) Vogiatzis, K. D.; Li Manni, G.; Stoneburner, S. J.; Ma, D.; Gagliardi, L. Systematic Expansion of Active Spaces beyond the CASSCF Limit: A GASSCF/SplitGAS Benchmark Study. *J. Chem. Theory Comput.* **2015**, *11*, 3010–3021.

(44) Olsen, J.; Roos, B. O.; Jørgensen, P.; Jensen, H. J. A. Determinant Based Configuration Interaction Algorithms for Complete and Restricted Configuration Interaction Spaces. J. Chem. Phys. **1988**, 89, 2185–2192.

(45) Malmqvist, P. A.; Rendell, A.; Roos, B. O. The Restricted Active Space Self-Consistent-Field Method, Implemented with a Split Graph Unitary Group Approach. *J. Phys. Chem.* **1990**, *94*, 5477–5482.

(46) Hermes, M. R.; Gagliardi, L. Multiconfigurational Self-Consistent Field Theory with Density Matrix Embedding: The Localized Active Space Self-Consistent Field Method. J. Chem. Theory Comput. 2019, 15, 972–986.

(47) Hermes, M. R.; Pandharkar, R.; Gagliardi, L. Variational Localized Active Space Self-Consistent Field Method. J. Chem. Theory Comput. 2020, 16, 4923–4937.

(48) Møller, C.; Plesset, M. S. Note on an Approximation Treatment for Many-Electron Systems. *Phys. Rev.* **1934**, *46*, 618–622.

(49) Head-Gordon, M.; Pople, J. A.; Frisch, M. J. MP2 Energy Evaluation by Direct Methods. *Chem. Phys. Lett.* **1988**, *153*, 503–506. (50) Whitten, J. L. Coulombic Potential Energy Integrals and Approximations. *J. Chem. Phys.* **1973**, *58*, 4496–4501.

(51) Dunlap, B. I.; Connolly, J. W. D.; Sabin, J. R. On Some Approximations in Applications of $X\alpha$ Theory. *J. Chem. Phys.* **1979**, *71*, 3396–3402.

(52) Feyereisen, M.; Fitzgerald, G.; Komornicki, A. Use of Approximate Integrals in Ab Initio Theory. An Application in MP2 Energy Calculations. *Chem. Phys. Lett.* **1993**, *208*, 359–363.

(53) Bernholdt, D. E.; Harrison, R. J. Large-Scale Correlated Electronic Structure Calculations: the RI-MP2 Method on Parallel Computers. *Chem. Phys. Lett.* **1996**, *250*, 477–484.

(54) Werner, H.-J.; Manby, F. R.; Knowles, P. J. Fast Linear Scaling Second-Order Møller-Plesset Perturbation Theory (MP2) using Local and Density Fitting Approximations. *J. Chem. Phys.* **2003**, *118*, 8149–8160.

(55) DiStasio, R. A.; Jung, Y.; Head-Gordon, M. A Resolution-Of-The-Identity Implementation of the Local Triatomics-In-Molecules Model for Second-Order Møller-Plesset Perturbation Theory with Application to Alanine Tetrapeptide Conformational Energies. J. Chem. Theory Comput. **2005**, *1*, 862–876.

(56) Beebe, N. H. F.; Linderberg, J. Simplifications in the Generation and Transformation of Two-Electron Integrals in Molecular Calculations. *Int. J. Quantum Chem.* **1977**, *12*, 683–705.

(57) Koch, H.; Sánchez de Merás, A.; Pedersen, T. B. Reduced Scaling in Electronic Structure Calculations using Cholesky Decompositions. *J. Chem. Phys.* **2003**, *118*, 9481–9484. (58) Aquilante, F.; Pedersen, T. B.; Lindh, R. Low-Cost Evaluation of the Exchange Fock Matrix from Cholesky and Density Fitting Representations of the Electron Repulsion Integrals. *J. Chem. Phys.* **2007**, *126*, 194106.

(59) Higham, N. J. Cholesky Factorization. Wiley Interdiscip. Rev. Comput. Stat. 2009, 1, 251–254.

(60) Aquilante, F.; Gagliardi, L.; Pedersen, T. B.; Lindh, R. Atomic Cholesky Decompositions: A Route to Unbiased Auxiliary Basis Sets for Density Fitting Approximation with Tunable Accuracy and Efficiency. J. Chem. Phys. **2009**, 130, 154107.

(61) Schütz, M.; Hetzer, G.; Werner, H.-J. Low-Order Scaling Local Electron Correlation Methods. I. Linear Scaling Local MP2. *J. Chem. Phys.* **1999**, *111*, 5691–5705.

(62) Lee, M. S.; Maslen, P. E.; Head-Gordon, M. Closely Approximating Second-Order Møller-Plesset Perturbation Theory with a Local Triatomics in Molecules Model. *J. Chem. Phys.* **2000**, *112*, 3592–3601.

(63) Cremer, D. Møller-Plesset Perturbation Theory: from Small Molecule Methods to Methods for Thousands of Atoms. *Wiley Interdiscip. Rev. Comput. Mol. Sci.* **2011**, *1*, 509–530.

(64) Ayala, P. Y.; Scuseria, G. E. Linear Scaling Second-Order Møller-Plesset Theory in the Atomic Orbital Basis for Large Molecular Systems. J. Chem. Phys. **1999**, 110, 3660–3671.

(65) Scuseria, G. E.; Ayala, P. Y. Linear Scaling Coupled Cluster and Perturbation Theories in the Atomic Orbital Basis. *J. Chem. Phys.* **1999**, *111*, 8330–8343.

(66) Doser, B.; Lambrecht, D. S.; Kussmann, J.; Ochsenfeld, C. Linear-Scaling Atomic Orbital-Based Second-Order Møller-Plesset Perturbation Theory by Rigorous Integral Screening Criteria. *J. Chem. Phys.* **2009**, *130*, No. 064107.

(67) Maurer, S. A.; Lambrecht, D. S.; Kussmann, J.; Ochsenfeld, C. Efficient Distance-Including Integral Screening in Linear-Scaling Møller-Plesset Perturbation Theory. *J. Chem. Phys.* **2013**, *138*, No. 014101.

(68) Baudin, P.; Ettenhuber, P.; Reine, S.; Kristensen, K.; Kjaergaard, T. Efficient Linear-Scaling Second-Order Møller-Plesset Perturbation Theory: The Divide-Expand-Consolidate RI-MP2 Model. *J. Chem. Phys.* **2016**, *144*, No. 054102.

(69) Nagy, P. R.; Samu, G.; Kállay, M. An Integral-Direct Linear-Scaling Second-Order Møller-Plesset Approach. J. Chem. Theory Comput. 2016, 12, 4897–4914.

(70) Sinnokrot, M. O.; Valeev, E. F.; Sherrill, C. D. Estimates of the Ab Initio Limit for π - π Interactions: The Benzene Dimer. *J. Am. Chem. Soc.* **2002**, *124*, 10887–10893.

(71) Jurečka, P.; Šponer, J.; Černy, J.; Hobza, P. Benchmark Database of Accurate (MP2 and CCSD(T) Complete Basis Set Limit) Interaction Energies of Small Model Complexes, DNA Base Pairs, and Amino Acid Pairs. *Phys. Chem. Chem. Phys.* **2006**, *8*, 1985–1993.

(72) Cybulski, S. M.; Lytle, M. L. The Origin of Deficiency of the Supermolecule Second-Order Møller-Plesset Approach for Evaluating Interaction Energies. *J. Chem. Phys.* **2007**, *127*, 141102.

(73) Grimme, S. Improved Second-Order Møller-Plesset Perturbation Theory by Separate Scaling of Parallel- and Antiparallel-Spin Pair Correlation Energies. J. Chem. Phys. **2003**, 118, 9095–9102.

(74) Jung, Y.; Lochan, R. C.; Dutoi, A. D.; Head-Gordon, M. Scaled Opposite-Spin Second Order Møller-Plesset Correlation Energy: An Economical Electronic Structure Method. *J. Chem. Phys.* **2004**, *121*, 9793–9802.

(75) Grimme, S. Accurate Calculation of the Heats of Formation for Large Main Group Compounds with Spin-Component Scaled MP2 Methods. *J. Phys. Chem. A* **2005**, *109*, 3067–3077.

(76) Lochan, R. C.; Jung, Y.; Head-Gordon, M. Scaled Opposite Spin Second Order Møller-Plesset Theory with Improved Physical Description of Long-Range Dispersion Interactions. *J. Phys. Chem. A* **2005**, *109*, 7598–7605.

(77) Distasio, R. A., Jr.; Head-Gordon, M. Optimized Spin-Component Scaled Second-Order Møller-Plesset Perturbation Theory for Intermolecular Interaction Energies. *Mol. Phys.* **2007**, *105*, 1073–1083.

(78) Lochan, R. C.; Shao, Y.; Head-Gordon, M. Quartic-Scaling Analytical Energy Gradient of Scaled Opposite-Spin Second-Order Møller-Plesset Perturbation Theory. *J. Chem. Theory Comput.* **2007**, *3*, 988–1003.

(79) Lochan, R. C.; Head-Gordon, M. Orbital-Optimized Opposite-Spin Scaled Second-Order Correlation: An Economical Method to Improve the Description of Open-Shell Molecules. *J. Chem. Phys.* **2007**, *126*, 164101.

(80) Neese, F.; Schwabe, T.; Kossmann, S.; Schirmer, B.; Grimme, S. Assessment of Orbital-Optimized, Spin-Component Scaled Second-Order Many-Body Perturbation Theory for Thermochemistry and Kinetics. J. Chem. Theory Comput. **2009**, *5*, 3060–3073.

(81) Bozkaya, U.; Turney, J. M.; Yamaguchi, Y.; Schaefer, H. F.; Sherrill, C. D. Quadratically Convergent Algorithm for Orbital Optimization in the Orbital-Optimized Coupled-Cluster Doubles Method and in Orbital-Optimized Second-Order Møller-Plesset Perturbation Theory. J. Chem. Phys. **2011**, 135, 104103.

(82) Lee, J.; Head-Gordon, M. Regularized Orbital-Optimized Second-Order Møller-Plesset Perturbation Theory: A Reliable Fifth-Order-Scaling Electron Correlation Model with Orbital Energy Dependent Regularizers. J. Chem. Theory Comput. 2018, 14, 5203– 5219.

(83) McGibbon, R. T.; Taube, A. G.; Donchev, A. G.; Siva, K.; Hernández, F.; Hargus, C.; Law, K.-H.; Klepeis, J. L.; Shaw, D. E. Improving the Accuracy of Møller-Plesset Perturbation Theory with Neural Networks. J. Chem. Phys. **2017**, *147*, 161725.

(84) Piris, M. Global Method for Electron Correlation. *Phys. Rev. Lett.* **2017**, *119*, No. 063002.

(85) Hannon, K. P.; Li, C.; Evangelista, F. A. An Integral-Factorized Implementation of the Driven Similarity Renormalization Group Second-Order Multireference Perturbation Theory. *J. Chem. Phys.* **2016**, *144*, 204111.

(86) Yost, S. R.; Head-Gordon, M. Size Consistent Formulations of the Perturb-then-Diagonalize Møller-Plesset Perturbation Theory Correction to Non-Orthogonal Configuration Interaction. *J. Chem. Phys.* **2016**, *145*, No. 054105.

(87) Rezáč, J.; Greenwell, C.; Beran, G. J. O. Accurate Noncovalent Interactions via Dispersion-Corrected Second-Order Møller-Plesset Perturbation Theory. J. Chem. Theory Comput. **2018**, 14, 4711–4721.

(88) Andersson, K.; Malmqvist, P.; Roos, B. O. Second-Order Perturbation Theory with a Complete Active Space Self-Consistent Field Reference Function. *J. Chem. Phys.* **1992**, *96*, 1218–1226.

(89) Finley, J.; Malmqvist, P. Å; Roos, B. O.; Serrano-Andrés, L. The Multi-State CASPT2 Method. *Chem. Phys. Lett.* **1998**, 288, 299–306.

(90) Malmqvist, P. Å.; Pierloot, K.; Shahi, A. R. M.; Cramer, C. J.; Gagliardi, L. The Restricted Active Space Followed by Second-Order Perturbation Theory Method: Theory and Application to the Study of CuO₂ and Cu₂O₂ Systems. *J. Chem. Phys.* **2008**, *128*, 204109.

(91) González, L.; Escudero, D.; Serrano-Andrés, L. Progress and Challenges in the Calculation of Electronic Excited States. *ChemPhysChem* **2012**, *13*, 28–51.

(92) Ruipérez, F.; Aquilante, F.; Ugalde, J. M.; Infante, I. Complete vs Restricted Active Space Perturbation Theory Calculation of the Cr_2 Potential Energy Surface. *J. Chem. Theory Comput.* **2011**, *7*, 1640– 1646.

(93) Malrieu, J. P.; Caballol, R.; Calzado, C. J.; de Graaf, C.; Guihéry, N. Magnetic Interactions in Molecules and Highly Correlated Materials: Physical Content, Analytical Derivation, and Rigorous Extraction of Magnetic Hamiltonians. *Chem. Rev.* **2014**, *114*, 429–492. (94) Sherrill, C. D.; Schaefer, H. F. In *The Configuration Interaction Method: Advances in Highly Correlated Approaches*; Löwdin, P.-O., Sabin, J. R., Zerner, M. C., Brändas, E., Eds.; Advances in Quantum Chemistry; Academic Press, 1999; Vol. 34; pp 143–26.

(95) Weinhold, F.; Wilson, E. B. Reduced Density Matrices of Atoms and Molecules. I. The 2 Matrix of Double-Occupancy, Configuration-Interaction Wavefunctions for Singlet States. J. Chem. Phys. **1967**, 46, 2752–2758.

(96) Alcoba, D. R.; Torre, A.; Lain, L.; Oña, O. B.; Capuzzi, P.; Van Raemdonck, M.; Bultinck, P.; Van Neck, D. A Hybrid Configuration Interaction Treatment Based on Seniority Number and Excitation Schemes. J. Chem. Phys. 2014, 141, 244118.

(97) Coe, J. P. Machine Learning Configuration Interaction. J. Chem. Theory Comput. **2018**, 14, 5739–5749.

(98) Coe, J. P. Machine Learning Configuration Interaction for ab Initio Potential Energy Curves. J. Chem. Theory Comput. **2019**, 15, 6179–6189.

(99) Romero, J.; Babbush, R.; McClean, J. R.; Hempel, C.; Love, P. J.; Aspuru-Guzik, A. Strategies for Quantum Computing Molecular Energies using the Unitary Coupled Cluster Ansatz. *Quantum Sci. Technol.* **2019**, *4*, No. 014008.

(100) Bishop, R. F.; Brandes, T.; Gernoth, K. A.; Walet, N. R.; Xian, Y. *Recent Progress in Many-Body Theories*; World Scientific, 2002.

(101) Bartlett, R. J.; Musiał, M. Coupled-cluster Theory in Quantum Chemistry. *Rev. Mod. Phys.* **2007**, *79*, 291–352.

(102) Bartlett, R. J.; Kucharski, S. A.; Noga, J. Alternative Coupled-Cluster Ansätze II. The Unitary Coupled-Cluster Method. *Chem. Phys. Lett.* **1989**, *155*, 133–140.

(103) Taube, A. G.; Bartlett, R. J. New Perspectives on Unitary Coupled-Cluster Theory. *Int. J. Quantum Chem.* **2006**, *106*, 3393–3401.

(104) Margraf, J. T.; Reuter, K. Making the Coupled Cluster Correlation Energy Machine-Learnable. *J. Phys. Chem. A* 2018, *122*, 6343–6348.

(105) Townsend, J.; Vogiatzis, K. D. Data-Driven Acceleration of the Coupled-Cluster Singles and Doubles Iterative Solver. J. Phys. Chem. Lett. 2019, 10, 4129–4135.

(106) Gyevi-Nagy, L.; Kállay, M.; Nagy, P. R. Integral-Direct and Parallel Implementation of the CCSD(T) Method: Algorithmic Developments and Large-Scale Applications. *J. Chem. Theory Comput.* **2020**, *16*, 366–384.

(107) McClain, J.; Sun, Q.; Chan, G. K.-L.; Berkelbach, T. C. Gaussian-Based Coupled-Cluster Theory for the Ground-State and Band Structure of Solids. *J. Chem. Theory Comput.* **2017**, *13*, 1209–1218.

(108) Becke, A. D. Density-Functional Exchange-Energy Approximation with Correct Asymptotic Behavior. *Phys. Rev. A: At., Mol., Opt. Phys.* **1988**, 38, 3098–3100.

(109) Stephens, P. J.; Devlin, F. J.; Chabalowski, C. F.; Frisch, M. J. Ab Initio Calculation of Vibrational Absorption and Circular Dichroism Spectra Using Density Functional Force Fields. *J. Phys. Chem.* **1994**, *98*, 11623–11627.

(110) Jones, L. O.; Mosquera, M. A.; Schatz, G. C.; Ratner, M. A. Embedding Methods for Quantum Chemistry: Applications from Materials to Life Sciences. J. Am. Chem. Soc. **2020**, *142*, 3281–3295.

(111) Ye, H.-Z.; Welborn, M.; Ricke, N. D.; Van Voorhis, T. Incremental Embedding: A Density Matrix Embedding Scheme for Molecules. J. Chem. Phys. 2018, 149, 194108.

(112) Ye, H.-Z.; Van Voorhis, T. Atom-Based Bootstrap Embedding For Molecules. J. Phys. Chem. Lett. **2019**, 10, 6368–6374.

(113) Ye, H.-Z.; Ricke, N. D.; Tran, H. K.; Van Voorhis, T. Bootstrap Embedding for Molecules. J. Chem. Theory Comput. **2019**, 15, 4497– 4506.

(114) Sun, Q.; Chan, G. K.-L. Quantum Embedding Theories. Acc. Chem. Res. 2016, 49, 2705–2712.

(115) Chibani, W.; Ren, X.; Scheffler, M.; Rinke, P. Self-Consistent Green's Function Embedding for Advanced Electronic Structure methods based on a dynamical mean-field concept. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2016**, *93*, 165106.

(116) Parr, R. G.; Donnelly, R. A.; Levy, M.; Palke, W. E. Electronegativity: The Density Functional Viewpoint. *J. Chem. Phys.* **1978**, *68*, 3801–3807.

(117) Cortona, P. Self-Consistently Determined Properties of Solids without Band-Structure Calculations. *Phys. Rev. B: Condens. Matter Mater. Phys.* **1991**, *44*, 8454–8458.

(118) Wesolowski, T. A.; Warshel, A. Frozen Density Functional Approach for Ab Initio Calculations of Solvated Molecules. J. Phys. Chem. **1993**, 97, 8050–8053.

(119) Knizia, G.; Chan, G. K.-L. Density Matrix Embedding: A Simple Alternative to Dynamical Mean-Field Theory. *Phys. Rev. Lett.* **2012**, *109*, 186404.

(120) Metzner, W.; Vollhardt, D. Correlated Lattice Fermions in $d = \infty$ Dimensions. *Phys. Rev. Lett.* **1989**, *62*, 324–327.

(121) Georges, A.; Krauth, W. Numerical Solution of the $d = \infty$ Hubbard Model: Evidence for a Mott Transition. *Phys. Rev. Lett.* **1992**, 69, 1240–1243.

(122) Georges, A.; Kotliar, G.; Krauth, W.; Rozenberg, M. J. Dynamical Mean-Field Theory of Strongly Correlated Fermion Systems and the Limit of Infinite Dimensions. *Rev. Mod. Phys.* **1996**, 68, 13–125.

(123) Kotliar, G.; Savrasov, S. Y.; Haule, K.; Oudovenko, V. S.; Parcollet, O.; Marianetti, C. A. Electronic Structure Calculations with Dynamical Mean-Field Theory. *Rev. Mod. Phys.* **2006**, *78*, 865–951.

(124) Maier, T.; Jarrell, M.; Pruschke, T.; Hettler, M. H. Quantum Cluster Theories. *Rev. Mod. Phys.* **2005**, *77*, 1027–1080.

(125) Wouters, S.; Jiménez-Hoyos, C. A.; Chan, K.-L. G. In *Fragmentation*; Gordon, M. S., Ed.; John Wiley & Sons, Ltd, 2017; Chapter 8, pp 227–243.

(126) Knizia, G.; Chan, G. K.-L. Density Matrix Embedding: A Strong-Coupling Quantum Embedding Theory. J. Chem. Theory Comput. 2013, 9, 1428–1432.

(127) Bulik, I. W.; Chen, W.; Scuseria, G. E. Electron Correlation in Solids via Density Embedding Theory. J. Chem. Phys. 2014, 141, No. 054113.

(128) Pham, H. Q.; Bernales, V.; Gagliardi, L. Can Density Matrix Embedding Theory with the Complete Activate Space Self-Consistent Field Solver Describe Single and Double Bond Breaking in Molecular Systems? *J. Chem. Theory Comput.* **2018**, *14*, 1960–1968.

(129) Tsuchimochi, T.; Welborn, M.; Van Voorhis, T. Density Matrix Embedding in an Antisymmetrized Geminal Power Bath. *J. Chem. Phys.* **2015**, *143*, No. 024107.

(130) Sandhoefer, B.; Chan, G. K.-L. Density Matrix Embedding Theory for Interacting Electron-Phonon Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2016**, *94*, No. 085115.

(131) Chen, Q.; Booth, G. H.; Sharma, S.; Knizia, G.; Chan, G. K.-L. Intermediate and Spin-Liquid Phase of the Half-Filled Honeycomb Hubbard Model. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2014**, *89*, 165134.

(132) LeBlanc, J. P. F.; Antipov, A. E.; Becca, F.; Bulik, I. W.; Chan, G. K.-L.; Chung, C.-M.; Deng, Y.; Ferrero, M.; Henderson, T. M.; Jiménez-Hoyos, C. A.; et al. Solutions of the Two-Dimensional Hubbard Model: Benchmarks and Results from a Wide Range of Numerical Algorithms. *Phys. Rev. X* 2015, *5*, No. 041041.

(133) Zheng, B.-X.; Chan, G. K.-L. Ground-State Phase Diagram of the Square Lattice Hubbard Model from Density Matrix Embedding Theory. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2016**, *93*, No. 035126.

(134) Wu, X.; Cui, Z.-H.; Tong, Y.; Lindsey, M.; Chan, G. K.-L.; Lin, L. Projected Density Matrix Embedding Theory with Applications to the Two-Dimensional Hubbard Model. *J. Chem. Phys.* **2019**, *151*, No. 064108.

(135) Wouters, S.; Jiménez-Hoyos, C. A.; Sun, Q.; Chan, G. K.-L. A Practical Guide to Density Matrix Embedding Theory in Quantum Chemistry. J. Chem. Theory Comput. **2016**, *12*, 2706–2719.

(136) Hachmann, J.; Cardoen, W.; Chan, G. K.-L. Multireference Correlation in Long Molecules with the Quadratic Scaling Density Matrix Renormalization Group. *J. Chem. Phys.* **2006**, *125*, 144101.

(137) Booth, G. H.; Chan, G. K.-L. Spectral Functions of Strongly Correlated Extended Systems via an Exact Quantum Embedding. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, *91*, 155107.

(138) Tran, H. K.; Van Voorhis, T.; Thom, A. J. W. Using SCF Metadynamics to Extend Density Matrix Embedding Theory to Excited States. *J. Chem. Phys.* **2019**, *151*, No. 034112.

(139) Gaita-Ariño, A.; Luis, F.; Hill, S.; Coronado, E. Molecular Spins for Quantum Computation. *Nat. Chem.* **2019**, *11*, 301–309.

(140) Bayliss, S. L.; Laorenza, D. W.; Mintun, P. J.; Diler, B.; Freedman, D. E.; Awschalom, D. D. Optically Addressable Molecular Spins for Quantum Information Processing. 2020; arXiv:2004.07998 [quant-ph]; https://arxiv.org/abs/2004.07998.

(141) DiVincenzo, D. P. The Physical Implementation of Quantum Computation. *Fortschr. Phys.* **2000**, *48*, 771–783.

(142) Wasielewski, M. R.; Forbes, M. D. E.; Frank, N. L.; Kowalski, K.; Scholes, G. D.; Yuen-Zhou, J.; Baldo, M. A.; Freedman, D. E.; Goldsmith, R. H.; Goodson, T.; et al. Exploiting Chemistry and Molecular Systems for Quantum Information Science. *Nat. Rev. Chem.* **2020**, *4*, 490–504.

(143) Wu, Y.; Jelezko, F.; Plenio, M. B.; Weil, T. Diamond Quantum Devices in Biology. *Angew. Chem., Int. Ed.* **2016**, *55*, 6586–6598.

(144) Murphy, D. M. Principles of Pulse Electron Paramagnetic Resonance. By A Schweiger and G Jeschke, Oxford University Press, UK, 2001, 578 pp. ISBN 0 19 850634 1. J. Chem. Technol. Biotechnol. **2004**, 79, 103–103.

(145) Sproules, S. *Electron Paramagnetic Resonance*; The Royal Society of Chemistry, 2017; Vol. 25; pp 61–97.

(146) Atzori, M.; Sessoli, R. The Second Quantum Revolution: Role and Challenges of Molecular Chemistry. J. Am. Chem. Soc. 2019, 141, 11339–11352.

(147) Chen, J.; Hu, C.; Stanton, J. F.; Hill, S.; Cheng, H.-P.; Zhang, X.-G. Decoherence in Molecular Electron Spin Qubits: Insights from Quantum Many-Body Simulations. *J. Phys. Chem. Lett.* **2020**, *11*, 2074–2078.

(148) Krzyaniak, M. D.; Kobr, L.; Rugg, B. K.; Phelan, B. T.; Margulies, E. A.; Nelson, J. N.; Young, R. M.; Wasielewski, M. R. Fast Photo-Driven Electron Spin Coherence Transfer: The Effect of Electron-Nuclear Hyperfine Coupling on Coherence Dephasing. *J. Mater. Chem. C* 2015, *3*, 7962–7967.

(149) Nelson, J. N.; Krzyaniak, M. D.; Horwitz, N. E.; Rugg, B. K.; Phelan, B. T.; Wasielewski, M. R. Zero Quantum Coherence in a Series of Covalent Spin-Correlated Radical Pairs. *J. Phys. Chem. A* **2017**, *121*, 2241–2252.

(150) Lombardi, F.; Lodi, A.; Ma, J.; Liu, J.; Slota, M.; Narita, A.; Myers, W. K.; Müllen, K.; Feng, X.; Bogani, L. Quantum Units from the Topological Engineering of Molecular Graphenoids. *Science* **2019**, *366*, 1107–1110.

(151) Ullah, A.; Cerdá, J.; Baldoví, J. J.; Varganov, S. A.; Aragó, J.; Gaita-Ariño, A. In Silico Molecular Engineering of Dysprosocenium-Based Complexes to Decouple Spin Energy Levels from Molecular Vibrations. J. Phys. Chem. Lett. **2019**, *10*, 7678–7683.

(152) Schrauben, J. N.; Dillman, K. L.; Beck, W. F.; McCusker, J. K. Vibrational Coherence in the Excited State Dynamics of Cr(acac)₃: Probing the Reaction Coordinate for Ultrafast Intersystem Crossing. *Chem. Sci.* **2010**, *1*, 405–410.

(153) Mazziotti, D. A. Quantum Chemistry without Wave Functions: Two-Electron Reduced Density Matrices. *Acc. Chem. Res.* **2006**, *39*, 207–215.

(154) Coleman, A. J. Structure of Fermion Density Matrices. *Rev. Mod. Phys.* **1963**, 35, 668–686.

(155) Erdahl, R. Two Algorithms for the Lower Bound Method of Reduced Density Matrix Theory. *Rep. Math. Phys.* **1979**, *15*, 147–162.

(156) Nakata, M.; Nakatsuji, H.; Ehara, M.; Fukuda, M.; Nakata, K.; Fujisawa, K. Variational Calculations of Fermion Second-Order Reduced Density Matrices by Semidefinite Programming Algorithm. J. Chem. Phys. 2001, 114, 8282–8292.

(157) Mazziotti, D. A.; Erdahl, R. M. Uncertainty Relations and Reduced Density Matrices: Mapping Many-Body Quantum Mechanics onto Four Particles. *Phys. Rev. A: At., Mol., Opt. Phys.* **2001**, *63*, No. 042113.

(158) Zhao, Z.; Braams, B. J.; Fukuda, M.; Overton, M. L.; Percus, J. K. The Reduced Density Matrix Method for Electronic Structure Calculations and the Role of Three-Index Representability Conditions. *J. Chem. Phys.* **2004**, *120*, 2095–2104.

(159) Mazziotti, D. A. Realization of Quantum Chemistry without Wave Functions through First-Order Semidefinite Programming. *Phys. Rev. Lett.* **2004**, 93, 213001. (160) Mazziotti, D. A. Structure of Fermionic Density Matrices: Complete N-Representability Conditions. *Phys. Rev. Lett.* **2012**, *108*, 263002.

(161) Mazziotti, D. A. Pure-N-Representability Conditions of Two-Fermion Reduced Density Matrices. *Phys. Rev. A: At., Mol., Opt. Phys.* **2016**, *94*, No. 032516.

(162) DePrince, A. E. Variational Optimization of the Two-Electron Reduced-Density Matrix under Pure-State N-Representability Conditions. J. Chem. Phys. **2016**, 145, 164109.

(163) Chakraborty, R.; Mazziotti, D. A. Generalized Pauli Conditions on the Spectra of One-Electron Reduced Density Matrices of Atoms and Molecules. *Phys. Rev. A: At., Mol., Opt. Phys.* **2014**, *89*, No. 042505.

(164) Chakraborty, R.; Mazziotti, D. A. Structure of the One-Electron Reduced Density Matrix from the Generalized Pauli Exclusion Principle. *Int. J. Quantum Chem.* **2015**, *115*, 1305–1310.

(165) Chakraborty, R.; Mazziotti, D. A. Role of the Generalized Pauli Constraints in the Quantum Chemistry of Excited States. *Int. J. Quantum Chem.* **2016**, *116*, 784–790.

(166) Legeza, O.; Schilling, C. Role of the Pair Potential for the Saturation of Generalized Pauli Constraints. *Phys. Rev. A: At., Mol., Opt. Phys.* **2018**, *97*, No. 052105.

(167) Chakraborty, R.; Mazziotti, D. A. Sparsity of the Wavefunction from the Generalized Pauli Exclusion Principle. *J. Chem. Phys.* **2018**, *148*, No. 054106.

(168) Schilling, C.; Altunbulak, M.; Knecht, S.; Lopes, A.; Whitfield, J. D.; Christandl, M.; Gross, D.; Reiher, M. Generalized Pauli Constraints in Small Atoms. *Phys. Rev. A: At., Mol., Opt. Phys.* **2018**, *97*, No. 052503. (169) Boyn, J.-N.; Mazziotti, D. A. Sparse Non-Orthogonal Wave Function Expansions from the Extension of the Generalized Pauli Constraints to the Two-Electron Reduced Density Matrix. J. Chem. Phys. **2019**, *150*, 144102.

(170) Smart, S. E.; Schuster, D. I; Mazziotti, D. A. Experimental Data from a Quantum Computer Verifies the Generalized Pauli Exclusion Principle. *Commun. Phys.* **2019**, *2*, 11.

(171) Garrod, C.; Percus, J. K. Reduction of the *N*-Particle Variational Problem. *J. Math. Phys.* **1964**, *5*, 1756–1776.

(172) DePrince, A. E.; Mazziotti, D. A. Parametric Approach to Variational Two-Electron Reduced-Density-Matrix Theory. *Phys. Rev.* A: At., Mol., Opt. Phys. **2007**, 76, No. 042501.

(173) DePrince, A. E., III; Mazziotti, D. A. Molecular Geometries and Harmonic Frequencies from the Parametric Two-Electron Reduced Density Matrix Method with Application to the HCN \leftrightarrow HNC Isomerization. J. Phys. Chem. B 2008, 112, 16158–16162.

(174) DePrince, A. E.; Kamarchik, E.; Mazziotti, D. A. Parametric Two-Electron Reduced-Density-Matrix Method Applied to Computing Molecular Energies and Properties at Nonequilibrium Geometries. *J. Chem. Phys.* **2008**, *128*, 234103.

(175) DePrince, A. E.; Mazziotti, D. A. Open-Shell Molecular Electronic States from the Parametric Two-Electron Reduced-Density-Matrix Method. *J. Chem. Phys.* **2009**, *130*, 164109.

(176) DePrince, A. E.; Mazziotti, D. A. Exploiting the Spatial Locality of Electron Correlation within the Parametric Two-Electron Reduced-Density-Matrix Method. *J. Chem. Phys.* **2010**, *132*, No. 034110.

(177) Schwerdtfeger, C. A.; DePrince, A. E.; Mazziotti, D. A. Testing the Parametric Two-Electron Reduced-Density-Matrix Method with Improved Functionals: Application to the Conversion of Hydrogen Peroxide to Oxywater. J. Chem. Phys. **2011**, 134, 174102.

(178) Schwerdtfeger, C. A.; Mazziotti, D. A. Treating Molecules in Arbitrary Spin States using the Parametric Two-Electron Reduced-Density-Matrix Method. *J. Chem. Phys.* **2012**, *137*, No. 034107.

(179) Hoy, E. P.; Schwerdtfeger, C. A.; Mazziotti, D. A. Isoelectronic Analogue of Oxywater: A Parametric Two-Electron Reduced-Density-Matrix Study of Ammonia Oxide. *Mol. Phys.* **2012**, *110*, 765–773.

(180) Sand, A. M.; Mazziotti, D. A. Parametric Two-Electron Reduced-Density-Matrix Method with Application to Diradical Rectangular H4. *Comput. Theor. Chem.* **2013**, 1003, 44–49.

(181) Sand, A. M.; Mazziotti, D. A. Effect of Molecular-Orbital Rotations on Ground-State Energies in the Parametric Two-Electron Reduced Density Matrix Method. J. Chem. Phys. **2013**, 138, 244102. (182) Shenvi, N.; van Aggelen, H.; Yang, Y.; Yang, W.; Schwerdtfeger, C.; Mazziotti, D. The Tensor Hypercontracted Parametric Reduced Density Matrix Algorithm: Coupled-cluster Accuracy with $O(r^{4}4)$ Scaling. *J. Chem. Phys.* **2013**, *139*, No. 054110.

(183) Hoy, E. P.; Schwerdtfeger, C. A.; Mazziotti, D. A. Relative Energies and Geometries of the cis- and trans-HO3 Radicals from the Parametric 2-Electron Density Matrix Method. *J. Phys. Chem. A* **2013**, *117*, 1817–1825.

(184) Valentine, A. J. S.; Mazziotti, D. A. Theoretical Prediction of the Structures and Energies of Olympicene and its Isomers. *J. Phys. Chem. A* **2013**, *117*, 9746–9752.

(185) McManus, A. L.; Hoy, E. P.; Mazziotti, D. A. Energies and Structures in Biradical Chemistry from the Parametric Two-Electron Reduced-Density Matrix Method: Applications to the Benzene and Cyclobutadiene Biradicals. *Phys. Chem. Chem. Phys.* **2015**, *17*, 12521– 12529.

(186) Valentine, A. J.; Mazziotti, D. A. Analytical Nuclear Derivatives for the Parametric Two-Electron Reduced Density Matrix Method. *Chem. Phys. Lett.* **2017**, 685, 300–304.

(187) Mazziotti, D. A. Anti-Hermitian Contracted Schrödinger Equation: Direct Determination of the Two-Electron Reduced Density Matrices of Many-Electron Molecules. *Phys. Rev. Lett.* **2006**, *97*, 143002.

(188) Mazziotti, D. A. Anti-Hermitian Part of the Contracted Schrödinger Equation for the Direct Calculation of Two-Electron Reduced Density Matrices. *Phys. Rev. A: At., Mol., Opt. Phys.* **2007**, *75*, No. 022505.

(189) Mazziotti, D. A. Two-Electron Reduced Density Matrices from the Anti-Hermitian Contracted Schrödinger Equation: Enhanced Energies and Properties with Larger Basis Sets. J. Chem. Phys. 2007, 126, 184101.

(190) DePrince, A. E.; Mazziotti, D. A. Cumulant Reconstruction of the Three-Electron Reduced Density Matrix in the Anti-Hermitian Contracted Schrödinger Equation. J. Chem. Phys. **2007**, *127*, 104104.

(191) Mazziotti, D. A. Energy Barriers in the Conversion of Bicyclobutane to gauche-1,3-Butadiene from the Anti-Hermitian Contracted Schrödinger Equation. J. Phys. Chem. A 2008, 112, 13684–13690.

(192) Foley, J. J.; Rothman, A. E.; Mazziotti, D. A. Activation Energies of Sigmatropic Shifts in Propene and Acetone Enolate from the Anti-Hermitian Contracted Schrödinger Equation. *J. Chem. Phys.* **2009**, *130*, 184112.

(193) Gidofalvi, G.; Mazziotti, D. A. Direct Calculation of Excited-State Electronic Energies and Two-Electron Reduced Density Matrices from the Anti-Hermitian Contracted Schrödinger Equation. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *80*, No. 022507.

(194) Rothman, A. E.; Foley, J. J.; Mazziotti, D. A. Open-Shell Energies and Two-Electron Reduced Density Matrices from the Anti-Hermitian Contracted Schrödinger Equation: A Spin-Coupled Approach. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *80*, No. 052508.

(195) Greenman, L.; Mazziotti, D. A. Energy Barriers of Vinylidene Carbene Reactions from the Anti-Hermitian Contracted Schrödinger Equation. J. Phys. Chem. A **2010**, 114, 583–588.

(196) Rothman, A. E.; Mazziotti, D. A. Nonequilibrium, Steady-State Electron Transport with *N*-Representable Density Matrices from the Anti-Hermitian Contracted Schrödinger Equation. *J. Chem. Phys.* **2010**, *132*, 104112.

(197) Foley, J. J.; Rothman, A. E.; Mazziotti, D. A. Strongly Correlated Mechanisms of a Photoexcited Radical Reaction from the Anti-Hermitian Contracted Schrödinger Equation. *J. Chem. Phys.* **2011**, *134*, No. 034111.

(198) Mayer, J. E. Electron Correlation. *Phys. Rev.* **1955**, *100*, 1579–1586.

(199) Erdahl, R.; Jin, B. In *Many-Electron Densities and Reduced Density Matrices*; Cioslowski, J., Ed.; Springer US: Boston, MA, 2000; pp 57–84.

(200) Hammond, J. R.; Mazziotti, D. A. Variational Reduced-Density-Matrix Calculation of the One-Dimensional Hubbard Model. *Phys. Rev. A: At., Mol., Opt. Phys.* **2006**, *73*, No. 062505. (201) Rothman, A. E.; Mazziotti, D. A. Variational Reduced-Density-Matrix Theory Applied to the Electronic Structure of Few-Electron Quantum Dots. *Phys. Rev. A: At, Mol., Opt. Phys.* **2008**, *78*, No. 032510.

(202) Greenman, L.; Mazziotti, D. A. Strong Electron Correlation in the Decomposition Reaction of Dioxetanone with Implications for Firefly Bioluminescence. J. Chem. Phys. **2010**, 133, 164110.

(203) Mazziotti, D. A. Large-Scale Semidefinite Programming for Many-Electron Quantum Mechanics. *Phys. Rev. Lett.* **2011**, *106*, No. 083001.

(204) Mazziotti, D. A. Two-Electron Reduced Density Matrix as the Basic Variable in Many-Electron Quantum Chemistry and Physics. *Chem. Rev.* **2012**, *112*, 244–262.

(205) Rubin, N. C.; Mazziotti, D. A. Strong Electron Correlation in Materials from Pair-Interacting Model Hamiltonians. *J. Phys. Chem. C* **2015**, *119*, 14706–14713.

(206) Garrod, C.; Mihailović, M. V.; Rosina, M. The Variational Approach to the Two-Body Density Matrix. J. Math. Phys. 1975, 16, 868–874.

(207) Mazziotti, D. A. Reduced-Density-Matrix Mechanics: With Application to Many-Electron Atoms and Molecules; John Wiley & Sons, Ltd, 2007; Chapter 3, pp 19–59.

(208) Gidofalvi, G.; Mazziotti, D. A. Molecular Properties from Variational Reduced-Density-Matrix Theory with Three-Particle *N*-Representability Conditions. *J. Chem. Phys.* **2007**, *126*, No. 024105.

(209) Verstichel, B.; van Aggelen, H.; Van Neck, D.; Ayers, P. W.; Bultinck, P. Variational Determination of the Second-Order Density Matrix for the Isoelectronic Series of Beryllium, Neon, and Silicon. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *80*, No. 032508.

(210) van Aggelen, H.; Verstichel, B.; Bultinck, P.; Van Neck, D.; Ayers, P. W.; Cooper, D. L. Chemical Verification of Variational Second-Order Density Matrix Based Potential Energy Surfaces for the N_2 Isoelectronic Series. J. Chem. Phys. **2010**, 132, 114112.

(211) Verstichel, B.; van Aggelen, H.; Van Neck, D.; Ayers, P. W.; Bultinck, P. Subsystem Constraints in Variational Second Order Density Matrix Optimization: Curing the Dissociative Behavior. *J. Chem. Phys.* **2010**, *132*, 114113.

(212) Mazziotti, D. A. Variational Two-Electron Reduced Density Matrix Theory for Many-Electron Atoms and Molecules: Implementation of the Spin- and Symmetry-Adapted T_2 Condition Through First-Order Semidefinite Programming. *Phys. Rev. A: At., Mol., Opt. Phys.* **2005**, 72, No. 032510.

(213) Vandenberghe, L.; Boyd, S. Semidefinite Programming. SIAM Rev. 1996, 38, 49–95.

(214) Erdahl, R. M. In *Reduced-Density-Matrix Mechanics With Applications to Many-electron Atoms and Molecules*; Mazziotti, D. A., Ed.; Advances in Chemical Physics; Wiley: New York, 2007; Chapter 3.

(215) Hemmatiyan, S.; Sajjan, M.; Schlimgen, A. W.; Mazziotti, D. A. Excited-State Spectra of Strongly Correlated Molecules from a Reduced-Density-Matrix Approach. *J. Phys. Chem. Lett.* **2018**, *9*, 5373–5378.

(216) Alcoba, D. R.; Torre, A.; Lain, L.; Massaccesi, G. E.; Oña, O. B.; Honoré, E. M.; Poelmans, W.; Van Neck, D.; Bultinck, P.; De Baerdemacker, S. Direct Variational Determination of the Two-Electron Reduced Density Matrix for Doubly Occupied-Configuration-Interaction Wave Functions: The Influence of Three-Index *N*-Representability Conditions. *J. Chem. Phys.* **2018**, *148*, No. 024105.

(217) Gidofalvi, G.; Mazziotti, D. A. Active-Space Two-Electron Reduced-Density-Matrix Method: Complete Active-Space Calculations without Diagonalization of the *N*-Electron Hamiltonian. *J. Chem. Phys.* **2008**, *129*, 134108.

(218) Greenman, L.; Mazziotti, D. A. Highly Multireferenced Arynes Studied with Large Active Spaces using Two-Electron Reduced Density Matrices. *J. Chem. Phys.* **2009**, *130*, 184101.

(219) Poelmans, W.; Van Raemdonck, M.; Verstichel, B.; De Baerdemacker, S.; Torre, A.; Lain, L.; Massaccesi, G. E.; Alcoba, D. R.; Bultinck, P.; Van Neck, D. Variational Optimization of the Second-Order Density Matrix Corresponding to a Seniority-Zero Configuration Interaction Wave Function. *J. Chem. Theory Comput.* **2015**, *11*, 4064–4076.

(220) Head-Marsden, K.; Mazziotti, D. A. Pair 2-Electron Reduced Density Matrix Theory using Localized Orbitals. J. Chem. Phys. 2017, 147, No. 084101.

(221) Alcoba, D. R.; Torre, A.; Lain, L.; Massaccesi, G. E.; Oña, O. B.; Ríos, E. Unrestricted Treatment for the Direct Variational Determination of the Two-Electron Reduced Density Matrix for Doubly Occupied-Configuration-Interaction Wave Functions. *J. Chem. Phys.* **2019**, *150*, 164106.

(222) Valentine, A. J. S.; Talapin, D. V.; Mazziotti, D. A. Orbitals, Occupation Numbers, and Band Structure of Short One-Dimensional Cadmium Telluride Polymers. *J. Phys. Chem. A* **2017**, *121*, 3142–3147.

(223) Maradzike, E.; Gidofalvi, G.; Turney, J. M.; Schaefer, H. F.; DePrince, A. E. Analytic Energy Gradients for Variational Two-Electron Reduced-Density-Matrix-Driven Complete Active Space Self-Consistent Field Theory. J. Chem. Theory Comput. 2017, 13, 4113– 4122.

(224) Montgomery, J. M.; Mazziotti, D. A. Strong Electron Correlation in Nitrogenase Cofactor, FeMoco. J. Phys. Chem. A 2018, 122, 4988–4996.

(225) Hoy, E. P.; Mazziotti, D. A.; Seideman, T. Development and Application of a 2-Electron Reduced Density Matrix Approach to Electron Transport via Molecular Junctions. *J. Chem. Phys.* **2017**, *147*, 184110.

(226) Sajjan, M.; Mazziotti, D. A. Current-Constrained Density-Matrix Theory to Calculate Molecular Conductivity with Increased Accuracy. *Commun. Chem.* **2018**, *1*, 31.

(227) Raeber, A. E.; Mazziotti, D. A. Current-Constrained One-Electron Reduced Density-Matrix Theory for Non-Equilibrium Steady-State Molecular Conductivity. *Phys. Chem. Chem. Phys.* **2019**, *21*, 12620–12624.

(228) Raeber, A. E.; Mazziotti, D. A. Non-Equilibrium Steady State Conductivity in Cyclo[18]carbon and Its Boron Nitride Analogue. 2020; arXiv:2008.06028 [physics.chem-ph]; https://arxiv.org/abs/ 2008.06028.

(229) Safaei, S.; Mazziotti, D. A. Quantum Signature of Exciton Condensation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *98*, No. 045122.

(230) Ewing, S.; Mazziotti, D. A. Strong Correlation in Molecular Periodic Systems from a Variational Reduced Density Matrix Theory. 2020; arXiv:2008.02779 [physics.chem-ph]; https://arxiv.org/abs/ 2008.02779.

(231) Mazziotti, D. A. Extraction of Electronic Excited States from the Ground-State Two-Particle Reduced Density Matrix. *Phys. Rev. A: At., Mol., Opt. Phys.* **2003**, *68*, No. 052501.

(232) Farnum, J. D.; Mazziotti, D. A. Extraction of Ionization Energies from the Ground-State Two-Particle Reduced Density Matrix. *Chem. Phys. Lett.* **2004**, 400, 90–93.

(233) Greenman, L.; Mazziotti, D. A. Electronic Excited-State Energies from a Linear Response Theory Based on the Ground-State Two-Electron Reduced Density Matrix. *J. Chem. Phys.* **2008**, *128*, 114109.

(234) van Aggelen, H.; Verstichel, B.; Acke, G.; Degroote, M.; Bultinck, P.; Ayers, P. W.; Van Neck, D. Extended Random Phase Approximation Method for Atomic Excitation Energies from Correlated and Variationally Optimized Second-Order Density Matrices. *Comput. Theor. Chem.* **2013**, 1003, 50–54.

(235) Bouten, M.; Van Leuven, P.; Mihailović, M.; Rosina, M. A New Particle-Hole Approach to Collective States. *Nucl. Phys. A* **1973**, *202*, 127–144.

(236) Mihailović, M.; Rosina, M. The Variational Approach to the Density Matrix for Light Nuclei. *Nucl. Phys. A* **1975**, *237*, 221–228.

(237) Rosina, M. Application of the Two-Body Density Matrix of the Ground State for Calculations of Some Excited States. *Int. J. Quantum Chem.* **1978**, *13*, 737–742.

(238) Valdemoro, C.; Alcoba, D. R.; Ona, O. B.; Tel, L. M.; Perez-Romero, E. Combining the G-Particle-Hole Hypervirial Equation and the Hermitian Operator Method to Study Electronic Excitations and De-excitations. J. Math. Chem. **2012**, *50*, 492–509. (239) Woodford, C. R.; Frady, E. P.; Smith, R. S.; Morey, B.; Canzi, G.; Palida, S. F.; Araneda, R. C.; Kristan, W. B.; Kubiak, C. P.; Miller, E. W.; et al. Improved PeT Molecules for Optically Sensing Voltage in Neurons. *J. Am. Chem. Soc.* **2015**, *137*, 1817–1824.

(240) Hemmatiyan, S.; Mazziotti, D. A. Unraveling the Band Gap Trend in the Narrowest Graphene Nanoribbons from the Spin-Adapted Excited-Spectra Reduced Density Matrix Method. *J. Phys. Chem. C* **2019**, *123*, 14619–14624.

(241) Weinhold, F.; Wilson, E. B. Reduced Density Matrices of Atoms and Molecules. II. On the *N*-Representability Problem. *J. Chem. Phys.* **1967**, *47*, 2298–2311.

(242) Alcoba, D. R.; Torre, A.; Lain, L.; Oña, O. B.; Ríos, E.; Massaccesi, G. E. Incorporating Dynamic Correlation into the Variational Determination Method of the Second-Order Reduced Density Matrix in the Doubly Occupied Configuration Interaction Space. Int. J. Quantum Chem. **2020**, 120, No. e26256.

(243) Head-Marsden, K.; Mazziotti, D. A. Active-Space Pair Two-Electron Reduced Density Matrix Theory for Strong Correlation. *J. Phys. Chem. A* **2020**, *124*, 4848–4854.

(244) Oña, O. B.; Torre, A.; Lain, L.; Alcoba, D. R.; Ríos, E.; Massaccesi, G. E. Variational Determination of the Two-Electron Reduced Density Matrix within the Doubly Occupied Configuration Interaction Scheme: An Extension to the Study of Open-Shell Systems. *J. Chem. Phys.* **2020**, *153*, No. 084101.

(245) Alcoba, D. R.; Capuzzi, P.; Rubio-García, A.; Dukelsky, J.; Massaccesi, G. E.; Oña, O. B.; Torre, A.; Lain, L. Variational Reduced Density Matrix Method in the Doubly Occupied Configuration Interaction Space using Three-Particle N-Representability Conditions. *J. Chem. Phys.* **2018**, *149*, 194105.

(246) Rubio-García, A.; Dukelsky, J.; Alcoba, D. R.; Capuzzi, P.; Oña, O. B.; Ríos, E.; Torre, A.; Lain, L. Variational Reduced Density Matrix Method in the Doubly-Occupied Configuration Interaction Space using Four-Particle N-Representability Conditions: Application to the XXZ Model of Quantum Magnetism. *J. Chem. Phys.* **2019**, *151*, 154104.

(247) Smart, S. E.; Mazziotti, D. A. Quantum-Classical Hybrid Algorithm using an Error-Mitigating N-Representability Condition to Compute the Mott Metal-Insulator Transition. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *100*, No. 022517.

(248) Rubin, N. C.; Babbush, R.; McClean, J. Application of Fermionic Marginal Constraints to Hybrid Quantum Algorithms. *New J. Phys.* **2018**, *20*, No. 053020.

(249) Pauli, W. Über den Zusammenhang des Abschlusses der Elektronengruppen im Atom mit der Komplexstruktur der Spektren. *Z. Phys.* **1925**, *31*, 765–783.

(250) Smith, D. W. N-Representability Problem for Fermion Density Matrices. II. The First-Order Density Matrix with N Even. *Phys. Rev.* **1966**, 147, 896–898.

(251) Borland, R. E.; Dennis, K. The Conditions on the One-Matrix for Three-Body Fermion Wavefunctions with One-Rank Equal to Six. *J. Phys. B: At. Mol. Phys.* **1972**, *5*, 7–15.

(252) Klyachko, A. A. Quantum Marginal Problem and N-Representability. J. Phys.: Conf. Ser. 2006, 36, 72-86.

(253) Altunbulak, M.; Klyachko, A. The Pauli Principle Revisited. Commun. Math. Phys. 2008, 282, 287–322.

(254) Schilling, C.; Gross, D.; Christandl, M. Pinning of Fermionic Occupation Numbers. *Phys. Rev. Lett.* **2013**, *110*, No. 040404.

(255) Schilling, C. Quasipinning and its Relevance for N-Fermion Quantum States. *Phys. Rev. A: At., Mol., Opt. Phys.* **2015**, *91*, No. 022105.

(256) Theophilou, I.; Lathiotakis, N. N.; Marques, M. A. L.; Helbig, N. Generalized Pauli Constraints in Reduced Density Matrix Functional Theory. J. Chem. Phys. **2015**, *142*, 154108.

(257) Benavides-Riveros, C. L.; Springborg, M. Quasipinning and Selection Rules for Excitations in Atoms and Molecules. *Phys. Rev. A: At., Mol., Opt. Phys.* **2015**, *92*, No. 012512.

(258) Schilling, C. Hubbard Model: Pinning of Occupation Numbers and Role of Symmetries. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, 92, 155149. (259) Tennie, F.; Vedral, V.; Schilling, C. Influence of the Fermionic Exchange Symmetry Beyond Pauli's Exclusion Principle. *Phys. Rev. A: At., Mol., Opt. Phys.* **2017**, *95*, No. 022336.

(260) Rabitz, H. A.; Hsieh, M. M.; Rosenthal, C. M. Quantum Optimally Controlled Transition Landscapes. *Science* **2004**, *303*, 1998–2001.

(261) Walter, M.; Doran, B.; Gross, D.; Christandl, M. Entanglement Polytopes: Multiparticle Entanglement from Single-Particle Information. *Science* **2013**, *340*, 1205–1208.

(262) Pernal, K.; Giesbertz, K. J. H. In *Density-Functional Methods for Excited States*; Ferré, N., Filatov, M., Huix-Rotllant, M., Eds.; Springer International Publishing: Cham, 2016; pp 125–183.

(263) Helbig, N.; Theodorakopoulos, G.; Lathiotakis, N. N. Fractional Spin in Reduced Density-Matrix Functional Theory. J. Chem. Phys. 2011, 135, No. 054109.

(264) Lathiotakis, N. N.; Marques, M. A. L. Benchmark Calculations for Reduced Density-Matrix Functional Theory. *J. Chem. Phys.* **2008**, *128*, 184103.

(265) Pernal, K. Effective Potential for Natural Spin Orbitals. *Phys. Rev. Lett.* **2005**, *94*, 233002.

(266) Piris, M.; Ugalde, J. M. Iterative Diagonalization for Orbital Optimization in Natural Orbital Functional Theory. *J. Comput. Chem.* **2009**, *30*, 2078–2086.

(267) Müller, A. Explicit Approximate Relation between Reduced Two- and One-Particle Density Matrices. *Phys. Lett. A* **1984**, *105*, 446–452.

(268) Buijse, M. A.; Baerends, E. J. An Approximate Exchange-Correlation Hole Density as a Functional of the Natural Orbitals. *Mol. Phys.* **2002**, *100*, 401–421.

(269) Gritsenko, O.; Pernal, K.; Baerends, E. J. An Improved Density Matrix Functional by Physically Motivated Repulsive Corrections. *J. Chem. Phys.* **2005**, *122*, 204102.

(270) Piris, M. A New Approach for the Two-Electron Cumulant in Natural Orbital Functional Theory. *Int. J. Quantum Chem.* **2006**, *106*, 1093–1104.

(271) Lathiotakis, N. N.; Sharma, S.; Dewhurst, J. K.; Eich, F. G.; Marques, M. A. L.; Gross, E. K. U. Density-Matrix-Power Functional: Performance for Finite Systems and the Homogeneous Electron Gas. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *79*, No. 040501.

(272) Lathiotakis, N.; Helbig, N.; Zacarias, A.; Gross, E. A Functional of the One-Body-Reduced Density Matrix Derived from the Homogeneous Electron Gas: Performance for Finite Systems. *J. Chem. Phys.* **2009**, *130*, No. 064109.

(273) Csányi, G.; Arias, T. A. Tensor Product Expansions for Correlation in Quantum Many-Body Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2000**, *61*, 7348–7352.

(274) Goedecker, S.; Umrigar, C. J. Natural Orbital Functional for the Many-Electron Problem. *Phys. Rev. Lett.* **1998**, *81*, 866–869.

(275) Rohr, D. R.; Pernal, K.; Gritsenko, O. V.; Baerends, E. J. A Density Matrix Functional with Occupation Number Driven Treatment of Dynamical and Nondynamical Correlation. *J. Chem. Phys.* **2008**, *129*, 164105.

(276) Rohr, D. R.; Toulouse, J.; Pernal, K. Combining Density-Functional Theory and Density-Matrix-Functional Theory. *Phys. Rev.* A: At., Mol., Opt. Phys. **2010**, 82, No. 052502.

(277) Sharma, S.; Dewhurst, J. K.; Lathiotakis, N. N.; Gross, E. K. U. Reduced Density Matrix Functional for Many-Electron Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2008**, *78*, 201103.

(278) Helbig, N.; Lathiotakis, N. N.; Gross, E. K. U. Discontinuity of the Chemical Potential in Reduced-Density-Matrix-Functional Theory for Open-Shell Systems. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *79*, No. 022504.

(279) Sharma, S.; Dewhurst, J. K.; Shallcross, S.; Gross, E. K. U. Spectral Density and Metal-Insulator Phase Transition in Mott Insulators within Reduced Density Matrix Functional Theory. *Phys. Rev. Lett.* **2013**, *110*, 116403.

(280) Di Sabatino, S.; Berger, J. A.; Reining, L.; Romaniello, P. Photoemission Spectra from Reduced Density Matrices: The Band Gap

in Strongly Correlated Systems. Phys. Rev. B: Condens. Matter Mater. Phys. 2016, 94, 155141.

(281) Theophilou, I.; Lathiotakis, N. N.; Helbig, N. Conditions for Describing Triplet States in Reduced Density Matrix Functional Theory. *J. Chem. Theory Comput.* **2016**, *12*, 2668–2678.

(282) Lathiotakis, N. N.; Helbig, N.; Rubio, A.; Gidopoulos, N. I. Local Reduced-Density-Matrix-Functional Theory: Incorporating Static Correlation Effects in Kohn-Sham Equations. *Phys. Rev. A: At., Mol., Opt. Phys.* **2014**, *90*, No. 032511.

(283) Theophilou, I.; Lathiotakis, N. N.; Gidopoulos, N. I.; Rubio, A.; Helbig, N. Orbitals from Local RDMFT: Are They Kohn-Sham or Natural Orbitals? *J. Chem. Phys.* **2015**, *143*, No. 054106.

(284) Pernal, K.; Gritsenko, O.; Baerends, E. J. Time-Dependent Density-Matrix-Functional Theory. *Phys. Rev. A: At., Mol., Opt. Phys.* **2007**, *75*, No. 012506.

(285) Pernal, K.; Giesbertz, K.; Gritsenko, O.; Baerends, E. J. Adiabatic Approximation of Time-Dependent Density Matrix Functional Response Theory. J. Chem. Phys. **200**7, 127, 214101.

(286) Pernal, K.; Cioslowski, J. Frequency-Dependent Response Properties and Excitation Energies from One-Electron Density Matrix Functionals. *Phys. Chem. Chem. Phys.* **2007**, *9*, 5956–5965.

(287) Hohenberg, P.; Kohn, W. Inhomogeneous Electron Gas. *Phys. Rev.* **1964**, *136*, B864–871.

(288) Jones, R. O. Density Functional Theory: Its Origins, Rise to Prominence, and Future. *Rev. Mod. Phys.* **2015**, *87*, 897–923.

(289) Marques, M. A.; Oliveira, M. J.; Burnus, T. Libxc: A Library of Exchange and Correlation Functionals for Density Functional Theory. *Comput. Phys. Commun.* **2012**, *183*, 2272–2281.

(290) Lehtola, S.; Steigemann, C.; Oliveira, M. J.; Marques, M. A. Recent Developments in Libxc - A Comprehensive Library of Functionals for Density Functional Theory. *SoftwareX* **2018**, *7*, 1–5.

(291) Kohn, W.; Sham, L. J. Self-Consistent Equations Including Exchange and Correlation Effects. *Phys. Rev.* **1965**, *140*, A1133–1138.

(292) Perdew, J. P.; Zunger, A. Self-Interaction Correction to Density-Functional Approximations for Many-Electron Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **1981**, 23, 5048–5079.

(293) Perdew, J. P.; Burke, K.; Ernzerhof, M. Generalized Gradient Approximation Made Simple. *Phys. Rev. Lett.* **1996**, *77*, 3865–3868.

(294) Perdew, J. P.; Ruzsinszky, A.; Csonka, G. I.; Vydrov, O. A.; Scuseria, G. E.; Constantin, L. A.; Zhou, X.; Burke, K. Restoring the Density-Gradient Expansion for Exchange in Solids and Surfaces. *Phys. Rev. Lett.* **2008**, *100*, 136406.

(295) Tsuneda, T. Density Functional Theory in Quantum Chemistry; Springer Japan: Tokyo, 2014.

(296) Sun, J.; Ruzsinszky, A.; Perdew, J. P. Strongly Constrained and Appropriately Normed Semilocal Density Functional. *Phys. Rev. Lett.* **2015**, *115*, No. 036402.

(297) Heyd, J.; Scuseria, G. E.; Ernzerhof, M. Hybrid Functionals based on a Screened Coulomb Potential. J. Chem. Phys. 2003, 118, 8207–8215.

(298) Adamo, C.; Barone, V. Toward Reliable Density Functional Methods without Adjustable Parameters: The PBE0 Model. *J. Chem. Phys.* **1999**, *110*, 6158–6170.

(299) Grimme, S. Semiempirical GGA-Type Density Functional Constructed with a Long-Range Dispersion Correction. J. Comput. Chem. 2006, 27, 1787–1799.

(300) Tkatchenko, A.; Scheffler, M. Accurate Molecular Van Der Waals Interactions from Ground-State Electron Density and Free-Atom Reference Data. *Phys. Rev. Lett.* **2009**, *102*, No. 073005.

(301) Grimme, S.; Antony, J.; Ehrlich, S.; Krieg, H. A Consistent and Accurate Ab Initio Parametrization of Density Functional Dispersion Correction (DFT-D) for the 94 Elements H-Pu. *J. Chem. Phys.* **2010**, 132, 154104.

(302) Vydrov, O. A.; Van Voorhis, T. Nonlocal van der Waals Density Functional: The Simpler the Better. *J. Chem. Phys.* **2010**, *133*, 244103. (303) Gerber, I. C.; Ángyán, J. G.; Marsman, M.; Kresse, G. Range Separated Hybrid Density Functional with Long-Range Hartree-Fock Exchange Applied to Solids. *J. Chem. Phys.* **2007**, *127*, No. 054101. (304) Grimme, S. Semiempirical Hybrid Density Functional with Perturbative Second-Order Correlation. J. Chem. Phys. 2006, 124, No. 034108.

(305) Trusheim, M. E.; Wan, N. H.; Chen, K. C.; Ciccarino, C. J.; Flick, J.; Sundararaman, R.; Malladi, G.; Bersin, E.; Walsh, M.; Lienhard, B.; et al. Lead-Related Quantum Emitters in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2019**, *99*, No. 075430.

(306) Harris, I.; Ciccarino, C. J.; Flick, J.; Englund, D. R.; Narang, P. Group III Quantum Defects in Diamond are Stable Spin-1 Color Centers. 2019; arXiv:1907.12548 [quant-ph]; https://arxiv.org/abs/1907.12548.

(307) Hayee, F.; Yu, L.; Zhang, J. L.; Ciccarino, C. J.; Nguyen, M.; Marshall, A. F.; Aharonovich, I.; Vučković, J.; Narang, P.; Heinz, T. F.; et al. Revealing Multiple Classes of Stable Quantum Emitters in Hexagonal Boron Nitride with Correlated Optical and Electron Microscopy. *Nat. Mater.* **2020**, *19*, 534–539.

(308) Ciccarino, C. J.; Flick, J.; Harris, I. B.; Trusheim, M. E.; Englund, D. R.; Narang, P. Strong Spin-Orbit Quenching via the Product Jahn-Teller Effect in Neutral Group IV Artificial Atom Qubits in Diamond. 2020; arXiv:2001.07743 [quant-ph]; https://arxiv.org/abs/2001.07743.

(309) Kagan, C. R.; Bassett, L. C.; Murray, C. B.; Thompson, S. M. Colloidal Quantum Dots As Platforms for Quantum Information Science. *Chem. Rev.* 2021, *121*. DOI: 10.1021/acs.chemrev.0c00831.

(310) Son, N. T.; Carlsson, P.; ul Hassan, J.; Janzén, E.; Umeda, T.; Isoya, J.; Gali, A.; Bockstedte, M.; Morishita, N.; Ohshima, T.; et al. Divacancy in 4H-SiC. *Phys. Rev. Lett.* **2006**, *96*, No. 055501.

(311) Umeda, T.; Son, N. T.; Isoya, J.; Janzén, E.; Ohshima, T.; Morishita, N.; Itoh, H.; Gali, A.; Bockstedte, M. Identification of the Carbon Antisite-Vacancy Pair in 4H-SiC. *Phys. Rev. Lett.* **2006**, *96*, 145501.

(312) Castelletto, S.; Johnson, B. C.; Ivády, V.; Stavrias, N.; Umeda, T.; Gali, A.; Ohshima, T. A Silicon Carbide Room-Temperature Single-Photon Source. *Nat. Mater.* **2014**, *13*, 151–156.

(313) Falk, A. L.; Klimov, P. V.; Buckley, B. B.; Ivády, V.; Abrikosov, I. A.; Calusine, G.; Koehl, W. F.; Gali, A.; Awschalom, D. D. Electrically and Mechanically Tunable Electron Spins in Silicon Carbide Color Centers. *Phys. Rev. Lett.* **2014**, *112*, 187601.

(314) Klimov, P. V.; Falk, A. L.; Buckley, B. B.; Awschalom, D. D. Electrically Driven Spin Resonance in Silicon Carbide Color Centers. *Phys. Rev. Lett.* **2014**, *112*, No. 087601.

(315) Falk, A. L.; Klimov, P. V.; Ivády, V.; Szász, K.; Christle, D. J.; Koehl, W. F.; Gali, A.; Awschalom, D. D. Optical Polarization of Nuclear Spins in Silicon Carbide. *Phys. Rev. Lett.* **2015**, *114*, 247603.

(316) Ivády, V.; Szász, K.; Falk, A. L.; Klimov, P. V.; Christle, D. J.; Janzén, E.; Abrikosov, I. A.; Awschalom, D. D.; Gali, A. Theoretical Model of Dynamic Spin Polarization of Nuclei Coupled to Paramagnetic Point Defects in Diamond and Silicon Carbide. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, *92*, 115206.

(317) Lohrmann, A.; Iwamoto, N.; Bodrog, Z.; Castelletto, S.; Ohshima, T.; Karle, T. J.; Gali, A.; Prawer, S.; McCallum, J. C.; Johnson, B. C. Single-Photon Emitting Diode in Silicon Carbide. *Nat. Commun.* **2015**, *6*, 7783.

(318) Widmann, M.; Lee, S.-Y.; Rendler, T.; Son, N. T.; Fedder, H.; Paik, S.; Yang, L.-P.; Zhao, N.; Yang, S.; Booker, I.; et al. Coherent Control of Single Spins in Silicon Carbide at Room Temperature. *Nat. Mater.* **2015**, *14*, 164–168.

(319) Ivády, V.; Klimov, P. V.; Miao, K. C.; Falk, A. L.; Christle, D. J.; Szász, K.; Abrikosov, I. A.; Awschalom, D. D.; Gali, A. High-Fidelity Bidirectional Nuclear Qubit Initialization in SiC. *Phys. Rev. Lett.* **2016**, *117*, 220503.

(320) Seo, H.; Falk, A. L.; Klimov, P. V.; Miao, K. C.; Galli, G.; Awschalom, D. D. Quantum Decoherence Dynamics of Divacancy Spins in Silicon Carbide. *Nat. Commun.* **2016**, *7*, 12935.

(321) de las Casas, C. F.; Christle, D. J.; Ul Hassan, J.; Ohshima, T.; Son, N. T.; Awschalom, D. D. Stark Tuning and Electrical Charge State Control of Single Divacancies in Silicon Carbide. *Appl. Phys. Lett.* **2017**, *111*, 262403. (322) Wolfowicz, G.; Anderson, C. P.; Yeats, A. L.; Whiteley, S. J.; Niklas, J.; Poluektov, O. G.; Heremans, F. J.; Awschalom, D. D. Optical Charge State Control of Spin Defects in 4H-SiC. *Nat. Commun.* **2017**, *8*, 1876.

(323) Christle, D. J.; Klimov, P. V.; de las Casas, C. F.; Szász, K.; Ivády, V.; Jokubavicius, V.; Ul Hassan, J.; Syväjärvi, M.; Koehl, W. F.; et al. Isolated Spin Qubits in SiC with a High-Fidelity Infrared Spin-to-Photon Interface. *Phys. Rev. X* **2017**, *7*, No. 021046.

(324) Udvarhelyi, P.; Thiering, G.; Londero, E.; Gali, A. Ab Initio Theory of the N_2V Defect in Diamond for Quantum Memory Implementation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2017**, *96*, 155211.

(325) Csóré, A.; von Bardeleben, H. J.; Cantin, J. L.; Gali, A. Characterization and Formation of NV centers in 3*C*,4*H*, and 6*H* SiC: An Ab Initio Study. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2017**, *96*, No. 085204.

(326) Wolfowicz, G.; Whiteley, S. J.; Awschalom, D. D. Electrometry by Optical Charge Conversion of Deep Defects in 4H-SiC. *Proc. Natl. Acad. Sci. U. S. A.* **2018**, *115*, 7879–7883.

(327) Whiteley, S. J.; Wolfowicz, G.; Anderson, C. P.; Bourassa, A.; Ma, H.; Ye, M.; Koolstra, G.; Satzinger, K. J.; Holt, M. V.; Heremans, F. J.; et al. Spin–Phonon Interactions in Silicon Carbide Addressed by Gaussian acoustics. *Nat. Phys.* **2019**, *15*, 490–495.

(328) Anderson, C. P.; Bourassa, A.; Miao, K. C.; Wolfowicz, G.; Mintun, P. J.; Crook, A. L.; Abe, H.; Ul Hassan, J.; Son, N. T.; Ohshima, T.; Awschalom, D. D. Electrical and Optical Control of Single Spins Integrated in Scalable Semiconductor Devices. *Science* **2019**, *366*, 1225–1230.

(329) Miao, K. C.; Bourassa, A.; Anderson, C. P.; Whiteley, S. J.; Crook, A. L.; Bayliss, S. L.; Wolfowicz, G.; Thiering, G.; Udvarhelyi, P.; Ivády, V.; et al. Electrically Driven Optical Interferometry with Spins in Silicon Carbide. *Sci. Adv.* **2019**, *5*, No. eaay0527.

(330) Nagy, R.; Niethammer, M.; Widmann, M.; Chen, Y.-C.; Udvarhelyi, P.; Bonato, C.; Hassan, J. U.; Karhu, R.; Ivanov, I. G.; Son, N. T.; et al. High-Fidelity Spin and Optical Control of Single Silicon-Vacancy Centres in Silicon Carbide. *Nat. Commun.* **2019**, *10*, 1954.

(331) Crook, A. L.; Anderson, C. P.; Miao, K. C.; Bourassa, A.; Lee, H.; Bayliss, S. L.; Bracher, D. O.; Zhang, X.; Abe, H.; Ohshima, T.; et al. Purcell Enhancement of a Single Silicon Carbide Color Center with Coherent Spin Control. *Nano Lett.* **2020**, *20*, 3427–3434.

(332) Wolfowicz, G.; Anderson, C. P.; Diler, B.; Poluektov, O. G.; Heremans, F. J.; Awschalom, D. D. Vanadium Spin Qubits as Telecom Quantum Emitters in Silicon Carbide. *Sci. Adv.* **2020**, *6*, No. eaaz1192.

(333) Castelletto, S.; Boretti, A. Silicon Carbide Color Centers for Quantum Applications. J. Phys. Photonics **2020**, 2, No. 022001.

(334) Diler, B.; Whiteley, S. J.; Anderson, C. P.; Wolfowicz, G.; Wesson, M. E.; Bielejec, E. S.; Joseph Heremans, F.; Awschalom, D. D. Coherent Control and High-Fidelity Readout of Chromium Ions in Commercial Silicon Carbide. *npj Quantum Inf.* **2020**, *6*, 11.

(335) Haldar, S.; Vovusha, H.; Yadav, M. K.; Eriksson, O.; Sanyal, B. Systematic Study of Structural, Electronic, and Optical Properties of Atomic-Scale Defects in the Two-Dimensional Transition Metal Dichalcogenides MX_2 ($M = M_0$, W; X = S, Se, Te). *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, *92*, 235408.

(336) Hong, J.; Hu, Z.; Probert, M.; Li, K.; Lv, D.; Yang, X.; Gu, L.; Mao, N.; Feng, Q.; Xie, L.; et al. Exploring Atomic Defects in Molybdenum Disulphide Monolayers. *Nat. Commun.* **2015**, *6*, 6293.

(337) He, Y.-M.; Clark, G.; Schaibley, J. R.; He, Y.; Chen, M.-C.; Wei, Y.-J.; Ding, X.; Zhang, Q.; Yao, W.; Xu, X.; et al. Single Quantum Emitters in Monolayer Semiconductors. *Nat. Nanotechnol.* **2015**, *10*, 497–502.

(338) Srivastava, A.; Sidler, M.; Allain, A. V.; Lembke, D. S.; Kis, A.; Imamoğlu, A. Optically Active Quantum Dots in Monolayer WSe 2. *Nat. Nanotechnol.* **2015**, *10*, 491–496.

(339) Chakraborty, C.; Kinnischtzke, L.; Goodfellow, K. M.; Beams, R.; Vamivakas, A. N. Voltage-Controlled Quantum Light From an Atomically Thin Semiconductor. *Nat. Nanotechnol.* **2015**, *10*, 507–511. (340) Palacios-Berraquero, C.; Barbone, M.; Kara, D. M.; Chen, X.; Goykhman, I.; Yoon, D.; Ott, A. K.; Beitner, J.; Watanabe, K.; Taniguchi, T.; et al. Atomically Thin Quantum Light-Emitting Diodes. *Nat. Commun.* **2016**, *7*, 12978.

(341) Jeong, H. Y.; Lee, S. Y.; Ly, T. H.; Han, G. H.; Kim, H.; Nam, H.; Jiong, Z.; Shin, B. G.; Yun, S. J.; Kim, J.; et al. Visualizing Point Defects in Transition-Metal Dichalcogenides Using Optical Microscopy. *ACS Nano* **2016**, *10*, 770–777.

(342) Palacios-Berraquero, C.; Kara, D. M.; Montblanch, A. R.-P.; Barbone, M.; Latawiec, P.; Yoon, D.; Ott, A. K.; Lončar, M.; Ferrari, A. C.; Atatüre, M. Large-Scale Quantum-Emitter Arrays in Atomically Thin Semiconductors. *Nat. Commun.* **2017**, *8*, 15093.

(343) Branny, A.; Kumar, S.; Proux, R.; Gerardot, B. D. Deterministic Strain-Induced Arrays of Quantum Emitters in a Two-Dimensional Semiconductor. *Nat. Commun.* **2017**, *8*, 15053.

(344) Chakraborty, C.; Goodfellow, K. M.; Dhara, S.; Yoshimura, A.; Meunier, V.; Vamivakas, A. N. Quantum-Confined Stark Effect of Individual Defects in a van der Waals Heterostructure. *Nano Lett.* **2017**, *17*, 2253–2258.

(345) Moody, G.; Tran, K.; Lu, X.; Autry, T.; Fraser, J. M.; Mirin, R. P.; Yang, L.; Li, X.; Silverman, K. L. Microsecond Valley Lifetime of Defect-Bound Excitons in Monolayer WSe₂. *Phys. Rev. Lett.* **2018**, *121*, No. 057403.

(346) Briggs, N.; Subramanian, S.; Lin, Z.; Li, X.; Zhang, X.; Zhang, K.; Xiao, K.; Geohegan, D.; Wallace, R.; Chen, L.-Q.; et al. A Roadmap for Electronic Grade 2D Materials. *2D Mater.* **2019**, *6*, No. 022001.

(347) Chakraborty, C.; Vamivakas, N.; Englund, D. Advances in Quantum Light Emission From 2D Materials. *Nanophotonics* **2019**, *8*, 2017–2032.

(348) Schuler, B.; Qiu, D. Y.; Refaely-Abramson, S.; Kastl, C.; Chen, C. T.; Barja, S.; Koch, R. J.; Ogletree, D. F.; Aloni, S.; Schwartzberg, A. M.; et al. Large Spin-Orbit Splitting of Deep In-Gap Defect States of Engineered Sulfur Vacancies in Monolayer WS 2. *Phys. Rev. Lett.* **2019**, *123*, No. 076801.

(349) Tian, X.; Kim, D. S.; Yang, S.; Ciccarino, C. J.; Gong, Y.; Yang, Y.; Yang, Y.; Duschatko, B.; Yuan, Y.; Ajayan, P. M.; et al. Correlating the Three-Dimensional Atomic Defects and Electronic Properties of Two-Dimensional Transition Metal Dichalcogenides. *Nat. Mater.* **2020**, *19*, 867–873.

(350) Chakraborty, C.; Chakraborty, C.; Mukherjee, A.; Moon, H.; Konthasinghe, K.; Qiu, L.; Hou, W.; Peña, T.; Watson, C.; Wu, S. M.; et al. Strain Tuning of the Emission Axis of Quantum Emitters in an Atomically Thin Semiconductor. *Optica* **2020**, *7*, 580–585.

(351) Qian, Q.; Peng, L.; Perea-Lopez, N.; Fujisawa, K.; Zhang, K.; Zhang, X.; Choudhury, T. H.; Redwing, J. M.; Terrones, M.; Ma, X.; et al. Defect Creation in WSe2 with a Microsecond Photoluminescence Lifetime by Focused Ion Beam Irradiation. *Nanoscale* **2020**, *12*, 2047–2056.

(352) Cochrane, K. A.; Lee, J.-H.; Kastl, C.; Haber, J. B.; Zhang, T.; Kozhakhmetov, A.; Robinson, J. A.; Terrones, M.; Repp, J.; Neaton, J. B. et al. Vibronic Response of a Spin-1/2 State From a Carbon Impurity in Two-Dimensional WS₂. 2020; arXiv:2008.12196 [cond-mat]; http://arxiv.org/abs/2008.12196.

(353) Maze, J. R.; Gali, A.; Togan, E.; Chu, Y.; Trifonov, A.; Kaxiras, E.; Lukin, M. D. Properties of Nitrogen-Vacancy Centers in Diamond: the Group Theoretic Approach. *New J. Phys.* **2011**, *13*, No. 025025.

(354) Doherty, M. W.; Manson, N. B.; Delaney, P.; Jelezko, F.; Wrachtrup, J.; Hollenberg, L. C. The Nitrogen-Vacancy Colour Centre in Diamond. *Phys. Rep.* **2013**, *528*, 1–45.

(355) Ivády, V.; Zheng, H.; Wickenbrock, A.; Bougas, L.; Chatzidrosos, G.; Nakamura, K.; Sumiya, H.; Ohshima, T.; Isoya, J.; Budker, D. et al. Photoluminescence at the Ground State Level Anticrossing of the Nitrogen-Vacancy Center in Diamond. 2020; arXiv:2006.05085 [cond-mat.mes-hall]; https://arxiv.org/abs/2006. 05085.

(356) Ashfold, M. N. R.; Goss, J. P.; Green, B. L.; May, P. W.; Newton, M. E.; Peaker, C. V. Nitrogen in Diamond. *Chem. Rev.* **2020**, *120*, 5745–5794.

(357) Hepp, C.; Müller, T.; Waselowski, V.; Becker, J. N.; Pingault, B.; Sternschulte, H.; Steinmüller-Nethl, D.; Gali, A.; Maze, J. R.; Atatüre, M.; et al. Electronic Structure of the Silicon Vacancy Color Center in Diamond. *Phys. Rev. Lett.* **2014**, *112*, No. 036405.

(358) Green, B. L.; Doherty, M. W.; Nako, E.; Manson, N. B.; D'Haenens-Johansson, U. F. S.; Williams, S. D.; Twitchen, D. J.; Newton, M. E. Electronic Structure of the Neutral Silicon-Vacancy Center in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2019**, *99*, 161112.

(359) Tran, T. T.; Bray, K.; Ford, M. J.; Toth, M.; Aharonovich, I. Quantum Emission from Hexagonal Boron Nitride Monolayers. *Nat. Nanotechnol.* **2016**, *11*, 37–41.

(360) Grosso, G.; Moon, H.; Lienhard, B.; Ali, S.; Efetov, D. K.; Furchi, M. M.; Jarillo-Herrero, P.; Ford, M. J.; Aharonovich, I.; Englund, D. Tunable and High-Purity Room Temperature Single-Photon Emission from Atomic Defects in Hexagonal Boron Nitride. *Nat. Commun.* **2017**, *8*, 705.

(361) Sajid, A.; Ford, M. J.; Reimers, J. R. Single-Photon Emitters in Hexagonal Boron Nitride: A Review of Progress. *Rep. Prog. Phys.* **2020**, 83, No. 044501.

(362) Grosso, G.; Moon, H.; Ciccarino, C. J.; Flick, J.; Mendelson, N.; Mennel, L.; Toth, M.; Aharonovich, I.; Narang, P.; Englund, D. R. Low-Temperature ElectronPhonon Interaction of Quantum Emitters in Hexagonal Boron Nitride. *ACS Photonics* **2020**, *7*, 1410–1417.

(363) Tawfik, S. A.; Ali, S.; Fronzi, M.; Kianinia, M.; Tran, T. T.; Stampfl, C.; Aharonovich, I.; Toth, M.; Ford, M. J. First-Principles Investigation of Quantum Emission from hBN Defects. *Nanoscale* **2017**, *9*, 13575–13582.

(364) Sajid, A.; Reimers, J. R.; Ford, M. J. Defect States in Hexagonal Boron Nitride: Assignments of Observed Properties and Prediction of Properties Relevant to Quantum Computation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *97*, No. 064101.

(365) Kianinia, M.; Bradac, C.; Sontheimer, B.; Wang, F.; Tran, T. T.; Nguyen, M.; Kim, S.; Xu, Z.-Q.; Jin, D.; Schell, A. W.; et al. All-Optical Control and Super-Resolution Imaging of Quantum Emitters in Layered Materials. *Nat. Commun.* **2018**, *9*, 874.

(366) Jungwirth, N. R.; Fuchs, G. D. Optical Absorption and Emission Mechanisms of Single Defects in Hexagonal Boron Nitride. *Phys. Rev. Lett.* **2017**, *119*, No. 057401.

(367) Turiansky, M. E.; Alkauskas, A.; Bassett, L. C.; Van de Walle, C. G. Dangling Bonds in Hexagonal Boron Nitride as Single-Photon Emitters. *Phys. Rev. Lett.* **2019**, *123*, 127401.

(368) Gottscholl, A.; Kianinia, M.; Soltamov, V.; Orlinskii, S.; Mamin, G.; Bradac, C.; Kasper, C.; Krambrock, K.; Sperlich, A.; Toth, M.; et al. Initialization and Read-Out of Intrinsic Spin Defects in a van der Waals Crystal at Room Temperature. *Nat. Mater.* **2020**, *19*, 540–545.

(369) Abdi, M.; Chou, J.-P.; Gali, A.; Plenio, M. B. Color Centers in Hexagonal Boron Nitride Monolayers: A Group Theory and Ab Initio Analysis. *ACS Photonics* **2018**, *5*, 1967–1976.

(370) Tran, T. T.; Elbadawi, C.; Totonjian, D.; Lobo, C. J.; Grosso, G.; Moon, H.; Englund, D. R.; Ford, M. J.; Aharonovich, I.; Toth, M. Robust Multicolor Single Photon Emission from Point Defects in Hexagonal Boron Nitride. *ACS Nano* **2016**, *10*, 7331–7338.

(371) Exarhos, A. L.; Hopper, D. A.; Patel, R. N.; Doherty, M. W.; Bassett, L. C. Magnetic-Field-Dependent Quantum Emission in Hexagonal Boron Nitride at Room Temperature. *Nat. Commun.* **2019**, *10*, 222.

(372) Dietrich, A.; Bürk, M.; Steiger, E. S.; Antoniuk, L.; Tran, T. T.; Nguyen, M.; Aharonovich, I.; Jelezko, F.; Kubanek, A. Observation of Fourier transform limited lines in hexagonal boron nitride. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *98*, No. 081414.

(373) Kianinia, M.; White, S.; Fröch, J. E.; Bradac, C.; Aharonovich, I. Generation of Spin Defects in Hexagonal Boron Nitride. *ACS Photonics* **2020**, *7*, 2147–2152.

(374) Tran, T. T.; Kianinia, M.; Nguyen, M.; Kim, S.; Xu, Z.-Q.; Kubanek, A.; Toth, M.; Aharonovich, I. Resonant Excitation of Quantum Emitters in Hexagonal Boron Nitride. *ACS Photonics* **2018**, *5*, 295–300.

(375) Wang, Q.; Zhao, Q.; Zhao, X.; Luo, X.; Wong, C. P. Y.; Wang, J.; Wan, D.; Venkatesan, T.; Pennycook, S. J.; Loh, K. P.; et al.

Photoluminescence Upconversion by Defects in Hexagonal Boron Nitride. *Nano Lett.* **2018**, *18*, 6898–6905.

(376) Wigger, D.; Schmidt, R.; Del Pozo-Zamudio, O.; Preuß, J. A.; Tonndorf, P.; Schneider, R.; Steeger, P.; Kern, J.; Khodaei, Y.; Sperling, J.; et al. Phonon-Assisted Emission and Absorption of Individual Color Centers in Hexagonal Boron Nitride. *2D Mater.* **2019**, *6*, No. 035006.

(377) Exarhos, A. L.; Hopper, D. A.; Grote, R. R.; Alkauskas, A.; Bassett, L. C. Optical Signatures of Quantum Emitters in Suspended Hexagonal Boron Nitride. *ACS Nano* **2017**, *11*, 3328–3336.

(378) Noh, G.; Choi, D.; Kim, J.-H.; Im, D.-G.; Kim, Y.-H.; Seo, H.; Lee, J. Stark Tuning of Single-Photon Emitters in Hexagonal Boron Nitride. *Nano Lett.* **2018**, *18*, 4710–4715.

(379) Weston, L.; Wickramaratne, D.; Mackoit, M.; Alkauskas, A.; Van de Walle, C. G. Native Point Defects and Impurities in Hexagonal Boron Nitride. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *97*, 214104.

(380) Jungwirth, N. R.; Calderon, B.; Ji, Y.; Spencer, M. G.; Flatté, M. E.; Fuchs, G. D. Temperature Dependence of Wavelength Selectable Zero-Phonon Emission from Single Defects in Hexagonal Boron Nitride. *Nano Lett.* **2016**, *16*, 6052–6057.

(381) Janotti, A.; Van de Walle, C. G. Native Point Defects in ZnO. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2007**, *76*, 165202.

(382) Seo, H.; Ping, Y.; Galli, G. Role of Point Defects in Enhancing the Conductivity of BiVO4. *Chem. Mater.* **2018**, *30*, 7793–7802.

(383) Siyushev, P.; Xia, K.; Reuter, R.; Jamali, M.; Zhao, N.; Yang, N.; Duan, C.; Kukharchyk, N.; Wieck, A. D.; Kolesov, R.; et al. Coherent Properties of Single Rare-Earth Spin Qubits. *Nat. Commun.* **2014**, *5*, 3895.

(384) Hanson, R.; Awschalom, D. D. Coherent Manipulation of Single Spins in Semiconductors. *Nature* **2008**, *453*, 1043–1049.

(385) Atatüre, M.; Englund, D.; Vamivakas, N.; Lee, S.-Y.; Wrachtrup, J. Material Platforms for Spin-Based Photonic Quantum Technologies. *Nat. Rev. Mater.* **2018**, *3*, 38–51.

(386) Awschalom, D. D.; Hanson, R.; Wrachtrup, J.; Zhou, B. B. Quantum Technologies with Optically Interfaced Solid-State Spins. *Nat. Photonics* **2018**, *12*, 516–527.

(387) Narang, P.; Ciccarino, C. J.; Flick, J.; Englund, D. Quantum Materials with Atomic Precision: Artificial Atoms in Solids: Ab Initio Design, Control, and Integration of Single Photon Emitters in Artificial Quantum Materials. *Adv. Funct. Mater.* **2019**, *29*, 1904557.

(388) Alkauskas, A.; Bassett, L. C.; Exarhos, A. L.; Fu, K.-M. C. Defects by Design: Quantum Nanophotonics in Emerging Materials. *Nanophotonics* **2019**, *8*, 1863–1865.

(389) Hsieh, S.; Bhattacharyya, P.; Zu, C.; Mittiga, T.; Smart, T. J.; Machado, F.; Kobrin, B.; Höhn, T. O.; Rui, N. Z.; Kamrani, M.; et al. Imaging Stress and Magnetism at High Pressures using a Nanoscale Quantum Sensor. *Science* **2019**, *366*, 1349–1354.

(390) Weber, J. R.; Koehl, W. F.; Varley, J. B.; Janotti, A.; Buckley, B. B.; Van de Walle, C. G.; Awschalom, D. D. Quantum Computing with Defects. *Proc. Natl. Acad. Sci. U. S. A.* **2010**, *107*, 8513–8518.

(391) Bradley, C. E.; Randall, J.; Abobeih, M. H.; Berrevoets, R. C.; Degen, M. J.; Bakker, M. A.; Markham, M.; Twitchen, D. J.; Taminiau, T. H. A Ten-Qubit Solid-State Spin Register with Quantum Memory up to One Minute. *Phys. Rev. X* **2019**, *9*, No. 031045.

(392) Nguyen, C. T.; Sukachev, D. D.; Bhaskar, M. K.; Machielse, B.; Levonian, D. S.; Knall, E. N.; Stroganov, P.; Riedinger, R.; Park, H.; Lončar, M.; et al. Quantum Network Nodes Based on Diamond Qubits with an Efficient Nanophotonic Interface. *Phys. Rev. Lett.* **2019**, *123*, 183602.

(393) Jelezko, F.; Gaebel, T.; Popa, I.; Gruber, A.; Wrachtrup, J. Observation of Coherent Oscillations in a Single Electron Spin. *Phys. Rev. Lett.* **2004**, *92*, No. 076401.

(394) Tamarat, P.; Gaebel, T.; Rabeau, J. R.; Khan, M.; Greentree, A. D.; Wilson, H.; Hollenberg, L. C. L.; Prawer, S.; Hemmer, P.; Jelezko, F.; et al. Stark Shift Control of Single Optical Centers in Diamond. *Phys. Rev. Lett.* **2006**, *97*, No. 083002.

(395) Childress, L.; Gurudev Dutt, M. V.; Taylor, J. M.; Zibrov, A. S.; Jelezko, F.; Wrachtrup, J.; Hemmer, P. R.; Lukin, M. D. Coherent Dynamics of Coupled Electron and Nuclear Spin Qubits in Diamond. *Science* **2006**, *314*, 281–285.

(396) Dutt, M. V. G.; Childress, L.; Jiang, L.; Togan, E.; Maze, J.; Jelezko, F.; Zibrov, A. S.; Hemmer, P. R.; Lukin, M. D. Quantum Register Based on Individual Electronic and Nuclear Spin Qubits in Diamond. *Science* **2007**, *316*, 1312–1316.

(397) Fuchs, G. D.; Dobrovitski, V. V.; Hanson, R.; Batra, A.; Weis, C. D.; Schenkel, T.; Awschalom, D. D. Excited-State Spectroscopy Using Single Spin Manipulation in Diamond. *Phys. Rev. Lett.* **2008**, *101*, 117601.

(398) Maze, J. R.; Stanwix, P. L.; Hodges, J. S.; Hong, S.; Taylor, J. M.; Cappellaro, P.; Jiang, L.; Dutt, M. V. G.; Togan, E.; Zibrov, A. S.; et al. Nanoscale Magnetic Sensing with an Individual Electronic Spin in Diamond. *Nature* **2008**, *455*, 644–647.

(399) Buckley, B. B.; Fuchs, G. D.; Bassett, L. C.; Awschalom, D. D. Spin-Light Coherence for Single-Spin Measurement and Control in Diamond. *Science* **2010**, *330*, 1212–1215.

(400) Fuchs, G. D.; Dobrovitski, V. V.; Toyli, D. M.; Heremans, F. J.; Weis, C. D.; Schenkel, T.; Awschalom, D. D. Excited-State Spin Coherence of a Single NitrogenVacancy Centre in Diamond. *Nat. Phys.* **2010**, *6*, 668–672.

(401) Fuchs, G. D.; Burkard, G.; Klimov, P. V.; Awschalom, D. D. A Quantum Memory Intrinsic to Single NitrogenVacancy Centres in Diamond. *Nat. Phys.* **2011**, *7*, 789–793.

(402) Bassett, L. C.; Heremans, F. J.; Yale, C. G.; Buckley, B. B.; Awschalom, D. D. Electrical Tuning of Single Nitrogen-Vacancy Center Optical Transitions Enhanced by Photoinduced Fields. *Phys. Rev. Lett.* **2011**, *107*, 266403.

(403) Dolde, F.; Fedder, H.; Doherty, M. W.; Nöbauer, T.; Rempp, F.; Balasubramanian, G.; Wolf, T.; Reinhard, F.; Hollenberg, L. C. L.; Jelezko, F.; et al. Electric-Field Sensing Using Single Diamond Spins. *Nat. Phys.* **2011**, *7*, 459–463.

(404) Aharonovich, I.; Greentree, A. D.; Prawer, S. Diamond Photonics. *Nat. Photonics* **2011**, *5*, 397–405.

(405) Robledo, L.; Childress, L.; Bernien, H.; Hensen, B.; Alkemade, P. F. A.; Hanson, R. High-fidelity Projective Read-Out of a Solid-State Spin Quantum Register. *Nature* **2011**, 477, 574–578.

(406) Edmonds, A. M.; D'Haenens-Johansson, U. F. S.; Cruddace, R. J.; Newton, M. E.; Fu, K.-M. C.; Santori, C.; Beausoleil, R. G.; Twitchen, D. J.; Markham, M. L. Production of Oriented Nitrogen-Vacancy Color Centers in Synthetic Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2012**, *86*, No. 035201.

(407) MacQuarrie, E. R.; Gosavi, T. A.; Jungwirth, N. R.; Bhave, S. A.; Fuchs, G. D. Mechanical Spin Control of Nitrogen-Vacancy Centers in Diamond. *Phys. Rev. Lett.* **2013**, *111*, 227602.

(408) Dobrovitski, V.; Fuchs, G.; Falk, A.; Santori, C.; Awschalom, D. Quantum Control over Single Spins in Diamond. *Annu. Rev. Condens. Matter Phys.* **2013**, *4*, 23–50.

(409) Childress, L.; Hanson, R. Diamond NV Centers for Quantum Computing and Quantum Networks. *MRS Bull.* **2013**, *38*, 134–138.

(410) Bernien, H.; Hensen, B.; Pfaff, W.; Koolstra, G.; Blok, M. S.; Robledo, L.; Taminiau, T. H.; Markham, M.; Twitchen, D. J.; Childress, L.; et al. Heralded Entanglement Between Solid-State Qubits Separated by Three Metres. *Nature* **2013**, *497*, 86–90.

(411) Schirhagl, R.; Chang, K.; Loretz, M.; Degen, C. L. Nitrogen-Vacancy Centers in Diamond: Nanoscale Sensors for Physics and Biology. *Annu. Rev. Phys. Chem.* **2014**, *65*, 83–105.

(412) Taminiau, T. H.; Cramer, J.; van der Sar, T.; Dobrovitski, V. V.; Hanson, R. Universal Control and Error Correction in Multi-Qubit Spin Registers in Diamond. *Nat. Nanotechnol.* **2014**, *9*, 171–176.

(413) MacQuarrie, E. R.; Gosavi, T. A.; Bhave, S. A.; Fuchs, G. D. Continuous Dynamical Decoupling of a Single Diamond Nitrogen-Vacancy Center Spin with a Mechanical Resonator. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, *92*, 224419.

(414) Aharonovich, I.; Englund, D.; Toth, M. Solid-State Single-Photon Emitters. *Nat. Photonics* **2016**, *10*, 631–641.

(415) Liu, G.-Q.; Pan, X.-Y. Quantum Information Processing with Nitrogen–Vacancy Centers in Diamond. *Chin. Phys. B* 2018, 27, No. 020304.

pubs.acs.org/CR

(416) Humphreys, P. C.; Kalb, N.; Morits, J. P. J.; Schouten, R. N.; Vermeulen, R. F. L.; Twitchen, D. J.; Markham, M.; Hanson, R. Deterministic Delivery of Remote Entanglement on a Quantum Network. *Nature* **2018**, 558, 268–273.

(417) Rozpedek, F.; Yehia, R.; Goodenough, K.; Ruf, M.; Humphreys, P. C.; Hanson, R.; Wehner, S.; Elkouss, D. Near-Term Quantum-Repeater Experiments with Nitrogen-Vacancy Centers: Overcoming the Limitations of Direct Transmission. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 052330.

(418) Gali, A. Ab Initio Theory of the Nitrogen-Vacancy Center in Diamond. *Nanophotonics* **2019**, *8*, 1907–1943.

(419) Goldman, M. L.; Patti, T. L.; Levonian, D.; Yelin, S. F.; Lukin, M. D. Optical Control of a Single Nuclear Spin in the Solid State. *Phys. Rev. Lett.* **2020**, *124*, 153203.

(420) Abtew, T. A.; Sun, Y. Y.; Shih, B.-C.; Dev, P.; Zhang, S. B.; Zhang, P. Dynamic Jahn-Teller Effect in the NV⁻ Center in Diamond. *Phys. Rev. Lett.* **2011**, *107*, 146403.

(421) Thiering, G. m. H.; Gali, A. Ab Initio Calculation of Spin-Orbit Coupling for an NV Center in Diamond Exhibiting Dynamic Jahn-Teller Effect. *Phys. Rev. B: Condens. Matter Mater. Phys.* 2017, 96, No. 081115.

(422) Sipahigil, A.; Jahnke, K. D.; Rogers, L. J.; Teraji, T.; Isoya, J.; Zibrov, A. S.; Jelezko, F.; Lukin, M. D. Indistinguishable Photons from Separated Silicon-Vacancy Centers in Diamond. *Phys. Rev. Lett.* **2014**, *113*, 113602.

(423) Neu, E.; Hepp, C.; Hauschild, M.; Gsell, S.; Fischer, M.; Sternschulte, H.; Steinmüller-Nethl, D.; Schreck, M.; Becher, C. Low-Temperature Investigations of Single Silicon Vacancy Colour Centres in Diamond. *New J. Phys.* **2013**, *15*, 043005.

(424) Iwasaki, T.; Mizuochi, N.; Ishibashi, F.; Hatano, M.; Doi, Y.; Miyamoto, Y.; Rogers, L. J.; Kobayashi, S.; Yamasaki, S.; Jelezko, F.; et al. Germanium-Vacancy Single Color Centers in Diamond. *Sci. Rep.* **2015**, *5*, 12882.

(425) Siyushev, P.; Metsch, M. H.; Iwasaki, T.; Miyamoto, Y.; Jelezko, F.; Taniguchi, T.; Hatano, M. Tin-Vacancy Quantum Emitters in Diamond. *Phys. Rev. Lett.* **2017**, *119*, 253601.

(426) Bhaskar, M. K.; Sukachev, D. D.; Sipahigil, A.; Evans, R. E.; Burek, M. J.; Nguyen, C. T.; Rogers, L. J.; Siyushev, P.; Metsch, M. H.; Park, H.; et al. Quantum Nonlinear Optics with a Germanium-Vacancy Color Center in a Nanoscale Diamond Waveguide. *Phys. Rev. Lett.* **2017**, *118*, 223603.

(427) Sukachev, D. D.; Sipahigil, A.; Nguyen, C. T.; Bhaskar, M. K.; Evans, R. E.; Jelezko, F.; Lukin, M. D. Silicon-Vacancy Spin Qubit in Diamond: A Quantum Memory Exceeding 10 ms with Single-Shot State Readout. *Phys. Rev. Lett.* **2017**, *119*, 223602.

(428) Evans, R. E.; Bhaskar, M. K.; Sukachev, D. D.; Nguyen, C. T.; Sipahigil, A.; Burek, M. J.; Machielse, B.; Zhang, G. H.; Zibrov, A. S.; Bielejec, E.; et al. Photon-Mediated Interactions between Quantum Emitters in a Diamond Nanocavity. *Science* **2018**, *362*, 662–665.

(429) Lemonde, M.-A.; Meesala, S.; Sipahigil, A.; Schuetz, M. J. A.; Lukin, M. D.; Lončar, M.; Rabl, P. Phonon Networks with Silicon-Vacancy Centers in Diamond Waveguides. *Phys. Rev. Lett.* **2018**, *120*, 213603.

(430) Trusheim, M. E.; Pingault, B.; Wan, N. H.; Gündoğan, M.; De Santis, L.; Debroux, R.; Gangloff, D.; Purser, C.; Chen, K. C.; Walsh, M.; et al. Transform-Limited Photons From a Coherent Tin-Vacancy Spin in Diamond. *Phys. Rev. Lett.* **2020**, *124*, No. 023602.

(431) Maity, S.; Shao, L.; Sohn, Y.-I.; Meesala, S.; Machielse, B.; Bielejec, E.; Markham, M.; Lončar, M. Spectral Alignment of Single-Photon Emitters in Diamond using Strain Gradient. *Phys. Rev. Appl.* **2018**, *10*, No. 024050.

(432) Maity, S.; Shao, L.; Bogdanović, S.; Meesala, S.; Sohn, Y.-I.; Sinclair, N.; Pingault, B.; Chalupnik, M.; Chia, C.; Zheng, L.; et al. Coherent Acoustic Control of a Single Silicon Vacancy Spin in Diamond. *Nat. Commun.* **2020**, *11*, 193.

(433) Wan, N. H.; Lu, T.-J.; Chen, K. C.; Walsh, M. P.; Trusheim, M. E.; De Santis, L.; Bersin, E. A.; Harris, I. B.; Mouradian, S. L.; Christen, I. R.; et al. Large-Scale Integration of Artificial Atoms in Hybrid Photonic Circuits. *Nature* **2020**, *583*, 226–231.

(434) Machielse, B.; Bogdanovic, S.; Meesala, S.; Gauthier, S.; Burek, M. J.; Joe, G.; Chalupnik, M.; Sohn, Y. I.; Holzgrafe, J.; Evans, R. E.; et al. Quantum Interference of Electromechanically Stabilized Emitters in Nanophotonic Devices. *Phys. Rev. X* **2019**, *9*, No. 031022.

(435) Sohn, Y.-I.; Meesala, S.; Pingault, B.; Atikian, H. A.; Holzgrafe, J.; Gündoğan, M.; Stavrakas, C.; Stanley, M. J.; Sipahigil, A.; Choi, J.; et al. Controlling the Coherence of a Diamond Spin Qubit Through its Strain Environment. *Nat. Commun.* **2018**, *9*, 2012.

(436) Meesala, S.; Sohn, Y.-I.; Pingault, B.; Shao, L.; Atikian, H. A.; Holzgrafe, J.; Gündoğan, M.; Stavrakas, C.; Sipahigil, A.; Chia, C.; et al. Strain Engineering of the Silicon-Vacancy Center in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *97*, 205444.

(437) Nguyen, C. T.; Sukachev, D. D.; Bhaskar, M. K.; Machielse, B.; Levonian, D. S.; Knall, E. N.; Stroganov, P.; Chia, C.; Burek, M. J.; Riedinger, R.; et al. An Integrated Nanophotonic Quantum Register Based on Silicon-Vacancy Spins in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2019**, *100*, 165428.

(438) Bradac, C.; Gao, W.; Forneris, J.; Trusheim, M. E.; Aharonovich, I. Quantum Nanophotonics with Group IV Defects in Diamond. *Nat. Commun.* **2019**, *10*, 5625.

(439) Thiering, G. m. H.; Gali, A. Ab Initio Magneto-Optical Spectrum of Group-IV Vacancy Color Centers in Diamond. *Phys. Rev.* X **2018**, *8*, No. 021063.

(440) Gali, A.; Maze, J. R. Ab Initio Study of the Split Silicon-Vacancy Defect in Diamond: Electronic Structure and Related Properties. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2013**, *88*, 235205.

(441) Thiering, G.; Gali, A. The $(e_g \otimes e_u) \otimes E_g$ product Jahn-Teller Effect in the Neutral Group-IV Vacancy Quantum Bits in Diamond. *NPJ. Comput. Mater.* **2019**, *5*, 18.

(442) D^THaenens-Johansson, U. F. S.; Edmonds, A. M.; Green, B. L.; Newton, M. E.; Davies, G.; Martineau, P. M.; Khan, R. U. A.; Twitchen, D. J. Optical Properties of the Neutral Silicon Split-Vacancy Center in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2011**, *84*, 245208.

(443) Green, B. L.; Mottishaw, S.; Breeze, B. G.; Edmonds, A. M.; D'Haenens-Johansson, U. F. S.; Doherty, M. W.; Williams, S. D.; Twitchen, D. J.; Newton, M. E. Neutral Silicon-Vacancy Center in Diamond: Spin Polarization and Lifetimes. *Phys. Rev. Lett.* **2017**, *119*, No. 096402.

(444) Rose, B. C.; Huang, D.; Zhang, Z.-H.; Stevenson, P.; Tyryshkin, A. M.; Sangtawesin, S.; Srinivasan, S.; Loudin, L.; Markham, M. L.; Edmonds, A. M.; et al. Observation of an Environmentally Insensitive Solid-State Spin Defect in Diamond. *Science* **2018**, *361*, 60–63.

(445) Zhang, Z.-H.; Stevenson, P.; Thiering, G.; Rose, B. C.; Huang, D.; Edmonds, A. M.; Markham, M. L.; Lyon, S. A.; Gali, A.; de Leon, N. P. Optically Detected Magnetic Resonance in the Neutral Silicon Vacancy Center in Diamond via Bound Exciton States. 2020; arXiv:2004.12544 [cond-mat.mes-hall]; https://arxiv.org/abs/2004. 12544.

(446) Alkauskas, A.; Lyons, J. L.; Steiauf, D.; Van de Walle, C. G. First-Principles Calculations of Luminescence Spectrum Line Shapes for Defects in Semiconductors: The Example of GaN and ZnO. *Phys. Rev. Lett.* **2012**, *109*, 267401.

(447) Alkauskas, A.; Buckley, B. B.; Awschalom, D. D.; Van de Walle, C. G. V. First-Principles Theory of the Luminescence Lineshape for the Triplet Transition in Diamond NV Centres. *New J. Phys.* **2014**, *16*, No. 073026.

(448) Freysoldt, C.; Grabowski, B.; Hickel, T.; Neugebauer, J.; Kresse, G.; Janotti, A.; Van de Walle, C. G. First-principles Calculations for Point Defects in Solids. *Rev. Mod. Phys.* **2014**, *86*, 253–305.

(449) Alkauskas, A.; McCluskey, M. D.; Van de Walle, C. G. Tutorial: Defects in Semiconductors—Combining Experiment and Theory. *J. Appl. Phys.* **2016**, *119*, 181101.

(450) Dreyer, C. E.; Alkauskas, A.; Lyons, J. L.; Janotti, A.; Van de Walle, C. G. First-Principles Calculations of Point Defects for Quantum Technologies. *Annu. Rev. Mater. Res.* **2018**, *48*, 1–26.

(451) Bersuker, I.; Polinger, V. Vibronic Interactions in Molecules and Crystals; Springer Series in Chemical Physics, Springer-Verlag: Berlin, Heidelberg, 1990; Vol. 49.

(452) Fu, K.-M. C.; Santori, C.; Barclay, P. E.; Rogers, L. J.; Manson, N. B.; Beausoleil, R. G. Observation of the Dynamic Jahn-Teller Effect in the Excited States of Nitrogen-Vacancy Centers in Diamond. *Phys. Rev. Lett.* **2009**, *103*, 256404.

(453) Ulbricht, R.; Dong, S.; Chang, I.-Y.; Mariserla, B. M. K.; Dani, K. M.; Hyeon-Deuk, K.; Loh, Z.-H. Jahn-Teller-Induced Femtosecond Electronic Depolarization Dynamics of the Nitrogen-Vacancy Defect in Diamond. *Nat. Commun.* **2016**, *7*, 13510.

(454) Jahn, H. A.; Teller, E. Stability of Polyatomic Molecules in Degenerate Electronic States - I - Orbital Degeneracy. *Proc. R. Soc. A* **1937**, *161*, 220–235.

(455) Ham, F. S. Dynamical Jahn-Teller Effect in Paramagnetic Resonance Spectra: Orbital Reduction Factors and Partial Quenching of Spin-Orbit Interaction. *Phys. Rev.* **1965**, *138*, A1727–A1740.

(456) Norambuena, A.; Jimenez, A.; Becher, C.; Maze, J. R. Effect of Phonons on the Electron Spin Resonance Absorption Spectrum. *New J. Phys.* **2020**, *22*, No. 073068.

(457) Peierls, R. E. *Quantum Theory of Solids*; Oxford classic texts in the physical sciences; Clarendon Press: Oxford, 1955.

(458) Kohn, W. Image of the Fermi Surface in the Vibration Spectrum of a Metal. *Phys. Rev. Lett.* **1959**, *2*, 393–394.

(459) Grüner, G. The Dynamics of Charge-Density Waves. *Rev. Mod. Phys.* **1988**, *60*, 1129–1181.

(460) Gruner, G. Density Waves in Solids; Frontiers in physics; CRC Press, 1994; Vol. 89.

(461) Johannes, M. D.; Mazin, I. I. Fermi Surface Nesting and the Origin of Charge Density Waves in Metals. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2008**, 77, 165135.

(462) Zhu, X.; Cao, Y.; Zhang, J.; Plummer, E. W.; Guo, J. Classification of Charge Density Waves Based on Their Nature. *Proc. Natl. Acad. Sci. U. S. A.* **2015**, *112*, 2367–2371.

(463) Bardeen, J.; Cooper, L. N.; Schrieffer, J. R. Microscopic Theory of Superconductivity. *Phys. Rev.* **1957**, *106*, 162–164.

(464) Bardeen, J.; Cooper, L. N.; Schrieffer, J. R. Theory of Superconductivity. *Phys. Rev.* **1957**, *108*, 1175–1204.

(465) McMillan, W. L. Transition Temperature of Strong-Coupled Superconductors. *Phys. Rev.* **1968**, *167*, 331–344.

(466) Allen, P. B.; Dynes, R. C. Transition Temperature of Strong-Coupled Superconductors Reanalyzed. *Phys. Rev. B* **1975**, *12*, 905–922.

(467) Marsiglio, F.; Carbotte, J. P. In *Superconductivity: Conventional and Unconventional Superconductors*; Bennemann, K. H., Ketterson, J. B., Eds.; Springer: Berlin, Heidelberg, 2008; pp 73–162.

(468) Giustino, F. Electron-Phonon Interactions from First Principles. *Rev. Mod. Phys.* **2017**, *89*, No. 015003.

(469) Zhang, C.; Liu, Z.; Chen, Z.; Xie, Y.; He, R.; Tang, S.; He, J.; Li, W.; Jia, T.; Rebec, S. N.; et al. Ubiquitous Strong Electron–Phonon Coupling at the Interface of FeSe/SrTiO3. *Nat. Commun.* **2017**, *8*, 14468.

(470) Song, Q.; Yu, T. L.; Lou, X.; Xie, B. P.; Xu, H. C.; Wen, C. H. P.; Yao, Q.; Zhang, S. Y.; Zhu, X. T.; Guo, J. D.; et al. Evidence of Cooperative Effect on the Enhanced Superconducting Transition Temperature at the FeSe/SrTiO₃ Interface. *Nat. Commun.* **2019**, *10*, 758.

(471) Born, M.; Oppenheimer, R. Zur Quantentheorie der Molekeln. Ann. Phys. (Berlin, Ger.) **1927**, 389, 457–484.

(472) Abedi, A.; Maitra, N. T.; Gross, E. K. U. Correlated Electron-Nuclear Dynamics: Exact Factorization of the Molecular Wavefunction. *J. Chem. Phys.* **2012**, *137*, 22A530.

(473) Albareda, G.; Appel, H.; Franco, I.; Abedi, A.; Rubio, A. Correlated Electron-Nuclear Dynamics with Conditional Wave Functions. *Phys. Rev. Lett.* **2014**, *113*, No. 083003.

(474) Capitani, J. F.; Nalewajski, R. F.; Parr, R. G. Non-Born-Oppenheimer Density Functional Theory of Molecular Systems. *J. Chem. Phys.* **1982**, *76*, 568–573.

(475) Kreibich, T.; Gross, E. K. U. Multicomponent Density-Functional Theory for Electrons and Nuclei. *Phys. Rev. Lett.* **2001**, *86*, 2984–2987.

(476) Kreibich, T.; van Leeuwen, R.; Gross, E. K. U. Multicomponent Density-Functional Theory for Electrons and Nuclei. *Phys. Rev. A: At., Mol., Opt. Phys.* **2008**, *78*, No. 022501.

(477) Chakraborty, A.; Pak, M. V.; Hammes-Schiffer, S. Development of Electron-Proton Density Functionals for Multicomponent Density Functional Theory. *Phys. Rev. Lett.* **2008**, *101*, 153001.

(478) Sirjoosingh, A.; Hammes-Schiffer, S. Proton-Coupled Electron Transfer versus Hydrogen Atom Transfer: Generation of Charge-Localized Diabatic States. J. Phys. Chem. A 2011, 115, 2367–2377.

(479) Requist, R.; Gross, E. K. U. Exact Factorization-Based Density Functional Theory of Electrons and Nuclei. *Phys. Rev. Lett.* **2016**, *117*, 193001.

(480) Pak, M. V.; Chakraborty, A.; Hammes-Schiffer, S. Density Functional Theory Treatment of Electron Correlation in the Nuclear-Electronic Orbital Approach. J. Phys. Chem. A 2007, 111, 4522–4526.

(481) Pavosević, F.; Culpitt, T.; Hammes-Schiffer, S. Multicomponent Coupled Cluster Singles and Doubles Theory within the Nuclear-Electronic Orbital Framework. J. Chem. Theory Comput. 2019, 15, 338–347.

(482) Baroni, S.; de Gironcoli, S.; Dal Corso, A.; Giannozzi, P. Phonons and Related Crystal Properties from Density-Functional Perturbation Theory. *Rev. Mod. Phys.* **2001**, *73*, 515–562.

(483) Narang, P.; Zhao, L.; Claybrook, S.; Sundararaman, R. Effects of Interlayer Coupling on Hot-Carrier Dynamics in Graphene-Derived van der Waals Heterostructures. *Adv. Opt. Mater.* **201**7, *5*, 1600914.

(484) Ciccarino, C. J.; Christensen, T.; Sundararaman, R.; Narang, P. Dynamics and Spin-Valley Locking Effects in Monolayer Transition Metal Dichalcogenides. *Nano Lett.* **2018**, *18*, 5709–5715.

(485) Cococcioni, M.; de Gironcoli, S. Linear Response Approach to the Calculation of the Effective Interaction Parameters in the LDA + U Method. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2005**, *71*, No. 035105.

(486) Kulik, H. J.; Cococcioni, M.; Scherlis, D. A.; Marzari, N. Density Functional Theory in Transition-Metal Chemistry: A Self-Consistent Hubbard U Approach. *Phys. Rev. Lett.* **2006**, *97*, 103001.

(487) Kulik, H. J.; Marzari, N. Systematic Study of First-Row Transition-Metal Diatomic Molecules: A Self-Consistent DFT+U Approach. J. Chem. Phys. **2010**, 133, 114103.

(488) Himmetoglu, B.; Floris, A.; de Gironcoli, S.; Cococcioni, M. Hubbard-Corrected DFT Energy Functionals: The LDA+U Description of Correlated Systems. *Int. J. Quantum Chem.* **2014**, *114*, 14–49.

(489) Agapito, L. A.; Curtarolo, S.; Buongiorno Nardelli, M. Reformulation of DFT + U as a Pseudohybrid Hubbard Density Functional for Accelerated Materials Discovery. *Phys. Rev. X* **2015**, *5*, No. 011006.

(490) Gopal, P.; De Gennaro, R.; Gusmao, M. S. d. S.; Al Rahal Al Orabi, R.; Wang, H.; Curtarolo, S.; Fornari, M.; Buongiorno Nardelli, M. Improved Electronic Structure and Magnetic Exchange Interactions in Transition Metal Oxides. *J. Phys.: Condens. Matter* **2017**, *29*, 444003.

(491) Held, K. Electronic Structure Calculations using Dynamical Mean Field Theory. *Adv. Phys.* **200**7, *56*, 829–926.

(492) Hampel, A.; Beck, S.; Ederer, C. Effect of Charge Self-Consistency in DFT + DMFT Calculations for Complex Transition Metal Oxides. *Phys. Rev. Research* **2020**, *2*, No. 033088.

(493) Runge, E.; Gross, E. K. U. Density-Functional Theory for Time-Dependent Systems. *Phys. Rev. Lett.* **1984**, *52*, 997–1000.

(494) Ullrich, C. Time-Dependent Density-Functional Theory: Concepts and Applications; Oxford Graduate Texts; OUP Oxford, 2012.

(495) Yabana, K.; Bertsch, G. F. Time-Dependent Local-Density Approximation in Real Time. *Phys. Rev. B: Condens. Matter Mater. Phys.* **1996**, *54*, 4484–4487.

(496) Casida, M. E. Recent Advances In Density Functional Methods (Part I); World Scientific, 1995; pp 155–192.

(497) Kaur, J.; Ospadov, E.; Staroverov, V. N. What Is the Accuracy Limit of Adiabatic Linear-Response TDDFT Using Exact Exchange–Correlation Potentials and Approximate Kernels? *J. Chem. Theory Comput.* **2019**, *15*, 4956–4964.

(498) Casida, M.; Huix-Rotllant, M. Progress in Time-Dependent Density-Functional Theory. *Annu. Rev. Phys. Chem.* **2012**, *63*, 287–323.

(499) Gali, A. Time-Dependent Density Functional Study on the Excitation Spectrum of Point Defects in Semiconductors. *Phys. Status Solidi B* **2011**, *248*, 1337–1346.

(500) Reimers, J. R.; Sajid, A.; Kobayashi, R.; Ford, M. J. Understanding and Calibrating Density-Functional-Theory Calculations Describing the Energy and Spectroscopy of Defect Sites in Hexagonal Boron Nitride. *J. Chem. Theory Comput.* **2018**, *14*, 1602–1613.

(501) Tancogne-Dejean, N.; Oliveira, M. J. T.; Rubio, A. Self-Consistent DFT + U Method for Real-Space Time-Dependent Density Functional Theory Calculations. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2017**, *96*, 245133.

(502) Varas, A.; García-González, P.; Feist, J.; García-Vidal, F.; Rubio, A. Quantum Plasmonics: From Jellium Models to Ab Initio Calculations. *Nanophotonics* **2016**, *5*, 409–426.

(503) Narang, P.; Sundararaman, R.; Atwater, H. A. Plasmonic Hot Carrier Dynamics in Solid-State and Chemical Systems for Energy Conversion. *Nanophotonics* **2016**, *5*, 96–111.

(504) Sundararaman, R.; Narang, P.; Jermyn, A. S.; Goddard, W. A., III; Atwater, H. A. Theoretical Predictions for Hot-Carrier Generation from Surface Plasmon Decay. *Nat. Commun.* **2014**, *5*, 5788.

(505) Su, M.-N.; Ciccarino, C. J.; Kumar, S.; Dongare, P. D.; Hosseini Jebeli, S. A.; Renard, D.; Zhang, Y.; Ostovar, B.; Chang, W.-S.; Nordlander, P.; et al. Ultrafast Electron Dynamics in Single Aluminum Nanostructures. *Nano Lett.* **2019**, *19*, 3091–3097.

(506) Lozan, O.; Sundararaman, R.; Ea-Kim, B.; Rampnoux, J.-M.; Narang, P.; Dilhaire, S.; Lalanne, P. Increased Rise Time of Electron Temperature During Adiabatic Plasmon Focusing. *Nat. Commun.* **2017**, *8*, 1656.

(507) Shirodkar, S. N.; Mattheakis, M.; Cazeaux, P.; Narang, P.; Soljačić, M.; Kaxiras, E. Quantum Plasmons with Optical-Range Frequencies in Doped Few-Layer Graphene. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *97*, 195435.

(508) Brown, A. M.; Sundararaman, R.; Narang, P.; Schwartzberg, A. M.; Goddard, W. A.; Atwater, H. A. Experimental and Ab Initio Ultrafast Carrier Dynamics in Plasmonic Nanoparticles. *Phys. Rev. Lett.* **2017**, *118*, No. 087401.

(509) Jermyn, A. S.; Tagliabue, G.; Atwater, H. A.; Goddard, W. A.; Narang, P.; Sundararaman, R. Transport of Hot Carriers in Plasmonic Nanostructures. *Phys. Rev. Mater.* **2019**, *3*, No. 075201.

(510) Fuks, J. I.; Lacombe, L.; Nielsen, S. E. B.; Maitra, N. T. Exploring Non-Adiabatic Approximations to the Exchange-Correlation Functional of TDDFT. *Phys. Chem. Chem. Phys.* **2018**, *20*, 26145–26160.

(511) Fuks, J. I.; Elliott, P.; Rubio, A.; Maitra, N. T. Dynamics of Charge-Transfer Processes with Time-Dependent Density Functional Theory. *J. Phys. Chem. Lett.* **2013**, *4*, 735–739.

(512) Ruggenthaler, M.; van Leeuwen, R. Global Fixed-Point Proof of Time-Dependent Density-Functional Theory. *EPL* **2011**, *95*, 13001.

(513) Nielsen, S. E. B.; Ruggenthaler, M.; van Leeuwen, R. Numerical Construction of the Density-Potential Mapping. *Eur. Phys. J. B* 2018, *91*, 235.

(514) Perdew, J. P. Density Functional Theory and the Band Gap Problem. Int. J. Quantum Chem. **1985**, 28, 497–523.

(515) Perdew, J. P.; Yang, W.; Burke, K.; Yang, Z.; Gross, E. K. U.; Scheffler, M.; Scuseria, G. E.; Henderson, T. M.; Zhang, I. Y.; Ruzsinszky, A.; et al. Understanding Band Gaps of Solids in Generalized Kohn–Sham Theory. *Proc. Natl. Acad. Sci. U. S. A.* **2017**, *114*, 2801– 2806.

(516) Yang, Z.-h.; Ullrich, C. A. Direct Calculation of Exciton Binding Energies with Time-Dependent Density-Functional Theory. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2013**, *87*, 195204.

(517) Byun, Y.-M.; Sun, J.; Ullrich, C. A. Time-Dependent Density-Functional Theory for Periodic Solids: Assessment of Excitonic Exchange–Correlation Kernels. *Electron. Struct.* **2020**, *2*, No. 023002.

(518) Zhang, X.; Lu, G.; Baer, R.; Rabani, E.; Neuhauser, D. Linear-Response Time-Dependent Density Functional Theory with Stochastic Range-Separated Hybrids. *J. Chem. Theory Comput.* **2020**, *16*, 1064–1072.

(519) Hedin, L. New Method for Calculating the One-Particle Green's Function with Application to the Electron-Gas Problem. *Phys. Rev.* **1965**, *139*, A796–A823.

(520) Hybertsen, M. S.; Louie, S. G. Electron Correlation in Semiconductors and Insulators: Band Gaps and Quasiparticle Energies. *Phys. Rev. B: Condens. Matter Mater. Phys.* **1986**, *34*, 5390–5413.

(521) Onida, G.; Reining, L.; Rubio, A. Electronic Excitations: Density-Functional versus Many-Body Green's-Function Approaches. *Rev. Mod. Phys.* 2002, 74, 601–659.

(522) Kotani, T.; van Schilfgaarde, M.; Faleev, S. V. Quasiparticle Self-Consistent *GW* method: A Basis for the Independent-Particle Approximation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2007**, *76*, 165106.

(523) Qiu, D. Y.; da Jornada, F. H.; Louie, S. G. Optical Spectrum of MoS₂: Many-Body Effects and Diversity of Exciton States. *Phys. Rev. Lett.* **2013**, *111*, 216805.

(524) Qiu, D. Y.; Cao, T.; Louie, S. G. Nonanalyticity, Valley Quantum Phases, and Lightlike Exciton Dispersion in Monolayer Transition Metal Dichalcogenides: Theory and First-Principles Calculations. *Phys. Rev. Lett.* **2015**, *115*, 176801.

(525) Scherpelz, P.; Govoni, M.; Hamada, I.; Galli, G. Implementation and Validation of Fully Relativistic GW Calculations: Spin–Orbit Coupling in Molecules, Nanocrystals, and Solids. *J. Chem. Theory Comput.* **2016**, *12*, 3523–3544.

(526) Nguyen, N. L.; Ma, H.; Govoni, M.; Gygi, F.; Galli, G. Finite-Field Approach to Solving the Bethe-Salpeter Equation. *Phys. Rev. Lett.* **2019**, *122*, 237402.

(527) Ma, H.; Govoni, M.; Gygi, F.; Galli, G. A Finite-Field Approach for GW Calculations beyond the Random Phase Approximation. *J. Chem. Theory Comput.* **2019**, *15*, 154–164.

(528) Feynman, R. P. Space-Time Approach to Quantum Electrodynamics. *Phys. Rev.* **1949**, *76*, 769–789.

(529) Salpeter, E. E.; Bethe, H. A. A Relativistic Equation for Bound-State Problems. *Phys. Rev.* **1951**, *84*, 1232–1242.

(530) Rohlfing, M.; Louie, S. G. Electron-Hole Excitations and Optical Spectra from First Principles. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2000**, *62*, 4927–4944.

(531) Wu, F.; Galatas, A.; Sundararaman, R.; Rocca, D.; Ping, Y. First-Principles Engineering of Charged Defects for Two-Dimensional Quantum Technologies. *Phys. Rev. Mater.* **2017**, *1*, No. 071001.

(532) van Setten, M. J.; Caruso, F.; Sharifzadeh, S.; Ren, X.; Scheffler, M.; Liu, F.; Lischner, J.; Lin, L.; Deslippe, J. R.; Louie, S. G.; et al. GW100: Benchmarking G0W0 for Molecular Systems. *J. Chem. Theory Comput.* **2015**, *11*, 5665–5687.

(533) Govoni, M.; Galli, G. GW100: Comparison of Methods and Accuracy of Results Obtained with the WEST Code. *J. Chem. Theory Comput.* **2018**, *14*, 1895–1909.

(534) Mueller, T.; Malic, E. Exciton Physics and Device Application of Two-Dimensional Transition Metal Dichalcogenide Semiconductors. *NPJ. 2D Mater. Appl.* **2018**, *2*, 29.

(535) Refaely-Abramson, S.; Qiu, D. Y.; Louie, S. G.; Neaton, J. B. Defect-Induced Modification of Low-Lying Excitons and Valley Selectivity in Monolayer Transition Metal Dichalcogenides. *Phys. Rev. Lett.* **2018**, *121*, 167402.

(536) Robledo, L.; Bernien, H.; Sar, T. v. d.; Hanson, R. Spin Dynamics in the Optical Cycle of Single Nitrogen-Vacancy Centres in Diamond. *New J. Phys.* **2011**, *13*, No. 025013.

(537) Choi, S.; Jain, M.; Louie, S. G. Mechanism for Optical Initialization of Spin in NV⁻ Center in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2012**, *86*, No. 041202.

(538) Bockstedte, M.; Schütz, F.; Garratt, T.; Ivády, V.; Gali, A. Ab Initio Description of Highly Correlated States in Defects for Realizing Quantum Bits. *NPJ. Quantum Mater.* **2018**, *3*, 31.

(539) Tiago, M. L.; Ismail-Beigi, S.; Louie, S. G. Effect of Semicore Orbitals on the Electronic Band Gaps of Si, Ge, and GaAs within the GW Approximation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2004**, *69*, 125212. (540) Bruneval, F.; Gonze, X. Accurate *GW* Self-Energies in a Plane-Wave Basis using Only a Few Empty States: Towards Large Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2008**, 78, No. 085125.

(541) Giustino, F.; Cohen, M. L.; Louie, S. G. GW Method with the Self-Consistent Sternheimer Equation. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2010**, *81*, 115105.

(542) Govoni, M.; Galli, G. Large Scale GW Calculations. J. Chem. Theory Comput. 2015, 11, 2680–2696.

(543) Deslippe, J.; Samsonidze, G.; Strubbe, D. A.; Jain, M.; Cohen, M. L.; Louie, S. G. BerkeleyGW: A Massively Parallel Computer Package for the Calculation of the Quasiparticle and Optical Properties of Materials and Nanostructures. *Comput. Phys. Commun.* **2012**, *183*, 1269–1289.

(544) Neuhauser, D.; Gao, Y.; Arntsen, C.; Karshenas, C.; Rabani, E.; Baer, R. Breaking the Theoretical Scaling Limit for Predicting Quasiparticle Energies: The Stochastic *GW* Approach. *Phys. Rev. Lett.* **2014**, *113*, No. 076402.

(545) Vlček, V.; Rabani, E.; Neuhauser, D.; Baer, R. Stochastic GW Calculations for Molecules. *J. Chem. Theory Comput.* **2017**, *13*, 4997–5003.

(546) Vlček, V. c. v.; Li, W.; Baer, R.; Rabani, E.; Neuhauser, D. Swift GW Beyond 10,000 Electrons using Sparse Stochastic Compression. Phys. Rev. B: Condens. Matter Mater. Phys. **2018**, 98, No. 075107.

(547) Rabani, E.; Baer, R.; Neuhauser, D. Time-Dependent Stochastic Bethe-Salpeter Approach. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2015**, *91*, 235302.

(548) Neaton, J. B.; Hybertsen, M. S.; Louie, S. G. Renormalization of Molecular Electronic Levels at Metal-Molecule Interfaces. *Phys. Rev. Lett.* **2006**, *97*, 216405.

(549) Körbel, S.; Boulanger, P.; Duchemin, I.; Blase, X.; Marques, M. A. L.; Botti, S. Benchmark Many-Body GW and Bethe–Salpeter Calculations for Small Transition Metal Molecules. *J. Chem. Theory Comput.* **2014**, *10*, 3934–3943.

(550) van Setten, M. J.; Caruso, F.; Sharifzadeh, S.; Ren, X.; Scheffler, M.; Liu, F.; Lischner, J.; Lin, L.; Deslippe, J. R.; Louie, S. G.; et al. *GW* 100: Benchmarking G_0W_0 for Molecular Systems. *J. Chem. Theory Comput.* **2015**, *11*, 5665–5687.

(551) Rangel, T.; Hamed, S. M.; Bruneval, F.; Neaton, J. B. Evaluating the GW Approximation with CCSD(T) for Charged Excitations Across the Oligoacenes. *J. Chem. Theory Comput.* **2016**, *12*, 2834–2842.

(552) Blase, X.; Duchemin, I.; Jacquemin, D. The Bethe–Salpeter Equation in Chemistry: Relations with TD-DFT, Applications and Challenges. *Chem. Soc. Rev.* **2018**, *47*, 1022–1043.

(553) Liu, Z.-F.; da Jornada, F. H.; Louie, S. G.; Neaton, J. B. Accelerating GW-Based Energy Level Alignment Calculations for Molecule–Metal Interfaces Using a Substrate Screening Approach. J. Chem. Theory Comput. 2019, 15, 4218–4227.

(554) Liu, C.; Kloppenburg, J.; Yao, Y.; Ren, X.; Appel, H.; Kanai, Y.; Blum, V. All-Electron ab initio Bethe-Salpeter Equation Approach to Neutral Excitations in Molecules with Numeric Atom-Centered Orbitals. J. Chem. Phys. **2020**, 152, No. 044105.

(555) Philbin, J. P.; Rabani, E. Electron–Hole Correlations Govern Auger Recombination in Nanostructures. *Nano Lett.* **2018**, *18*, 7889– 7895.

(556) Lee, N.-E.; Zhou, J.-J.; Chen, H.-Y.; Bernardi, M. Ab initio Electron-Two-Phonon Scattering in GaAs from Next-to-Leading Order Perturbation Theory. *Nat. Commun.* **2020**, *11*, 1607.

(557) Li, Z.; Antonius, G.; Wu, M.; da Jornada, F. H.; Louie, S. G. Electron-Phonon Coupling from Ab Initio Linear-Response Theory within the *GW* Method: Correlation-Enhanced Interactions and Superconductivity in Ba_{1-x}K_xBiO₃. *Phys. Rev. Lett.* **2019**, *122*, 186402.

(558) McAvoy, R. L.; Govoni, M.; Galli, G. Coupling First-Principles Calculations of Electron–Electron and Electron–Phonon Scattering, and Applications to Carbon-Based Nanostructures. *J. Chem. Theory Comput.* **2018**, *14*, 6269–6275.

(559) Basov, D. N.; Averitt, R. D.; Hsieh, D. Towards Properties on Demand in Quantum Materials. *Nat. Mater.* **2017**, *16*, 1077–1088.

(560) Rivera, N.; Christensen, T.; Narang, P. Phonon Polaritonics in Two-Dimensional Materials. *Nano Lett.* **2019**, *19*, 2653–2660. (561) Dai, S.; Fang, W.; Rivera, N.; Stehle, Y.; Jiang, B.-Y.; Shen, J.; Tay, R. Y.; Ciccarino, C. J.; Ma, Q.; Rodan-Legrain, D.; et al. Phonon Polaritons in Monolayers of Hexagonal Boron Nitride. *Adv. Mater.* **2019**, *31*, 1806603.

(562) Ebbesen, T. W. Hybrid Light–Matter States in a Molecular and Material Science Perspective. *Acc. Chem. Res.* **2016**, *49*, 2403–2412.

(563) Feist, J.; Galego, J.; Garcia-Vidal, F. J. Polaritonic Chemistry with Organic Molecules. *ACS Photonics* **2018**, *5*, 205–216.

(564) Ruggenthaler, M.; Tancogne-Dejean, N.; Flick, J.; Appel, H.; Rubio, A. From a Quantum-Electrodynamical Light–Matter Description to Novel Spectroscopies. *Nat. Rev. Chem.* **2018**, *2*, No. 0118.

(565) Hertzog, M.; Wang, M.; Mony, J.; Börjesson, K. Strong Light– Matter Interactions: a New Direction within Chemistry. *Chem. Soc. Rev.* **2019**, *48*, 937–961.

(566) Forn-Díaz, P.; Lamata, L.; Rico, E.; Kono, J.; Solano, E. Ultrastrong Coupling Regimes of Light-Matter Interaction. *Rev. Mod. Phys.* **2019**, *91*, No. 025005.

(567) Frisk Kockum, A.; Miranowicz, A.; De Liberato, S.; Savasta, S.; Nori, F. Ultrastrong Coupling Between Light and Matter. *Nat. Rev. Phys.* **2019**, *1*, 19–40.

(568) Herrera, F.; Owrutsky, J. Molecular Polaritons for Controlling Chemistry with Quantum Optics. J. Chem. Phys. **2020**, 152, 100902.

(569) Le Boité, A. Theoretical Methods for Ultrastrong LightMatter Interactions. *Adv. Quantum Technol.* **2020**, *3*, 1900140.

(570) Rivera, N.; Kaminer, I. Light-matter interactions with photonic quasiparticles. *Nat. Rev. Phys.* **2020**, *2*, 538–561.

(571) Horton, M. J.; Ojambati, O. S.; Chikkaraddy, R.; Deacon, W. M.; Kongsuwan, N.; Demetriadou, A.; Hess, O.; Baumberg, J. J. Nanoscopy through a Plasmonic Nanolens. *Proc. Natl. Acad. Sci. U. S. A.* **2020**, *117*, 2275–2281.

(572) Benz, F.; Schmidt, M. K.; Dreismann, A.; Chikkaraddy, R.; Zhang, Y.; Demetriadou, A.; Carnegie, C.; Ohadi, H.; de Nijs, B.; Esteban, R.; et al. Single-Molecule Optomechanics in "Picocavities. *Science* **2016**, 354, 726–729.

(573) Thomas, A.; Lethuillier-Karl, L.; Nagarajan, K.; Vergauwe, R. M. A.; George, J.; Chervy, T.; Shalabney, A.; Devaux, E.; Genet, C.; Moran, J.; et al. Tilting a Ground-State Reactivity Landscape by Vibrational Strong Coupling. *Science* **2019**, *363*, 615–619.

(574) Hutchison, J. A.; Schwartz, T.; Genet, C.; Devaux, E.; Ebbesen, T. W. Modifying Chemical Landscapes by Coupling to Vacuum Fields. *Angew. Chem., Int. Ed.* **2012**, *51*, 1592–1596.

(575) Munkhbat, B.; Wersäll, M.; Baranov, D. G.; Antosiewicz, T. J.; Shegai, T. Suppression of Photo-Oxidation of Organic Chromophores by Strong Coupling to Plasmonic Nanoantennas. *Sci. Adv.* **2018**, *4*, No. eaas9552.

(576) Jayaprakash, R.; Georgiou, K.; Coulthard, H.; Askitopoulos, A.; Rajendran, S. K.; Coles, D. M.; Musser, A. J.; Clark, J.; Samuel, I. D. W.; Turnbull, G. A.; et al. A Hybrid Organic–Inorganic Polariton LED. *Light: Sci. Appl.* **2019**, *8*, 81.

(577) Doppagne, B.; Neuman, T.; Soria-Martinez, R.; López, L. E. P.; Bulou, H.; Romeo, M.; Berciaud, S.; Scheurer, F.; Aizpurua, J.; Schull, G. Single-Molecule Tautomerization Tracking through Space- and Time-Resolved Fluorescence Spectroscopy. *Nat. Nanotechnol.* **2020**, *15*, 207–211.

(578) Eizner, E.; Martínez-Martínez, L. A.; Yuen-Zhou, J.; Kéna-Cohen, S. Inverting Singlet and Triplet Excited States using Strong Light-Matter Coupling. *Sci. Adv.* **2019**, *5*, No. eaax4482.

(579) Polak, D.; Jayaprakash, R.; Lyons, T. P.; Martínez-Martínez, L. Á.; Leventis, A.; Fallon, K. J.; Coulthard, H.; Bossanyi, D. G.; Georgiou, K.; Anthony J. Petty, I.; et al. Manipulating Molecules with Strong Coupling: Harvesting Triplet Excitons in Organic Exciton Microcavities. *Chemical Science* **2020**, *11*, 343–354.

(580) DelPo, C. A.; Kudisch, B.; Park, K. H.; Khan, S.-U.-Z.; Fassioli, F.; Fausti, D.; Rand, B. P.; Scholes, G. D. Polariton Transitions in Femtosecond Transient Absorption Studies of Ultrastrong Light–Molecule Coupling. J. Phys. Chem. Lett. **2020**, *11*, 2667–2674.

(581) Mueller, N. S.; Okamura, Y.; Vieira, B. G. M.; Juergensen, S.; Lange, H.; Barros, E. B.; Schulz, F.; Reich, S. Deep Strong Light–Matter Coupling in Plasmonic Nanoparticle Crystals. *Nature* **2020**, *583*, 780–784.

(582) Xiang, B.; Ribeiro, R. F.; Du, M.; Chen, L.; Yang, Z.; Wang, J.; Yuen-Zhou, J.; Xiong, W. Intermolecular Vibrational Energy Transfer Enabled by Microcavity Strong Light–Matter Coupling. *Science* **2020**, 368, 665–667.

(583) Hernández, F. J.; Herrera, F. Multi-Level Quantum Rabi Model for Anharmonic Vibrational Polaritons. *J. Chem. Phys.* **2019**, *151*, 144116.

(584) Neuman, T.; Esteban, R.; Casanova, D.; García-Vidal, F. J.; Aizpurua, J. Coupling of Molecular Emitters and Plasmonic Cavities beyond the Point-Dipole Approximation. *Nano Lett.* **2018**, *18*, 2358– 2364.

(585) Neuman, T. c. v.; Esteban, R.; Giedke, G.; Schmidt, M. K.; Aizpurua, J. Quantum Description of Surface-Enhanced Resonant Raman Scattering within a Hybrid-Optomechanical Model. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *100*, No. 043422.

(586) Neuman, T.; Aizpurua, J.; Esteban, R. Quantum Theory of Surface-Enhanced Resonant Raman Scattering (SERRS) of Molecules in Strongly Coupled PlasmonExciton Systems. *Nanophotonics* **2020**, *9*, 295–308.

(587) Neuman, T.; Aizpurua, J. Origin of the Asymmetric Light Emission from Molecular Exciton Polaritons. *Optica* **2018**, *5*, 1247–1255.

(588) Schütz, S.; Schachenmayer, J.; Hagenmüller, D.; Brennen, G. K.; Volz, T.; Sandoghdar, V.; Ebbesen, T. W.; Genes, C.; Pupillo, G. Ensemble-Induced Strong Light-Matter Coupling of a Single Quantum Emitter. *Phys. Rev. Lett.* **2020**, *124*, 113602.

(589) Du, M.; Martínez-Martínez, L. A.; Ribeiro, R. F.; Hu, Z.; Menon, V. M.; Yuen-Zhou, J. Theory for Polariton-Assisted Remote Energy Transfer. *Chem. Sci.* **2018**, *9*, 6659–6669.

(590) Mandal, A.; Krauss, T. D.; Huo, P. Polariton-Mediated Electron Transfer via Cavity Quantum Electrodynamics. *J. Phys. Chem. B* **2020**, *124*, 6321–6340.

(591) Schäfer, C.; Ruggenthaler, M.; Appel, H.; Rubio, A. Modification of Excitation and Charge Transfer in Cavity Quantum-Electrodynamical Chemistry. *Proc. Natl. Acad. Sci. U. S. A.* **2019**, *116*, 4883–4892.

(592) Ruggenthaler, M.; Flick, J.; Pellegrini, C.; Appel, H.; Tokatly, I. V.; Rubio, A. Quantum-Electrodynamical Density-Functional Theory: Bridging Quantum Optics and Electronic-Structure Theory. *Phys. Rev. A: At., Mol., Opt. Phys.* **2014**, *90*, No. 012508.

(593) Tokatly, I. V. Time-Dependent Density Functional Theory for Many-Electron Systems Interacting with Cavity Photons. *Phys. Rev. Lett.* **2013**, *110*, 233001.

(594) Flick, J.; Ruggenthaler, M.; Appel, H.; Rubio, A. Kohn–Sham Approach to Quantum Electrodynamical Density-Functional Theory: Exact Time-Dependent Effective Potentials in Real Space. *Proc. Natl. Acad. Sci. U. S. A.* **2015**, *112*, 15285–15290.

(595) Dimitrov, T.; Flick, J.; Ruggenthaler, M.; Rubio, A. Exact Functionals for Correlated Electron–Photon Systems. *New J. Phys.* **2017**, *19*, 113036.

(596) Pellegrini, C.; Flick, J.; Tokatly, I. V.; Appel, H.; Rubio, A. Optimized Effective Potential for Quantum Electrodynamical Time-Dependent Density Functional Theory. *Phys. Rev. Lett.* **2015**, *115*, No. 093001.

(597) Flick, J.; Schäfer, C.; Ruggenthaler, M.; Appel, H.; Rubio, A. Ab Initio Optimized Effective Potentials for Real Molecules in Optical Cavities: Photon Contributions to the Molecular Ground State. *ACS Photonics* **2018**, *5*, 992–1005.

(598) Tancogne-Dejean, N.; Oliveira, M. J. T.; Andrade, X.; Appel, H.; Borca, C. H.; Le Breton, G.; Buchholz, F.; Castro, A.; Corni, S.; Correa, A. A.; et al. Octopus, a Computational Framework for Exploring Light-Driven Phenomena and Quantum Dynamics in Extended and Finite Systems. J. Chem. Phys. **2020**, 152, 124119.

(599) Flick, J.; Welakuh, D. M.; Ruggenthaler, M.; Appel, H.; Rubio, A. Light–Matter Response in Nonrelativistic Quantum Electrodynamics. *ACS Photonics* **2019**, *6*, 2757–2778. (600) Flick, J.; Narang, P. Ab Initio Polaritonic Potential-Energy Surfaces for Excited-State Nanophotonics and Polaritonic Chemistry. J. Chem. Phys. **2020**, 153, No. 094116.

(601) Wang, D. S.; Neuman, T.; Flick, J.; Narang, P. Weak-to-Strong Light-Matter Coupling and Dissipative Dynamics from First Principles. 2020; arXiv:2002.10461 [quant-ph]; https://arxiv.org/abs/2002. 10461.

(602) Rivera, N.; Flick, J.; Narang, P. Variational Theory of Nonrelativistic Quantum Electrodynamics. *Phys. Rev. Lett.* **2019**, *122*, 193603.

(603) Vendrell, O. Coherent Dynamics in Cavity Femtochemistry: Application of the Multi-Configuration Time-Dependent Hartree Method. *Chem. Phys.* **2018**, *509*, 55–65.

(604) Mordovina, U.; Bungey, C.; Appel, H.; Knowles, P. J.; Rubio, A.; Manby, F. R. Polaritonic Coupled-Cluster Theory. *Phys. Rev. Res.* **2020**, *2*, No. 023262.

(605) Haugland, T. S.; Ronca, E.; Kjønstad, E. F.; Rubio, A.; Koch, H. Coupled Cluster Theory for Molecular Polaritons: Changing Ground and Excited States. 2020; arXiv:2005.04477 [physics.chem-ph]; https://arxiv.org/abs/2005.04477.

(606) Hoffmann, N. M.; Appel, H.; Rubio, A.; Maitra, N. T. Light-Matter Interactions via the Exact Factorization Approach. *Eur. Phys. J. B* **2018**, *91*, 180.

(607) Abedi, A.; Khosravi, E.; Tokatly, I. V. Shedding Light on Correlated Electron–Photon States using the Exact Factorization. *Eur. Phys. J. B* **2018**, *91*, 194.

(608) Albareda, G.; Kelly, A.; Rubio, A. Nonadiabatic Quantum Dynamics without Potential Energy Surfaces. *Phys. Rev. Mater.* **2019**, *3*, No. 023803.

(609) Galego, J.; Garcia-Vidal, F. J.; Feist, J. Cavity-Induced Modifications of Molecular Structure in the Strong-Coupling Regime. *Phys. Rev. X* **2015**, *5*, No. 041022.

(610) Flick, J.; Ruggenthaler, M.; Appel, H.; Rubio, A. Atoms and Molecules in Cavities, from Weak to Strong Coupling in Quantum-Electrodynamics (QED) Chemistry. *Proc. Natl. Acad. Sci. U. S. A.* 2017, *114*, 3026–3034.

(611) Flick, J.; Appel, H.; Ruggenthaler, M.; Rubio, A. Cavity Born-Oppenheimer Approximation for Correlated Electron-Nuclear-Photon Systems. J. Chem. Theory Comput. **2017**, *13*, 1616–1625.

(612) Covington, C.; Kidd, D.; Buckner, H.; Appel, H.; Varga, K. Time Propagation of the Coupled Maxwell and Kohn-Sham Equations using the Riemann-Silberstein Formalism. *Phys. Rev. E: Stat. Phys., Plasmas, Fluids, Relat. Interdiscip. Top.* **2019**, *100*, No. 053301.

(613) Jestädt, R.; Ruggenthaler, M.; Oliveira, M. J. T.; Rubio, A.; Appel, H. Light-Matter Interactions within the Ehrenfest–Maxwell– Pauli–Kohn–Sham Framework: Fundamentals, Implementation, and Nano-Optical Applications. *Adv. Phys.* **2019**, *68*, 225–333.

(614) Buhmann, S. Y.; Welsch, D.-G. Dispersion Forces in Macroscopic Quantum Electrodynamics. *Prog. Quantum Electron.* 2007, 31, 51–130.

(615) Buhmann, S. Y.; Butcher, D. T.; Scheel, S. Macroscopic Quantum Electrodynamics in Nonlocal and Nonreciprocal Media. *New J. Phys.* **2012**, *14*, No. 083034.

(616) Feist, J.; Fernández-Dománguez, A. I.; García-Vidal, F. J. Macroscopic QED for Quantum Nanophotonics: Emitter-Centered Modes as a Minimal Basis for Multi-Emitter Problems. 2020; arXiv:2008.02106 [quant-ph]; https://arxiv.org/abs/2008.02106.

(617) Cortese, E.; Carusotto, I.; Colombelli, R.; De Liberato, S. Strong Coupling of Ionizing Transitions. *Optica* **2019**, *6*, 354–361.

(618) Sidler, D.; Ruggenthaler, M.; Appel, H.; Rubio, A. Chemistry in Quantum Cavities: Exact Results, the Impact of Thermal Velocities, and Modified Dissociation. *J. Phys. Chem. Lett.* **2020**, *11*, 7525–7530. (619) Cortese, E.; Tran, N.-L.; Manceau, J.-M.; Bousseksou, A.; Carusotto, I.; Biasiol, G.; Colombelli, R.; De Liberato, S. Excitons Bound by Photon Exchange. *Nat. Phys.* **2020**, DOI: 10.1038/s41567-020-0994-6.

(620) Baranov, D. G.; Munkhbat, B.; Zhukova, E.; Bisht, A.; Canales, A.; Rousseaux, B.; Johansson, G.; Antosiewicz, T. J.; Shegai, T.

Ultrastrong Coupling Between Nanoparticle Plasmons and Cavity Photons at Ambient Conditions. *Nat. Commun.* **2020**, *11*, 2715.

(621) Niemczyk, T.; Deppe, F.; Huebl, H.; Menzel, E. P.; Hocke, F.; Schwarz, M. J.; Garcia-Ripoll, J. J.; Zueco, D.; Hümmer, T.; Solano, E.; et al. Circuit Quantum Electrodynamics in the Ultrastrong-Coupling Regime. *Nat. Phys.* **2010**, *6*, 772–776.

(622) Galego, J.; Climent, C.; Garcia-Vidal, F. J.; Feist, J. Cavity Casimir-Polder Forces and Their Effects in Ground-State Chemical Reactivity. *Phys. Rev. X* 2019, *9*, No. 021057.

(623) Thomas, A.; Devaux, E.; Nagarajan, K.; Chervy, T.; Seidel, M.; Hagenmüller, D.; Schütz, S.; Schachenmayer, J.; Genet, C.; Pupillo, G. et al. Exploring Superconductivity under Strong Coupling with the Vacuum Electromagnetic Field. 2019; arXiv:1911.01459 [condmat.supr-con]; https://arxiv.org/abs/1911.01459.

(624) Huang, J.; Liu, Y.-H.; Huang, J.-F.; Liao, J.-Q. Generation of Macroscopic Entangled Cat States in a Longitudinally Coupled Cavity-QED Model. *Phys. Rev. A: At., Mol., Opt. Phys.* **2020**, *101*, No. 043841.

(625) Nova, T. F.; Disa, A. S.; Fechner, M.; Cavalleri, A. Metastable Ferroelectricity in Optically Strained SrTiO3. *Science* **2019**, *364*, 1075– 1079.

(626) Juraschek, D. M.; Neuman, T.; Flick, J.; Narang, P. Cavity Control of Nonlinear Phononics. 2019; arXiv:1912.00122 [condmat.mtrl-sci]; https://arxiv.org/abs/1912.00122.

(627) Mazza, G.; Georges, A. Superradiant Quantum Materials. *Phys. Rev. Lett.* **2019**, *122*, No. 017401.

(628) Nicoletti, D.; Cavalleri, A. Nonlinear Light-Matter Interaction at Terahertz Frequencies. *Adv. Opt. Photonics* **2016**, *8*, 401–464.

(629) Kennes, D. M.; Wilner, E. Y.; Reichman, D. R.; Millis, A. J. Transient superconductivity from electronic squeezing of optically pumped phonons. *Nat. Phys.* **2017**, *13*, 479–483.

(630) Polli, D.; Rini, M.; Wall, S.; Schoenlein, R. W.; Tomioka, Y.; Tokura, Y.; Cerullo, G.; Cavalleri, A. Coherent Orbital Waves in the Photo-Induced Insulator-Metal Dynamics of a Magnetoresistive Manganite. *Nat. Mater.* **2007**, *6*, 643–647.

(631) Wang, D.; Liu, Z.; Cao, J.; Fan, H. Tunable Band Topology Reflected by Fractional Quantum Hall States in Two-Dimensional Lattices. *Phys. Rev. Lett.* **2013**, *111*, 186804.

(632) McIver, J. W.; Schulte, B.; Stein, F.-U.; Matsuyama, T.; Jotzu, G.; Meier, G.; Cavalleri, A. Light-Induced Anomalous Hall Effect in Graphene. *Nat. Phys.* **2020**, *16*, 38–41.

(633) Li, P.-F.; Liao, W.-Q.; Tang, Y.-Y.; Qiao, W.; Zhao, D.; Ai, Y.; Yao, Y.-F.; Xiong, R.-G. Organic Enantiomeric High-Tc Ferroelectrics. *Proc. Natl. Acad. Sci. U. S. A.* **2019**, *116*, 5878–5885.

(634) Hu, W.; Kaiser, S.; Nicoletti, D.; Hunt, C. R.; Gierz, I.; Hoffmann, M. C.; Le Tacon, M.; Loew, T.; Keimer, B.; Cavalleri, A. Optically Enhanced Coherent Transport in YBa2Cu3O6.5 by Ultrafast Redistribution of Interlayer Coupling. *Nat. Mater.* **2014**, *13*, 705–711.

(635) Mitrano, M.; Cantaluppi, A.; Nicoletti, D.; Kaiser, S.; Perucchi, A.; Lupi, S.; Di Pietro, P.; Pontiroli, D.; Riccò, M.; Clark, S. R.; et al. Possible Light-Induced Superconductivity in K3C60 at High Temperature. *Nature* **2016**, *530*, 461–464.

(636) Mankowsky, R.; Subedi, A.; Först, M.; Mariager, S.; Chollet, M.; Lemke, H.; Robinson, J.; Glownia, J.; Minitti, M.; Frano, A. et al. Transient Atomic Structure of Vibrationally Excited YBCO with Enhanced Superconducting Coherence Above Tc. Presented at the 19th International Conference on Ultrafast Phenomena. 2014; p 07.Mon.D.1.

(637) Deng, H.; Weihs, G.; Santori, C.; Bloch, J.; Yamamoto, Y. Condensation of Semiconductor Microcavity Exciton Polaritons. *Science* **2002**, *298*, 199–202.

(638) Lerario, G.; Fieramosca, A.; Barachati, F.; Ballarini, D.; Daskalakis, K. S.; Dominici, L.; De Giorgi, M.; Maier, S. A.; Gigli, G.; Kéna-Cohen, S.; Sanvitto, D. Room-Temperature Superfluidity in a Polariton Condensate. *Nat. Phys.* **2017**, *13*, 837–841.

(639) Slootsky, M.; Liu, X.; Menon, V. M.; Forrest, S. R. Room Temperature Frenkel-Wannier-Mott Hybridization of Degenerate Excitons in a Strongly Coupled Microcavity. *Phys. Rev. Lett.* **2014**, *112*, No. 076401. (640) Li, J.; Golez, D.; Mazza, G.; Millis, A. J.; Georges, A.; Eckstein, M. Electromagnetic Coupling in Tight-Binding Models for Strongly Correlated Light and Matter. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2020**, *101*, 205140.

(641) Sun, Z.; Millis, A. J. Bardasis-Schrieffer Polaritons in Excitonic Insulators. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2020**, *102*, No. 041110.

(642) Ashida, Y.; Imamoglu, A.; Faist, J.; Jaksch, D.; Cavalleri, A.; Demler, E. Quantum Electrodynamic Control of Matter: Cavity-Enhanced Ferroelectric Phase Transition. 2020; arXiv:2003.13695 [cond-mat.mes-hall]; https://arxiv.org/abs/2003.13695.

(643) Laussy, F. P.; Kavokin, A. V.; Shelykh, I. A. Exciton-Polariton Mediated Superconductivity. *Phys. Rev. Lett.* **2010**, *104*, 106402.

(644) Sentef, M. A.; Ruggenthaler, M.; Rubio, A. Cavity Quantum-Electrodynamical Polaritonically Enhanced Electron-Phonon Coupling and its Influence on Superconductivity. *Sci. Adv.* **2018**, *4*, No. eaau6969.

(645) Schlawin, F.; Cavalleri, A.; Jaksch, D. Cavity-Mediated Electron-Photon Superconductivity. *Phys. Rev. Lett.* **2019**, *122*, 133602.

(646) Curtis, J. B.; Raines, Z. M.; Allocca, A. A.; Hafezi, M.; Galitski, V. M. Cavity Quantum Eliashberg Enhancement of Superconductivity. *Phys. Rev. Lett.* **2019**, *122*, 167002.

(647) Gao, H.; Schlawin, F.; Buzzi, M.; Cavalleri, A.; Jaksch, D. Photoinduced Electron Pairing in a Driven Cavity. *Phys. Rev. Lett.* **2020**, *125*, No. 053602.

(648) Manikandan, P.; Keshavamurthy, S. Dynamical Traps Lead to the Slowing Down of Intramolecular Vibrational Energy Flow. *Proc. Natl. Acad. Sci. U. S. A.* **2014**, *111*, 14354–14359.

(649) Grafton, A. B.; Dunkelberger, A. D.; Simpkins, B. S.; Triana, J. F.; Hernandez, F. J.; Herrera, F.; Owrutsky, J. Excited-State Vibration-Polariton Transitions and Dynamics in Nitroprusside. 2020; chemrxiv:12518555.v1; https://chemrxiv.org/articles/preprint/Excited-State_Vibration-Polariton_Transitions_and_Dynamics_in_ Nitroprusside/12518555.

(650) Carleo, G.; Troyer, M. Solving the Quantum Many-Body Problem with Artificial Neural Networks. *Science* 2017, 355, 602–606.
(651) Choo, K.; Mezzacapo, A.; Carleo, G. Fermionic Neural-Network States for Ab-Initio Electronic Structure. *Nat. Commun.* 2020, 11, 2368.

(652) Welborn, M.; Cheng, L.; Miller, T. F. Transferability in Machine Learning for Electronic Structure via the Molecular Orbital Basis. *J. Chem. Theory Comput.* **2018**, *14*, 4772–4779.

(653) Cheng, L.; Welborn, M.; Christensen, A. S.; Miller, T. F. A Universal Density Matrix Functional from Molecular Orbital-Based Machine Learning: Transferability across Organic Molecules. *J. Chem. Phys.* **2019**, *150*, 131103.

(654) Cheng, L.; Kovachki, N. B.; Welborn, M.; Miller, T. F. Regression Clustering for Improved Accuracy and Training Costs with Molecular-Orbital-Based Machine Learning. *J. Chem. Theory Comput.* **2019**, *15*, 6668–6677.

(655) Pfau, D.; Spencer, J. S.; de G. Matthews, A. G.; Foulkes, W. M. C. Ab-Initio Solution of the Many-Electron Schrödinger Equation with Deep Neural Networks. *Phys. Rev. Res.* **2020**, *2* (3), 033429.

(656) Schütt, K. T.; Gastegger, M.; Tkatchenko, A.; Müller, K.-R.; Maurer, R. J. Unifying Machine Learning and Quantum Chemistry with a Deep Neural Network for Molecular Wavefunctions. *Nat. Commun.* **2019**, *10*, 5024.

(657) Hermann, J.; Schätzle, Z.; Noé, F. Deep Neural Network Solution of the Electronic Schrödinger equation. *Nat. Chem.* **2020**, *12*, 891–897.

(658) Feynman, R. P. Simulating Physics with Computers. Int. J. Theor. Phys. **1982**, 21, 467–488.

(659) Bernien, H.; Schwartz, S.; Keesling, A.; Levine, H.; Omran, A.; Pichler, H.; Choi, S.; Zibrov, A. S.; Endres, M.; Greiner, M.; et al. Probing Many-Body Dynamics on a 51-Atom Quantum Simulator. *Nature* **2017**, 551, 579–584.

(660) Schilling, A.; Cantoni, M.; Guo, J. D.; Ott, H. R. Superconductivity Above 130 K in the Hg-Ba-Ca-Cu-O system. *Nature* **1993**, 363, 56–58. (661) Drozdov, A. P.; Kong, P. P.; Minkov, V. S.; Besedin, S. P.; Kuzovnikov, M. A.; Mozaffari, S.; Balicas, L.; Balakirev, F. F.; Graf, D. E.; Prakapenka, V. B.; et al. Superconductivity at 250 K in Lanthanum Hydride Under High Pressures. *Nature* **2019**, *569*, 528–531.

(662) Reiher, M.; Wiebe, N.; Svore, K. M.; Wecker, D.; Troyer, M. Elucidating Reaction Mechanisms on Quantum Computers. *Proc. Natl. Acad. Sci. U. S. A.* **2017**, *114*, 7555–7560.

(663) Preskill, J. Quantum Computing in the NISQ Era and Beyond. *Quantum* **2018**, *2*, 79.

(664) Cao, Y.; Romero, J.; Olson, J. P.; Degroote, M.; Johnson, P. D.; Kieferová, M.; Kivlichan, I. D.; Menke, T.; Peropadre, B.; Sawaya, N. P. D.; et al. Quantum Chemistry in the Age of Quantum Computing. *Chem. Rev.* **2019**, *119*, 10856–10915.

(665) Bauer, B.; Bravyi, S.; Motta, M.; Chan, G. K.-L. Quantum Algorithms for Quantum Chemistry and Quantum Materials Science. 2020; arXiv:2001.03685 [quant-ph]; https://arxiv.org/abs/2001. 03685.

(666) McArdle, S.; Endo, S.; Aspuru-Guzik, A.; Benjamin, S. C.; Yuan, X. Quantum Computational Chemistry. *Rev. Mod. Phys.* **2020**, *92*, No. 015003.

(667) Grover, L. K. A Fast Quantum Mechanical Algorithm for Database Search. *Proceedings of the Twenty-Eighth Annual ACM Symposium on Theory of Computing*. New York, NY, USA, 1996; p 212–219.

(668) Shor, P. W. Polynomial-Time Algorithms for Prime Factorization and Discrete Logarithms on a Quantum Computer. *SIAM J. Comput.* **1997**, *26*, 1484–1509.

(669) Lidar, D. A.; Biham, O. Simulating Ising Spin Glasses on a Quantum Computer. *Phys. Rev. E: Stat. Phys., Plasmas, Fluids, Relat. Interdiscip. Top.* **1997**, *56*, 3661–3681.

(670) Lidar, D. A.; Wang, H. Calculating the Thermal Rate Constant with Exponential Speedup on a Quantum Computer. *Phys. Rev. E: Stat. Phys., Plasmas, Fluids, Relat. Interdiscip. Top.* **1999**, *59*, 2429–2438.

(671) Ortiz, G.; Gubernatis, J. E.; Knill, E.; Laflamme, R. Quantum Algorithms for Fermionic Simulations. *Phys. Rev. A: At., Mol., Opt. Phys.* **2001**, *64*, No. 022319.

(672) Kitaev, A. Y. Quantum Measurements and the Abelian Stabilizer Problem. 1995; arXiv:quant-ph/9511026 [quant-ph]; https://arxiv. org/abs/quant-ph/9511026.

(673) Kivlichan, I. D.; Gidney, C.; Berry, D. W.; Wiebe, N.; McClean, J.; Sun, W.; Jiang, Z.; Rubin, N.; Fowler, A.; Aspuru-Guzik, A.; et al. Improved Fault-Tolerant Quantum Simulation of Condensed-Phase Correlated Electrons via Trotterization. *Quantum* **2020**, *4*, 296.

(674) Nielson, M. A.; Chuang, I. L. Quantum Computation and Quantum Information, 10th Anniversary ed.; Cambridge University Press, 2011.

(675) Abrams, D. S.; Lloyd, S. Simulation of Many-Body Fermi Systems on a Universal Quantum Computer. *Phys. Rev. Lett.* **1997**, *79*, 2586–2589.

(676) Abrams, D. S.; Lloyd, S. Quantum Algorithm Providing Exponential Speed Increase for Finding Eigenvalues and Eigenvectors. *Phys. Rev. Lett.* **1999**, *83*, 5162–5165.

(677) Aspuru-Guzik, A.; Dutoi, A. D.; Love, P. J.; Head-Gordon, M. Simulated Quantum Computation of Molecular Energies. *Science* **2005**, 309, 1704–1707.

(678) Wang, H.; Kais, S.; Aspuru-Guzik, A.; Hoffmann, M. R. Quantum Algorithm for Obtaining the Energy Spectrum of Molecular systems. *Phys. Chem. Chem. Phys.* **2008**, *10*, 5388–5393.

(679) Bravyi, S.; DiVincenzo, D. P.; Loss, D.; Terhal, B. M. Quantum Simulation of Many-Body Hamiltonians Using Perturbation Theory with Bounded-Strength Interactions. *Phys. Rev. Lett.* **2008**, *101*, No. 070503.

(680) Lloyd, S. Universal Quantum Simulators. *Science* **1996**, *273*, 1073–1078.

(681) Peruzzo, A.; McClean, J.; Shadbolt, P.; Yung, M.-H.; Zhou, X.-Q.; Love, P. J.; Aspuru-Guzik, A.; O'Brien, J. L. A Variational Eigenvalue Solver on a Photonic Quantum Processor. *Nat. Commun.* **2014**, *5*, 4213. (682) McClean, J. R.; Romero, J.; Babbush, R.; Aspuru-Guzik, A. The Theory of Variational Hybrid Quantum-Classical Algorithms. *New J. Phys.* **2016**, *18*, No. 023023.

(683) Kandala, A.; Mezzacapo, A.; Temme, K.; Takita, M.; Brink, M.; Chow, J. M.; Gambetta, J. M. Hardware-Efficient Variational Quantum Eigensolver for Small Molecules and Quantum Magnets. *Nature* **2017**, *549*, 242–246.

(684) Li, Y.; Benjamin, S. C. Efficient Variational Quantum Simulator Incorporating Active Error Minimization. *Phys. Rev. X* 2017, *7*, No. 021050.

(685) Kutzelnigg, W. Error Analysis and Improvements of Coupled-Cluster Theory. *Theor. Chim. Acta* **1991**, *80*, 349–386.

(686) Harsha, G.; Shiozaki, T.; Scuseria, G. E. On the Difference between Variational and Unitary Coupled Cluster Theories. *J. Chem. Phys.* **2018**, *148*, No. 044107.

(687) Steudtner, M.; Wehner, S. Fermion-to-Qubit Mappings with Varying Resource Requirements for Quantum Simulation. *New J. Phys.* **2018**, *20*, No. 063010.

(688) Seeley, J. T.; Richard, M. J.; Love, P. J. The Bravyi-Kitaev transformation for quantum computation of electronic structure. *J. Chem. Phys.* **2012**, *137*, 224109.

(689) Bravyi, S. B.; Kitaev, A. Y. Fermionic Quantum Computation. *Ann. Phys. (Amsterdam, Neth.)* **2002**, *298*, 210–226.

(690) Jiang, Z.; Kalev, A.; Mruczkiewicz, W.; Neven, H. Optimal Fermion-to-Qubit Mapping via Ternary Trees with Applications to Reduced Quantum States Learning. *Quantum* **2020**, *4*, 276–286.

(691) Poulin, D.; Hastings, M. B.; Wecker, D.; Wiebe, N.; Doherty, A. C.; Troyer, M. The Trotter Step Size Required for Accurate Quantum Simulation of Quantum Chemistry. 2014; arXiv:1406.4920 [quant-ph]; https://arxiv.org/abs/1406.4920.

(692) O'Malley, P. J. J.; Babbush, R.; Kivlichan, I. D.; Romero, J.; McClean, J. R.; Barends, R.; Kelly, J.; Roushan, P.; Tranter, A.; Ding, N.; et al. Scalable Quantum Simulation of Molecular Energies. *Phys. Rev. X* **2016**, *6*, No. 031007.

(693) Grimsley, H. R.; Economou, S. E.; Barnes, E.; Mayhall, N. J. An Adaptive Variational Algorithm for Exact Molecular Simulations on a Quantum Computer. *Nat. Commun.* **2019**, *10*, 3007.

(694) Evangelista, F. A. Alternative Single-Reference Coupled Cluster Approaches for Multireference Problems: The Simpler, the Better. *J. Chem. Phys.* **2011**, *134*, 224102.

(695) Xia, R.; Kais, S. Coupled Cluster Singles and Doubles Variational Quantum Eigensolver Ansatz for Electronic Structure Calculations. 2020; arXiv:2005.08451 [quant-ph]; https://arxiv.org/abs/2005.08451.

(696) Ryabinkin, I. G.; Genin, S. N.; Izmaylov, A. F. Constrained Variational Quantum Eigensolver: Quantum Computer Search Engine in the Fock Space. J. Chem. Theory Comput. **2019**, *15*, 249–255.

(697) Colless, J. I.; Ramasesh, V. V.; Dahlen, D.; Blok, M. S.; Kimchi-Schwartz, M. E.; McClean, J. R.; Carter, J.; de Jong, W. A.; Siddiqi, I. Computation of Molecular Spectra on a Quantum Processor with an Error-Resilient Algorithm. *Phys. Rev. X* **2018**, *8*, No. 011021.

(698) Shen, Y.; Zhang, X.; Zhang, S.; Zhang, J.-N.; Yung, M.-H.; Kim, K. Quantum Implementation of the Unitary Coupled Cluster for Simulating Molecular Electronic Structure. *Phys. Rev. A: At., Mol., Opt. Phys.* **2017**, *95*, No. 020501.

(699) Hempel, C.; Maier, C.; Romero, J.; McClean, J.; Monz, T.; Shen, H.; Jurcevic, P.; Lanyon, B. P.; Love, P.; Babbush, R.; et al. Quantum Chemistry Calculations on a Trapped-Ion Quantum Simulator. *Phys. Rev. X* **2018**, *8*, No. 031022.

(700) Lee, J.; Huggins, W. J.; Head-Gordon, M.; Whaley, K. B. Generalized Unitary Coupled Cluster Wave functions for Quantum Computation. *J. Chem. Theory Comput.* **2019**, *15*, 311–324.

(701) Gard, B. T.; Zhu, L.; Barron, G. S.; Mayhall, N. J.; Economou, S. E.; Barnes, E. Efficient Symmetry-Preserving State Preparation Circuits for the Variational Quantum Eigensolver Algorithm. *NPJ. Quantum Inf.* **2020**, *6*, 10.

(702) Ma, H.; Govoni, M.; Galli, G. Quantum Simulations of Materials on Near-Term Quantum Computers. *NPJ. Comput. Mater.* **2020**, *6*, 85.

(703) Bauer, B.; Wecker, D.; Millis, A. J.; Hastings, M. B.; Troyer, M. Hybrid Quantum-Classical Approach to Correlated Materials. *Phys. Rev. X* **2016**, *6*, No. 031045.

(704) Rungger, I.; Fitzpatrick, N.; Chen, H.; Alderete, C. H.; Apel, H.; Cowtan, A.; Patterson, A.; Ramo, D. M.; Zhu, Y.; Nguyen, N. H. et al. Dynamical Mean Field Theory Algorithm and Experiment on Quantum Computers. 2019; arXiv:1910.04735 [quant-ph]; https://arxiv.org/ abs/1910.04735.

(705) Jaderberg, B.; Agarwal, A.; Leonhardt, K.; Kiffner, M.; Jaksch, D. Minimum Hardware Requirements for Hybrid Quantum–Classical DMFT. *Quantum Sci. Technol.* **2020**, *5*, No. 034015.

(706) Rubin, N. C. A Hybrid Classical/Quantum Approach for Large-Scale Studies of Quantum Systems with Density Matrix Embedding Theory. 2016; arXiv:1610.06910 [quant-ph]; https://arxiv.org/abs/1610.06910.

(707) McClean, J. R.; Kimchi-Schwartz, M. E.; Carter, J.; de Jong, W. A. Hybrid Quantum-Classical Hierarchy for Mitigation of Decoherence and Determination of Excited States. *Phys. Rev. A: At., Mol., Opt. Phys.* **2017**, *95*, No. 042308.

(708) Higgott, O.; Wang, D.; Brierley, S. Variational Quantum Computation of Excited States. *Quantum* **2019**, *3*, 156–166.

(709) Nakanishi, K. M.; Mitarai, K.; Fujii, K. Subspace-Search Variational Quantum Eigensolver for Excited States. *Phys. Rev. Res.* **2019**, *1*, No. 033062.

(710) Santagati, R.; Wang, J.; Gentile, A. A.; Paesani, S.; Wiebe, N.; McClean, J. R.; Morley-Short, S.; Shadbolt, P. J.; Bonneau, D.; Silverstone, J. W.; et al. Witnessing Eigenstates for Quantum Simulation of Hamiltonian Spectra. *Sci. Adv.* **2018**, *4*, No. eaap9646.

(711) Parrish, R. M.; Hohenstein, E. G.; McMahon, P. L.; Martínez, T. J. Quantum Computation of Electronic Transitions Using a Variational Quantum Eigensolver. *Phys. Rev. Lett.* **2019**, *122*, 230401.

(712) Yoshioka, N.; Nakagawa, Y. O.; Mitarai, K.; Fujii, K. Variational Quantum Algorithm for Non-equilibrium Steady States. 2019; arXiv:1908.09836 [quant-ph]; https://arxiv.org/abs/1908.09836.

(713) Schlegel, H. B. Geometry Optimization. Wiley Interdiscip. Rev.: Comput. Mol. Sci. 2011, 1, 790–809.

(714) Schuld, M.; Bergholm, V.; Gogolin, C.; Izaac, J.; Killoran, N. Evaluating Analytic Gradients on Quantum Hardware. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 032331.

(715) Mitarai, K.; Nakagawa, Y. O.; Mizukami, W. Theory of Analytical Energy Derivatives for the Variational Quantum Eigensolver. *Phys. Rev. Res.* **2020**, *2*, No. 013129.

(716) Stober, S. T.; Harwood, S. M.; Greenberg, D.; Gujarati, T. P.; Mostame, S.; Raman, S.; Trenev, D. Computing Thermodynamic Observables on Noisy Quantum Computers with Chemical Accuracy. 2020; arXiv:2003.02303 [physics.chem-ph]; https://arxiv.org/abs/ 2003.02303.

(717) Seki, K.; Shirakawa, T.; Yunoki, S. Symmetry-Adapted Variational Quantum Eigensolver. *Phys. Rev. A: At., Mol., Opt. Phys.* **2020**, *101*, No. 052340.

(718) Takeshita, T.; Rubin, N. C.; Jiang, Z.; Lee, E.; Babbush, R.; McClean, J. R. Increasing the Representation Accuracy of Quantum Simulations of Chemistry without Extra Quantum Resources. *Phys. Rev.* X **2020**, *10*, No. 011004.

(719) Smart, S. E.; Mazziotti, D. A. Efficient Two-Electron Ansatz for Benchmarking Quantum Chemistry on a Quantum Computer. *Phys. Rev. Res.* **2020**, *2*, No. 023048.

(720) Smart, S. E.; Mazziotti, D. A. Quantum Solver of Contracted Eigenvalue Equations for Scalable Molecular Simulations on Quantum Computing Devices. 2020; arXiv:2004.11416 [quant-ph]; https://arxiv.org/abs/2004.11416.

(721) Garey, M. R.; Johnson, D. S. Computers and Intractability; A Guide to the Theory of NP-Completeness; W. H. Freeman & Co.: USA, 1990.

(722) Farhi, E.; Goldstone, J.; Gutmann, S. A Quantum Approximate Optimization Algorithm. 2014; arXiv:1411.4028 [quant-ph]; https://arXiv.org/abs/1411.4028.

(723) Moll, N.; Barkoutsos, P.; Bishop, L. S.; Chow, J. M.; Cross, A.; Egger, D. J.; Filipp, S.; Fuhrer, A.; Gambetta, J. M.; Ganzhorn, M.; et al. Quantum optimization using variational algorithms on near-term quantum devices. *Quantum Sci. Technol.* **2018**, *3*, No. 030503.

(724) Zhou, L.; Wang, S.-T.; Choi, S.; Pichler, H.; Lukin, M. D. Quantum Approximate Optimization Algorithm: Performance, Mechanism, and Implementation on Near-Term Devices. *Phys. Rev. X* **2020**, *10*, No. 021067.

(725) Goto, H.; Tatsumura, K.; Dixon, A. R. Combinatorial Optimization by Simulating Adiabatic Bifurcations in Nonlinear Hamiltonian Systems. *Sci. Adv.* **2019**, *5*, eaav2372.

(726) Harrow, A. W.; Hassidim, A.; Lloyd, S. Quantum Algorithm for Linear Systems of Equations. *Phys. Rev. Lett.* **2009**, *103*, 150502.

(727) Berry, D. W. High-order Quantum Algorithm for Solving Linear Differential Equations. J. Phys. A: Math. Theor. **2014**, 47, 105301.

(728) Wiebe, N.; Braun, D.; Lloyd, S. Quantum Algorithm for Data Fitting. *Phys. Rev. Lett.* **2012**, *109*, No. 050505.

(729) Lubasch, M.; Joo, J.; Moinier, P.; Kiffner, M.; Jaksch, D. Variational Quantum Algorithms for Nonlinear Problems. *Phys. Rev. A: At., Mol., Opt. Phys.* **2020**, *101*, No. 010301.

(730) King, A. D.; Carrasquilla, J.; Raymond, J.; Ozfidan, I.; Andriyash, E.; Berkley, A.; Reis, M.; Lanting, T.; Harris, R.; Altomare, F.; et al. Observation of Topological Phenomena in a Programmable Lattice of 1,800 Qubits. *Nature* **2018**, *560*, 456–460.

(731) Babbush, R.; Love, P. J.; Aspuru-Guzik, A. Adiabatic Quantum Simulation of Quantum Chemistry. *Sci. Rep.* **2015**, *4*, 6603.

(732) Lechner, W.; Hauke, P.; Zoller, P. A Quantum Annealing Architecture with All-to-All Connectivity from Local Interactions. *Sci. Adv.* **2015**, *1*, No. e1500838.

(733) Cong, I.; Choi, S.; Lukin, M. D. Quantum Convolutional Neural Networks. *Nat. Phys.* **2019**, *15*, 1273–1278.

(734) Temme, K.; Bravyi, S.; Gambetta, J. M. Error Mitigation for Short-Depth Quantum Circuits. *Phys. Rev. Lett.* **2017**, *119*, 180509.

(735) Kandala, A.; Temme, K.; Córcoles, A. D.; Mezzacapo, A.; Chow, J. M.; Gambetta, J. M. Error Mitigation Extends the Computational Reach of a Noisy Quantum Processor. *Nature* **2019**, *567*, 491–495.

(736) Otten, M.; Gray, S. K. Recovering Noise-Free Quantum Observables. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 012338.

(737) Bonet-Monroig, X.; Sagastizabal, R.; Singh, M.; O'Brien, T. E. Low-Cost Error Mitigation by Symmetry Verification. *Phys. Rev. A: At., Mol., Opt. Phys.* **2018**, *98*, No. 062339.

(738) Urbanek, M.; Camps, D.; Van Beeumen, R.; de Jong, W. A. Chemistry on Quantum Computers with Virtual Quantum Subspace Expansion. *J. Chem. Theory Comput.* **2020**, *16*, 5425–5431.

(739) McClean, J. R.; Jiang, Z.; Rubin, N. C.; Babbush, R.; Neven, H. Decoding Quantum Errors with Subspace Expansions. *Nat. Commun.* **2020**, *11*, 636.

(740) Kitaev, A. Fault-Tolerant Quantum Computation by Anyons. *Ann. Phys. (Amsterdam, Neth.)* **2003**, 303, 2–30.

(741) Fowler, A. G.; Mariantoni, M.; Martinis, J. M.; Cleland, A. N. Surface codes: Towards Practical Large-Scale Quantum Computation. *Phys. Rev. A: At., Mol., Opt. Phys.* **2012**, *86*, 032324.

(742) Bravyi, S.; Englbrecht, M.; König, R.; Peard, N. Correcting Coherent Errors with Surface Codes. *NPJ. Quantum Inf.* **2018**, *4*, 55.

(743) Sager, L. M.; Smart, S. E.; Mazziotti, D. A. Preparation of an Exciton Condensate of Photons on a 53-Qubit Quantum Computer. 2020; arXiv:2004.13868 [quant-ph]; https://arxiv.org/abs/2004. 13868.

(744) Fil, D. V.; Shevchenko, S. I. Electron-hole Superconductivity (Review). *Low Temp. Phys.* **2018**, *44*, 867–909.

(745) Huh, J.; Guerreschi, G. G.; Peropadre, B.; McClean, J. R.; Aspuru-Guzik, A. Boson Sampling for Molecular Vibronic Spectra. *Nat. Photonics* **2015**, *9*, 615–620.

(746) Sparrow, C.; Martín-López, E.; Maraviglia, N.; Neville, A.; Harrold, C.; Carolan, J.; Joglekar, Y. N.; Hashimoto, T.; Matsuda, N.; O'Brien, J. L.; et al. Simulating the Vibrational Quantum Dynamics of Molecules using Photonics. *Nature* **2018**, *557*, 660–667.

(747) Wang, C. S.; Curtis, J. C.; Lester, B. J.; Zhang, Y.; Gao, Y. Y.; Freeze, J.; Batista, V. S.; Vaccaro, P. H.; Chuang, I. L.; Frunzio, L.; et al. Efficient Multiphoton Sampling of Molecular Vibronic Spectra on a Superconducting Bosonic Processor. Phys. Rev. X 2020, 10, No. 021060.

(748) Siraichi, M. Y.; Santos, V. F. d.; Collange, S.; Pereira, F. M. Q. Qubit Allocation. *Proceedings of the 2018 International Symposium on Code Generation and Optimization*. New York, NY, USA, 2018; p 113–125.

(749) Murali, P.; Linke, N. M.; Martonosi, M.; Abhari, A. J.; Nguyen, N. H.; Alderete, C. H. Full-Stack, Real-System Quantum Computer Studies: Architectural Comparisons and Design Insights. *Proceedings of the 46th International Symposium on Computer Architecture*. New York, NY, USA, 2019; p 527–540.

(750) Finigan, W.; Cubeddu, M.; Lively, T.; Flick, J.; Narang, P. Qubit Allocation for Noisy Intermediate-Scale Quantum Computers. 2018; arXiv:1810.08291 [quant-ph]; https://arxiv.org/abs/1810.08291.

(751) Cubeddu, M.; Finigan, W.; Lively, T.; Flick, J.; Narang, P. Introducing Control Flow in Qubit Allocation for Quantum Turing Machines. 2019; arXiv:1907.07113 [quant-ph]; https://arxiv.org/abs/1907.07113.

(752) Litteken, A.; Fan, Y.-C.; Singh, D.; Martonosi, M. R.; Chong, F. An Updated LLVM-based Quantum Research Compiler with Further OpenQASM Support. *Quantum Sci. Technol.* **2020**, *5*, No. 034013.

(753) Tannu, S. S.; Qureshi, M. K. Not All Qubits Are Created Equal: A Case for Variability-Aware Policies for NISQ-Era Quantum Computers. Proceedings of the Twenty-Fourth International Conference on Architectural Support for Programming Languages and Operating Systems. New York, NY, USA, 2019; pp 987–999.

(754) Breuer, H. P.; Petruccione, F.The Theory of Open Quantum Systems; Oxford University Press, 2002.

(755) Weiss, U. *Quantum Dissipative Systems*; World Scientific, 2012. (756) Fassioli, F.; Nazir, A.; Olaya-Castro, A. Quantum State Tuning of Energy Transfer in a Correlated Environment. *J. Phys. Chem. Lett.* **2010**, *1*, 2139–2143.

(757) Yeh, S.-H.; Zhu, J.; Kais, S. Population and Coherence Dynamics in Light Harvesting Complex II (LH2). *J. Chem. Phys.* **2012**, 137, No. 084110.

(758) Rivas, A.; Huelga, S. F.; Plenio, M. B. Entanglement and Non-Markovianity of Quantum Evolutions. *Phys. Rev. Lett.* **2010**, *105*, No. 050403.

(759) Breuer, H.-P. Foundations and Measures of Quantum Non-Markovianity. J. Phys. B: At., Mol. Opt. Phys. 2012, 45, 154001.

(760) Rivas, A.; Huelga, S. F.; Plenio, M. B. Quantum Non-Markovianity: Characterization, Quantification and Detection. *Rep. Prog. Phys.* **2014**, *77*, No. 094001.

(761) Breuer, H.-P.; Laine, E.-M.; Piilo, J.; Vacchini, B. Colloquium: Non-Markovian Dynamics in Open Quantum Systems. *Rev. Mod. Phys.* **2016**, *88*, No. 021002.

(762) Li, C.-F.; Guo, G.-C.; Piilo, J. Non-Markovian Quantum Dynamics: What Does it Mean? *EPL* **2019**, *127*, 50001.

(763) Wolf, M. M.; Eisert, J.; Cubitt, T. S.; Cirac, J. I. Assessing Non-Markovian Quantum Dynamics. *Phys. Rev. Lett.* **2008**, *101*, 150402.

(764) Lu, X.-M.; Wang, X.; Sun, C. P. Quantum Fisher Information Flow and Non-Markovian Processes of Open Systems. *Phys. Rev. A: At., Mol., Opt. Phys.* **2010**, *82*, No. 042103.

(765) Breuer, H.-P.; Laine, E.-M.; Piilo, J. Measure for the Degree of Non-Markovian Behavior of Quantum Processes in Open Systems. *Phys. Rev. Lett.* **2009**, *103*, 210401.

(766) Luo, S.; Fu, S.; Song, H. Quantifying Non-Markovianity via Correlations. *Phys. Rev. A: At., Mol., Opt. Phys.* **2012**, *86*, No. 044101.

(767) Lorenzo, S.; Plastina, F.; Paternostro, M. Geometrical Characterization of Non-Markovianity. *Phys. Rev. A: At., Mol., Opt. Phys.* **2013**, 88, No. 020102.

(768) Fanchini, F. F.; Karpat, G.; Çakmak, B.; Castelano, L. K.; Aguilar, G. H.; Farías, O. J.; Walborn, S. P.; Ribeiro, P. H. S.; de Oliveira, M. C. Non-Markovianity through Accessible Information. *Phys. Rev. Lett.* **2014**, *112*, 210402.

(769) Bylicka, B.; Chruściński, D.; Maniscalco, S. Non-Markovianity and Reservoir Memory of Quantum Channels: A Quantum Information Theory Perspective. *Sci. Rep.* **2015**, *4*, 5720. (770) Hou, S. C.; Yi, X. X.; Yu, S. X.; Oh, C. H. Alternative Non-Markovianity Measure by Divisibility of Dynamical Maps. *Phys. Rev. A: At., Mol., Opt. Phys.* **2011**, 83, No. 062115.

(771) Hall, M. J. W.; Cresser, J. D.; Li, L.; Andersson, E. Canonical Form of Master Equations and Characterization of Non-Markovianity. *Phys. Rev. A: At., Mol., Opt. Phys.* **2014**, *89*, No. 042120.

(772) Chruściński, D.; Macchiavello, C.; Maniscalco, S. Detecting Non-Markovianity of Quantum Evolution via Spectra of Dynamical Maps. *Phys. Rev. Lett.* **201**7, *118*, No. 080404.

(773) Amato, G.; Breuer, H.-P.; Vacchini, B. Generalized Trace Distance Approach to Quantum Non-Markovianity and Detection of Initial Correlations. *Phys. Rev. A: At., Mol., Opt. Phys.* **2018**, *98*, No. 012120.

(774) Jaynes, E. T.; Cummings, F. W. Comparison of Quantum and Semiclassical Radiation Theories with Application to the Beam Maser. *Proc. IEEE* **1963**, *51*, 89–109.

(775) Mirkin, N.; Poggi, P.; Wisniacki, D. Information Backflow as a Resource for Entanglement. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 062327.

(776) Fan, Y.; Chen, L.; Cao, H.; Meng, H. Quantum Non-Markovianity via the Covariance Matrix. *Int. J. Theor. Phys.* **2018**, *57*, 2722–2737.

(777) He, Z.; Zeng, H.-S.; Li, Y.; Wang, Q.; Yao, C. Non-Markovianity Measure Based on the Relative Entropy of Coherence in an Extended Space. *Phys. Rev. A: At., Mol., Opt. Phys.* **2017**, *96*, No. 022106.

(778) Wißmann, S.; Karlsson, A.; Laine, E.-M.; Piilo, J.; Breuer, H.-P. Optimal State Pairs for Non-Markovian Quantum Dynamics. *Phys. Rev. A: At., Mol., Opt. Phys.* **2012**, *86*, No. 062108.

(779) Pineda, C.; Gorin, T.; Davalos, D.; Wisniacki, D. A.; García-Mata, I. Measuring and Using Non-Markovianity. *Phys. Rev. A: At., Mol., Opt. Phys.* **2016**, 93, No. 022117.

(780) Bennink, R. S.; Lougovski, P. Quantum Process Identification: A Method for Characterizing Non-Markovian Quantum Dynamics. *New J. Phys.* **2019**, *21*, No. 083013.

(781) Mirkin, N.; Poggi, P.; Wisniacki, D. Entangling Protocols due to Non-Markovian Dynamics. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 020301.

(782) de Vega, I.; Alonso, D. Dynamics of Non-Markovian Open Quantum Systems. *Rev. Mod. Phys.* 2017, *89*, No. 015001.

(783) Berkelbach, T. C.; Thoss, M. Special Topic on Dynamics of Open Quantum Systems. J. Chem. Phys. 2020, 152, No. 020401.

(784) Lindblad, G. On the Generators of Quantum Dynamical Semigroups. *Commun. Math. Phys.* **1976**, *48*, 119–130.

(785) Gorini, V.; Kossakowski, A.; Sudarshan, E. C. G. Complete Positive Dynamical Semigroups of *N*-Level Systems. *J. Math. Phys.* **1976**, 17 (5), 821–825.

(786) Nguyen, T. S.; Nanguneri, R.; Parkhill, J. How Electronic Dynamics with Pauli Exclusion Produces Fermi–Dirac Statistics. *J. Chem. Phys.* **2015**, *142*, 134113.

(787) Head-Marsden, K.; Mazziotti, D. A. Communication: Satisfying Fermionic Statistics in the Modeling of Open Time-Dependent Quantum Systems with One-Electron Reduced Density Matrices. *J. Chem. Phys.* **2015**, *142*, No. 051102.

(788) Head-Marsden, K.; Mazziotti, D. A. Ensemble of Lindblad's Trajectories for Non-Markovian Dynamics. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *99*, No. 022109.

(789) Head-Marsden, K.; Mazziotti, D. A. Satisfying Fermionic Statistics in the Modeling of Non-Markovian Dynamics with One-Electron Reduced Density Matrices. *J. Chem. Phys.* 2019, 151, No. 034111.

(790) Müller, C.; Stace, T. M. Deriving Lindblad Master Equations with Keldysh Diagrams: Correlated Gain and Loss in Higher Order Perturbation Theory. *Phys. Rev. A: At., Mol., Opt. Phys.* **201**7, *95*, No. 013847.

(791) Redfield, A. In *Advances in Magnetic Resonance*; Waugh, J. S., Ed.; Advances in Magnetic and Optical Resonance; Academic Press, 1965; Vol. 1; pp 1–32.

(792) Kondov, I.; Kleinekathöfer, U.; Schreiber, M. Efficiency of Different Numerical Methods for Solving Redfield Equations. *J. Chem. Phys.* **2001**, *114*, 1497–1504.

(793) Farina, D.; Giovannetti, V. Open-Quantum-System Dynamics: Recovering Positivity of the Redfield Equation via the Partial Secular Approximation. *Phys. Rev. A: At., Mol., Opt. Phys.* **2019**, *100*, No. 012107.

(794) Schröder, M.; Schreiber, M.; Kleinekathöfer, U. A Time-Dependent Modified Redfield Theory for Absorption Spectra Applied to Light-Harvesting Systems. *J. Lumin.* **200**7, *125*, 126–132.

(795) Ishizaki, A.; Fleming, G. R. On the Adequacy of the Redfield Equation and Related Approaches to the Study of Quantum Dynamics in Electronic Energy Transfer. *J. Chem. Phys.* **2009**, *130*, 234110.

(796) Jeske, J.; Ing, D. J.; Plenio, M. B.; Huelga, S. F.; Cole, J. H. Bloch-Redfield Equations for Modeling Light-Harvesting Complexes. *J. Chem. Phys.* **2015**, *142*, No. 064104.

(797) Rivas, A.; Huelga, S.Open Quantum Systems: An Introduction; Springer-Verlag, Berlin, 2012.

(798) Barnett, S. M.; Stenholm, S. Hazards of Reservoir Memory. *Phys. Rev. A: At., Mol., Opt. Phys.* 2001, 64, No. 033808.

(799) Ai, Q.; Yen, T.-C.; Jin, B.-Y.; Cheng, Y.-C. Clustered Geometries Exploiting Quantum Coherence Effects for Efficient Energy Transfer in Light Harvesting. *J. Phys. Chem. Lett.* **2013**, *4*, 2577–2584.

(800) Montoya-Castillo, A.; Berkelbach, T. C.; Reichman, D. R. Extending the Applicability of Redfield Theories into Highly Non-Markovian Regimes. *J. Chem. Phys.* **2015**, *143*, 194108.

(801) Schiro, M.; Scarlatella, O. Quantum Impurity Models Coupled to Markovian and Non-Markovian Baths. J. Chem. Phys. 2019, 151, No. 044102.

(802) McConnell, C.; Nazir, A. Electron Counting Statistics for Non-Additive Environments. J. Chem. Phys. **2019**, 151, No. 054104.

(803) Rudge, S. L.; Kosov, D. S. Counting Quantum Jumps: A Summary and Comparison of Fixed-Time and Fluctuating-Time Statistics in Electron Transport. *J. Chem. Phys.* **2019**, *151*, No. 034107.

(804) Gu, B.; Franco, I. When Can Quantum Decoherence be Mimicked by Classical Noise? J. Chem. Phys. 2019, 151, No. 014109.

(805) Schile, A. J.; Limmer, D. T. Simulating Conical Intersection Dynamics in the Condensed Phase with Hybrid Quantum Master Equations. J. Chem. Phys. **2019**, 151, No. 014106.

(806) Jang, S. J. Fourth Order Expressions for the Electronic Absorption Lineshape of Molecular Excitons. *J. Chem. Phys.* **2019**, *151*, No. 044110.

(807) Shabani, A.; Lidar, D. A. Completely Positive Post-Markovian Master Equation via a Measurement Approach. *Phys. Rev. A: At., Mol., Opt. Phys.* **2005**, *71*, No. 020101.

(808) Maniscalco, S.; Petruccione, F. Non-Markovian Dynamics of a Qubit. *Phys. Rev. A: At., Mol., Opt. Phys.* **2006**, *73*, No. 012111.

(809) Budini, A. A. Post-Markovian Quantum Master Equations from Classical Environment Fluctuations. *Phys. Rev. E* **2014**, *89*, No. 012147.

(810) Chruściński, D.; Kossakowski, A. Generalized Semi-Markov Quantum Evolution. *Phys. Rev. A: At., Mol., Opt. Phys.* 2017, 95, No. 042131.

(811) Shibata, F.; Takahashi, Y.; Hashitsume, N. A Generalized Stochastic Liouville Equation. Non-Markovian versus Memoryless Master Equations. *J. Stat. Phys.* **1977**, *17*, 171–187.

(812) Nakajima, S. On Quantum Theory of Transport Phenomena: Steady Diffusion. *Prog. Theor. Phys.* **1958**, *20*, 948–959.

(813) Zwanzig, R. Ensemble Method in the Theory of Irreversibility. *J. Chem. Phys.* **1960**, *33*, 1338–1341.

(814) Vacchini, B. Generalized Master Equations Leading to Completely Positive Dynamics. *Phys. Rev. Lett.* **2016**, *117*, 230401.

(815) Xu, M.; Yan, Y.; Liu, Y.; Shi, Q. Convergence of High Order Memory Kernels in the Nakajima-Zwanzig Generalized Master Equation and Rate Constants: Case Study of the Spin-Boson Model. *J. Chem. Phys.* **2018**, *148*, 164101.

(816) Ivanov, A.; Breuer, H.-P. Extension of the Nakajima-Zwanzig Approach to Multitime Correlation Functions of Open Systems. *Phys. Rev. A: At., Mol., Opt. Phys.* **2015**, *92*, No. 032113. (817) Shi, Q.; Geva, E. A New Approach to Calculating the Memory Kernel of the Generalized Quantum Master Equation for an Arbitrary System-Bath Coupling. *J. Chem. Phys.* **2003**, *119*, 12063–12076.

(818) Zhang, M.-L.; Ka, B. J.; Geva, E. Nonequilibrium Quantum Dynamics in the Condensed Phase via the Generalized Quantum Master Equation. J. Chem. Phys. **2006**, 125, No. 044106.

(819) Cerrillo, J.; Cao, J. Non-Markovian Dynamical Maps: Numerical Processing of Open Quantum Trajectories. *Phys. Rev. Lett.* **2014**, *112*, 110401.

(820) Mulvihill, E.; Schubert, A.; Sun, X.; Dunietz, B. D.; Geva, E. A Modified Approach for Simulating Electronically Nonadiabatic Dynamics via the Generalized Quantum Master Equation. *J. Chem. Phys.* **2019**, *150*, No. 034101.

(821) Mulvihill, E.; Gao, X.; Liu, Y.; Schubert, A.; Dunietz, B. D.; Geva, E. Combining the Mapping Hamiltonian Linearized Semiclassical Approach with the Generalized Quantum Master Equation to Simulate Electronically Nonadiabatic Molecular Dynamics. *J. Chem. Phys.* **2019**, *151*, No. 074103.

(822) Pfalzgraff, W. C.; Montoya-Castillo, A.; Kelly, A.; Markland, T. E. Efficient Construction of Generalized Master Equation Memory Kernels for Multi-State Systems from Nonadiabatic Quantum-Classical Dynamics. *J. Chem. Phys.* **2019**, *150*, 244109.

(823) Yan, Y.; Xu, M.; Liu, Y.; Shi, Q. Theoretical Study of Charge Carrier Transport in Organic Molecular Crystals using the Nakajima-Zwanzig-Mori Generalized Master Equation. *J. Chem. Phys.* **2019**, *150*, 234101.

(824) Kelly, A.; Markland, T. E. Efficient and Accurate Surface Hopping for Long Time Nonadiabatic Quantum Dynamics. *J. Chem. Phys.* **2013**, *139*, No. 014104.

(825) Montoya-Castillo, A.; Reichman, D. R. Approximate but Accurate Quantum Dynamics from the Mori Formalism: I. Nonequilibrium Dynamics. *J. Chem. Phys.* **2016**, *144*, 184104.

(826) Montoya-Castillo, A.; Reichman, D. R. Approximate but Accurate Quantum Dynamics from the Mori Formalism. II. Equilibrium Time Correlation Functions. J. Chem. Phys. 2017, 146, No. 084110.

(827) Kidon, L.; Wang, H.; Thoss, M.; Rabani, E. On the Memory Kernel and the Reduced System Propagator. *J. Chem. Phys.* **2018**, *149*, 104105.

(828) Breuer, H.-P.; Kappler, B.; Petruccione, F. The Time-Convolutionless Projection Operator Technique in the Quantum Theory of Dissipation and Decoherence. *Ann. Phys. (Amsterdam, Neth.)* **2001**, *291*, 36–70.

(829) Smirne, A.; Vacchini, B. Nakajima-Zwanzig versus Time-Convolutionless Master Equation for the Non-Markovian Dynamics of a Two-Level System. *Phys. Rev. A: At., Mol., Opt. Phys.* **2010**, *82*, No. 022110.

(830) Hartmann, R.; Strunz, W. T. Accuracy Assessment of Perturbative Master Equations: Embracing Nonpositivity. *Phys. Rev.* A: At., Mol., Opt. Phys. **2020**, 101, No. 012103.

(831) Fetherolf, J. H.; Berkelbach, T. C. Linear and Nonlinear Spectroscopy from Quantum Master Equations. J. Chem. Phys. 2017, 147, 244109.

(832) Kidon, L.; Wilner, E. Y.; Rabani, E. Exact Calculation of the Time Convolutionless Master Equation Generator: Application to the Nonequilibrium Resonant Level Model. *J. Chem. Phys.* **2015**, *143*, 234110.

(833) Feynman, R.; Vernon, F. The Theory of a General Quantum System Interacting with a Linear Dissipative System. *Ann. Phys.* (*Amsterdam, Neth.*) **1963**, *24*, 118–173.

(834) Makri, N. Improved Feynman Propagators on a Grid and Non-Adiabatic Corrections within the Path Integral Framework. *Chem. Phys. Lett.* **1992**, *193*, 435–445.

(835) Makri, N.; Makarov, D. E. Tensor Propagator for Iterative Quantum Time Evolution of Reduced Density Matrices. I. Theory. *J. Chem. Phys.* **1995**, *102*, 4600–4610.

(836) Makri, N.; Makarov, D. E. Tensor Propagator for Iterative Quantum Time Evolution of Reduced Density Matrices. II. Numerical Methodology. *J. Chem. Phys.* **1995**, *102*, 4611–4618.

(837) Fosdick, L. D.; Jordan, H. F. Path-Integral Calculation of the Two-Particle Slater Sum for He⁴. *Phys. Rev.* **1966**, *143*, 58–66.

(838) Lawande, S.; Jensen, C.; Sahlin, H. Monte Carlo Integration of the Feynman Propagator in Imaginary Time. *J. Comput. Phys.* **1969**, *3*, 416–443.

(839) Caldeira, A.; Leggett, A. Path Integral Approach to Quantum Brownian Motion. *Phys. A* **1983**, *121*, 587–616.

(840) Dekker, H. Noninteracting-Blip Approximation for a Two-Level System Coupled to a Heat Bath. *Phys. Rev. A: At., Mol., Opt. Phys.* **1987**, *35*, 1436–1437.

(841) Walters, P. L.; Banerjee, T.; Makri, N. On Iterative Path Integral Calculations for a System Interacting with a Shifted Dissipative Bath. *J. Chem. Phys.* **2015**, *143*, No. 074112.

(842) Makri, N. Iterative Blip-Summed Path Integral for Quantum Dynamics in Strongly Dissipative Environments. *J. Chem. Phys.* 2017, 146, 134101.

(843) Thorwart, M.; Eckel, J.; Mucciolo, E. R. Non-Markovian Dynamics of Double Quantum Dot Charge Qubits due to Acoustic Phonons. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2005**, *72*, 235320.

(844) Eckel, J.; Weiss, S.; Thorwart, M. Phonon-Induced Decoherence and Dissipation in Donor-Based Charge Qubits. *Eur. Phys. J. B* **2006**, 53, 91–98.

(845) Strathearn, A.; Lovett, B. W.; Kirton, P. Efficient Real-Time Path Integrals for Non-Markovian Spin-Boson Models. *New J. Phys.* **201**7, *19*, No. 093009.

(846) Makri, N. Communication: Modular Path Integral: Quantum Dynamics via Sequential Necklace Linking. *J. Chem. Phys.* **2018**, *148*, 101101.

(847) Tanimura, Y.; Kubo, R. Time Evolution of a Quantum System in Contact with a Nearly Gaussian-Markoffian Noise Bath. *J. Phys. Soc. Jpn.* **1989**, *58*, 101–114.

(848) Ishizaki, A.; Tanimura, Y. Quantum Dynamics of System Strongly Coupled to Low-Temperature Colored Noise Bath: Reduced Hierarchy Equations Approach. J. Phys. Soc. Jpn. **2005**, 74, 3131–3134.

(849) Chen, L.; Zheng, R.; Shi, Q.; Yan, Y. Optical Line Shapes of Molecular Aggregates: Hierarchical Equations of Motion Method. *J. Chem. Phys.* **2009**, *131*, No. 094502.

(850) Tanimura, Y. Reduced Hierarchical Equations of Motion in Real and Imaginary Time: Correlated Initial States and Thermodynamic Quantities. J. Chem. Phys. 2014, 141, No. 044114.

(851) Schinabeck, C.; Erpenbeck, A.; Härtle, R.; Thoss, M. Hierarchical Quantum Master Equation Approach to Electronic-Vibrational Coupling in Nonequilibrium Transport through Nanosystems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2016**, *94*, 201407.

(852) Iwamoto, Y.; Tanimura, Y. Open Quantum Dynamics of a Three-Dimensional Rotor Calculated using a Rotationally Invariant System-Bath Hamiltonian: Linear and Two-Dimensional Rotational Spectra. J. Chem. Phys. **2019**, 151, No. 044105.

(853) Cui, L.; Zhang, H.-D.; Zheng, X.; Xu, R.-X.; Yan, Y. Highly Efficient and Accurate Sum-Over-Poles Expansion of Fermi and Bose Functions at Near Zero Temperatures: Fano Spectrum Decomposition Scheme. *J. Chem. Phys.* **2019**, *151*, No. 024110.

(854) Dunn, I. S.; Tempelaar, R.; Reichman, D. R. Removing Instabilities in the Hierarchical Equations of Motion: Exact and Approximate Projection Approaches. J. Chem. Phys. **2019**, 150, 184109.

(855) Rahman, H.; Kleinekathöfer, U. Chebyshev Hierarchical Equations of Motion for Systems with Arbitrary Spectral Densities and Temperatures. J. Chem. Phys. **2019**, 150, 244104.

(856) Chen, L.; Gelin, M. F.; Domcke, W. Orientational Relaxation of a Quantum Linear Rotor in a Dissipative Environment: Simulations with the Hierarchical Equations-of-Motion Method. *J. Chem. Phys.* **2019**, *151*, No. 034101.

(857) Borrelli, R. Density Matrix Dynamics in Twin-Formulation: An Efficient Methodology Based on Tensor-Train Representation of Reduced Equations of Motion. *J. Chem. Phys.* **2019**, *150*, 234102.

(858) Shi, Q.; Chen, L.; Nan, G.; Xu, R.-X.; Yan, Y. Efficient Hierarchical Liouville Space Propagator to Quantum Dissipative Dynamics. J. Chem. Phys. **2009**, 130, No. 084105. (859) Ishizaki, A.; Fleming, G. R. Unified Treatment of Quantum Coherent and Incoherent Hopping Dynamics in Electronic Energy Transfer: Reduced Hierarchy Equation Approach. *J. Chem. Phys.* **2009**, 130, 234111.

(860) Zhu, J.; Kais, S.; Rebentrost, P.; Aspuru-Guzik, A. Modified Scaled Hierarchical Equation of Motion Approach for the Study of Quantum Coherence in Photosynthetic Complexes. *J. Phys. Chem. B* **2011**, *115*, 1531–1537.

(861) Dijkstra, A. G.; Beige, A. Efficient Long-Distance Energy Transport in Molecular Systems through Adiabatic Passage. *J. Chem. Phys.* **2019**, *151*, No. 034114.

(862) Xu, M.; Song, L.; Song, K.; Shi, Q. Convergence of High Order Perturbative Expansions in Open System Quantum Dynamics. *J. Chem. Phys.* **2017**, *146*, No. 064102.

(863) Dalibard, J.; Castin, Y.; Mølmer, K. Wave-function Approach to Dissipative Processes in Quantum Optics. *Phys. Rev. Lett.* **1992**, *68*, 580–583.

(864) Dum, R.; Zoller, P.; Ritsch, H. Monte Carlo Simulation of the Atomic Master Equation for Spontaneous Emission. *Phys. Rev. A: At., Mol., Opt. Phys.* **1992**, *45*, 4879–4887.

(865) Mølmer, K.; Castin, Y.; Dalibard, J. Monte Carlo Wave-Function Method in Quantum Optics. J. Opt. Soc. Am. B **1993**, *10*, 524–538.

(866) Plenio, M. B.; Knight, P. L. The Quantum-Jump Approach to Dissipative Dynamics in Quantum Optics. *Rev. Mod. Phys.* **1998**, *70*, 101–144.

(867) Piilo, J.; Maniscalco, S.; Härkönen, K.; Suominen, K.-A. Non-Markovian Quantum Jumps. *Phys. Rev. Lett.* **2008**, *100*, 180402.

(868) Piilo, J.; Härkönen, K.; Maniscalco, S.; Suominen, K.-A. Open System Dynamics with Non-Markovian Quantum Jumps. *Phys. Rev. A: At., Mol., Opt. Phys.* **2009**, *79*, No. 062112.

(869) Breuer, H.-P.; Piilo, J. Stochastic Jump Processes for Non-Markovian Quantum Dynamics. *EPL* **2009**, *85*, 50004.

(870) Rebentrost, P.; Chakraborty, R.; Aspuru-Guzik, A. Non-Markovian Quantum Jumps in Excitonic Energy Transfer. *J. Chem. Phys.* **2009**, *131*, 184102.

(871) Ai, Q.; Fan, Y.-J.; Jin, B.-Y.; Cheng, Y.-C. An Efficient Quantum Jump Method for Coherent Energy Transfer Dynamics in Photosynthetic Systems under the Influence of Laser Fields. *New J. Phys.* **2014**, *16*, No. 053033.

(872) Yang, W.; Liu, R.-B. Quantum Many-Body Theory of Qubit Decoherence in a Finite-Size Spin Bath. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2008**, *78*, No. 085315.

(873) Witzel, W. M.; Das Sarma, S. Quantum Theory for Electron Spin Decoherence Induced by Nuclear Spin Dynamics in Semiconductor Quantum Computer Architectures: Spectral Diffusion of Localized Electron Spins in the Nuclear Solid-State Environment. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2006**, *74*, No. 035322.

(874) Witzel, W. M.; Sarma, S. D. Multiple-Pulse Coherence Enhancement of Solid State Spin Qubits. *Phys. Rev. Lett.* **2007**, *98*, No. 077601.

(875) Witzel, W. M.; Das Sarma, S. Concatenated Dynamical Decoupling in a Solid-State Spin Bath. *Phys. Rev. B: Condens. Matter Mater. Phys.* 2007, 76, 241303.

(876) Saikin, S. K.; Yao, W.; Sham, L. J. Single-Electron Spin Decoherence by Nuclear Spin Bath: Linked-Cluster Expansion Approach. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2007**, *75*, 125314.

(877) Yao, W.; Liu, R.-B.; Sham, L. J. Theory of Electron Spin Decoherence by Interacting Nuclear Spins in a Quantum Dot. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2006**, *74*, 195301.

(878) Yao, W.; Liu, R.-B.; Sham, L. J. Restoring Coherence Lost to a Slow Interacting Mesoscopic Spin Bath. *Phys. Rev. Lett.* **2007**, *98*, No. 077602.

(879) Liu, R.-B.; Yao, W.; Sham, L. J. Control of Electron Spin Decoherence Caused by Electron–Nuclear Spin Dynamics in a Quantum Dot. *New J. Phys.* **2007**, *9*, 226–226.

(880) Yang, W.; Liu, R.-B. Quantum Many-Body Theory of Qubit Decoherence in a Finite-Size Spin Bath. II. Ensemble Dynamics. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2009**, *79*, 115320.

(881) Yang, Z.-S.; Wang, Y.-X.; Tao, M.-J.; Yang, W.; Zhang, M.; Ai, Q.; Deng, F.-G. Longitudinal Relaxation of a Nitrogen-Vacancy Center in a Spin Bath by Generalized Cluster-Correlation Expansion Method. *Ann. Phys. (Amsterdam, Neth.)* **2020**, *413*, 168063.

(882) Jørgensen, M. R.; Pollock, F. A. Exploiting the Causal Tensor Network Structure of Quantum Processes to Efficiently Simulate Non-Markovian Path Integrals. *Phys. Rev. Lett.* **2019**, *123*, 240602.

(883) Luchnikov, I. A.; Vintskevich, S. V.; Ouerdane, H.; Filippov, S. N. Simulation Complexity of Open Quantum Dynamics: Connection with Tensor Networks. *Phys. Rev. Lett.* **2019**, *122*, 160401.

(884) Smolensky, P. In *Parallel Distributed Processing: Explorations in the Microstructure of Cognition*; Rumelhart, D. E., McClelland, J. L., Eds.; MIT Press, 1986; pp 194–281.

(885) Melko, R. G.; Carleo, G.; Carrasquilla, J.; Cirac, J. I. Restricted Boltzmann Machines in Quantum Physics. *Nat. Phys.* **2019**, *15*, 887– 892.

(886) Yoshioka, N.; Hamazaki, R. Constructing Neural Stationary States for Open Quantum Many-Body Systems. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2019**, *99*, 214306.

(887) Nagy, A.; Savona, V. Variational Quantum Monte Carlo Method with a Neural-Network Ansatz for Open Quantum Systems. *Phys. Rev. Lett.* **2019**, *122*, 250501.

(888) Hartmann, R.; Werther, M.; Grossmann, F.; Strunz, W. T. Exact Open Quantum System Dynamics: Optimal Frequency vs Time Representation of Bath Correlations. *J. Chem. Phys.* **2019**, *150*, 234105.

(889) Vicentini, F.; Biella, A.; Regnault, N.; Ciuti, C. Variational Neural-Network Ansatz for Steady States in Open Quantum Systems. *Phys. Rev. Lett.* **2019**, *122*, 250503.

(890) Liu, H.-Y.; Wu, Y.-C.; Guo, G.-C.; Guo, G.-P.Construct Stationary States Ansatz of Open Quantum Systems with Parameterized Quantum Circuits. arXiv preprint arXiv:2001.02552 2020, arXiv:2001.02552 [quant-ph]; https://arxiv.org/abs/2001.02552.

(891) Wang, H.; Ashhab, S.; Nori, F. Quantum Algorithm for Simulating the Dynamics of an Open Quantum System. *Phys. Rev. A: At., Mol., Opt. Phys.* **2011**, 83, No. 062317.

(892) Sweke, R.; Sinayskiy, I.; Bernard, D.; Petruccione, F. Universal Simulation of Markovian Open Quantum Systems. *Phys. Rev. A: At., Mol., Opt. Phys.* **2015**, *91*, No. 062308.

(893) Wei, S.-J.; Ruan, D.; Long, G.-L. Duality Quantum Algorithm Efficiently Simulates Open Quantum Systems. *Sci. Rep.* **2016**, *6*, 30727.

(894) García-Pérez, G.; Rossi, M. A. C.; Maniscalco, S. IBM Q Experience as a Versatile Experimental Testbed for Simulating Open Quantum Systems. *NPJ Quantum Inf.* **2020**, *6*, 1.

(895) Stinespring, W. F. Positive Functions on C*-Algebras. *Proc. Am. Math. Soc.* **1955**, *6*, 211–216.

(896) Langer, H. B. Sz.-Nagy and C. Foias, Harmonic Analysis of Operators on Hilbert Space. VIII + 387 S. Budapest/Amsterdam/ London 1970. Akadémiai Kiadó/North-Holland Publishing Company. *Z. Angew. Math. Mech.* **1972**, *52*, 501–501.

(897) Paulsen, V. Completely Bounded Maps and Operator Algebras; Cambridge University Press, 2002.

(898) Levy, E.; Shalit, O. M. Dilation Theory in Finite Dimensions: The Possible, the Impossible and the Unknown. *Rocky Mt. J. Math.* **2014**, 44, 203–221.

(899) Hu, Z.; Xia, R.; Kais, S. A Quantum Algorithm for Evolving Open Quantum Dynamics on Quantum Computing Devices. *Sci. Rep.* **2020**, *10*, 3301.

(900) Wang, D.-S.; Berry, D. W.; de Oliveira, M. C.; Sanders, B. C. Solovay-Kitaev Decomposition Strategy for Single-Qubit Channels. *Phys. Rev. Lett.* **2013**, *111*, 130504.

(901) Head-Marsden, K.; Krastanov, S.; Mazziotti, D. A.; Narang, P. Capturing Non-Markovian Dynamics on Near-Term Quantum Computers. 2020; arXiv:2005.00029 [quant-ph]; https://arxiv.org/abs/2005.00029.

(902) Prior, J.; Chin, A. W.; Huelga, S. F.; Plenio, M. B. Efficient Simulation of Strong System-Environment Interactions. *Phys. Rev. Lett.* **2010**, *105*, No. 050404. (903) Berkelbach, T. C.; Markland, T. E.; Reichman, D. R. Reduced Density Matrix Hybrid Approach: Application to Electronic Energy Transfer. J. Chem. Phys. **2012**, 136, No. 084104.

(904) Wang, Z.-H.; Takahashi, S. Spin Decoherence and Electron Spin Bath Noise of a Nitrogen-Vacancy Center in Diamond. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2013**, *87*, 115122.

(905) Haase, J. F.; Vetter, P. J.; Unden, T.; Smirne, A.; Rosskopf, J.; Naydenov, B.; Stacey, A.; Jelezko, F.; Plenio, M. B.; Huelga, S. F. Controllable Non-Markovianity for a Spin Qubit in Diamond. *Phys. Rev. Lett.* **2018**, *121*, No. 060401.

(906) Peng, S.; Xu, X.; Xu, K.; Huang, P.; Wang, P.; Kong, X.; Rong, X.; Shi, F.; Duan, C.; Du, J. Observation of non-Markovianity at Room Temperature by Prolonging Entanglement in Solids. *Sci. Bull.* **2018**, *63*, 336–339.

(907) Wang, F.; Hou, P.-Y.; Huang, Y.-Y.; Zhang, W.-G.; Ouyang, X.-L.; Wang, X.; Huang, X.-Z.; Zhang, H.-L.; He, L.; Chang, X.-Y.; et al. Observation of Entanglement Sudden Death and Rebirth by Controlling a Solid-State Spin Bath. *Phys. Rev. B: Condens. Matter Mater. Phys.* **2018**, *98*, No. 064306.

(908) Agraït, N.; Yeyati, A. L.; van Ruitenbeek, J. M. Quantum Properties of Atomic-Sized Conductors. *Phys. Rep.* **2003**, 377 (2), 81–279.

(909) Komoto, Y.; Fujii, S.; Iwane, M.; Kiguchi, M. Single-Molecule Junctions for Molecular Electronics. *J. Mater. Chem. C* **2016**, *4*, 8842–8858.

(910) Kelly, A. Mean Field Theory of Thermal Energy Transport in Molecular Junctions. J. Chem. Phys. **2019**, *150*, 204107.

(911) Peskin, U. Quantum Transport in the Presence of Contact Vibrations. J. Chem. Phys. 2019, 151, No. 024108.

(912) Ridley, M.; Gull, E.; Cohen, G. Lead Geometry and Transport Statistics in Molecular Junctions. *J. Chem. Phys.* **2019**, *150*, 244107.

(913) Teh, H.-H.; Jin, B.-Y.; Cheng, Y.-C. Frozen-Mode Small Polaron Quantum Master Equation with Variational Bound for Excitation Energy Transfer in Molecular Aggregates. *J. Chem. Phys.* **2019**, *150*, 224110.

(914) Ishizaki, A.; Fleming, G. R. Theoretical Examination of Quantum Coherence in a Photosynthetic System at Physiological Temperature. *Proc. Natl. Acad. Sci. U. S. A.* **2009**, *106*, 17255–17260.

(915) Mohseni, M.; Rebentrost, P.; Lloyd, S.; Aspuru-Guzik, A. Environment-Assisted Quantum Walks in Photosynthetic Energy Transfer. J. Chem. Phys. 2008, 129, 174106.

(916) Palmieri, B.; Abramavicius, D.; Mukamel, S. Lindblad Equations for Strongly Coupled Populations and Coherences in Photosynthetic Complexes. J. Chem. Phys. **2009**, 130, 204512.

(917) Mohseni, M.; Shabani, A.; Lloyd, S.; Rabitz, H. Energy-Scales Convergence for Optimal and Robust Quantum Transport in Photosynthetic Complexes. J. Chem. Phys. **2014**, 140, No. 035102.

(918) Alicki, R. A Quantum Open System Model of Molecular Battery Charged by Excitons. J. Chem. Phys. **2019**, 150, 214110.

(919) Engel, G. S.; Calhoun, T. R.; Read, E. L.; Ahn, T.-K.; Mančal, T.; Cheng, Y.-C.; Blankenship, R. E.; Fleming, G. R. Evidence for Wavelike Energy Transfer through Quantum Coherence in Photosynthetic Systems. *Nature* **2007**, *446*, 782–786.

(920) Panitchayangkoon, G.; Hayes, D.; Fransted, K. A.; Caram, J. R.; Harel, E.; Wen, J.; Blankenship, R. E.; Engel, G. S. Long-Lived Quantum Coherence in Photosynthetic Complexes at Physiological Temperature. *Proc. Natl. Acad. Sci. U. S. A.* **2010**, *107*, 12766–12770.

(921) Rebentrost, P.; Serban, I.; Schulte-Herbrüggen, T.; Wilhelm, F. K. Optimal Control of a Qubit Coupled to a Non-Markovian Environment. *Phys. Rev. Lett.* **2009**, *102*, No. 090401.

(922) Pachón, L. A.; Brumer, P. Mechanisms in Environmentally Assisted One-Photon Phase Control. *J. Chem. Phys.* **2013**, *139*, 164123. (923) Reich, D. M.; Katz, N.; Koch, C. P. Exploiting Non-Markovianity for Quantum Control. *Sci. Rep.* **2015**, *5*, 12430.

(924) Poggi, P. M.; Lombardo, F. C.; Wisniacki, D. A. Driving-Induced Amplification of Non-Markovianity in Open Quantum Systems Evolution. *EPL* **2017**, *118*, 20005. (925) Ho, L. B.; Matsuzaki, Y.; Matsuzaki, M.; Kondo, Y. Realization of Controllable Open System with NMR. *New J. Phys.* **2019**, *21*, No. 093008.

(926) Liu, B.-H.; Hu, X.-M.; Huang, Y.-F.; Li, C.-F.; Guo, G.-C.; Karlsson, A.; Laine, E.-M.; Maniscalco, S.; Macchiavello, C.; Piilo, J. Efficient Superdense Coding in the Presence of non-Markovian Noise. *EPL* **2016**, *114*, 10005.

(927) Laine, E.-M.; Breuer, H.-P.; Piilo, J. Nonlocal Memory Effects Allow Perfect Teleportation with Mixed States. *Sci. Rep.* **2014**, *4*, 4620.

(928) Li, C.-F.; Guo, G.-C.; Piilo, J. Non-Markovian Quantum Dynamics: What is it Good For? *EPL* **2019**, *128*, 30001.

(929) Haase, J. F.; Smirne, A.; Huelga, S. F. Non-Monotonic Population and Coherence Evolution in Markovian Open-System Dynamics. *Advances in Open Systems and Fundamental Tests of Quantum Mechanics*; Springer: Cham, 2019; pp 41–57.

(930) Siudzińska, K.; Chruściński, D. Engineering Fidelity of the Generalized Pauli Channels via Legitimate Memory Kernels. *Phys. Rev.* A: At., Mol., Opt. Phys. **2019**, 100, No. 012303.

(931) Krastanov, S.; Zhou, S.; Flammia, S. T.; Jiang, L. Stochastic Estimation of Dynamical Variables. *Quantum Sci. Technol.* **2019**, *4*, No. 035003.

(932) Flurin, E.; Martin, L. S.; Hacohen-Gourgy, S.; Siddiqi, I. Using a Recurrent Neural Network to Reconstruct Quantum Dynamics of a Superconducting Qubit from Physical Observations. *Phys. Rev. X* 2020, *10*, No. 011006.

(933) Duan, L.-M.; Guo, G.-C. Preserving Coherence in Quantum Computation by Pairing Quantum Bits. *Phys. Rev. Lett.* **1997**, *79*, 1953–1956.

(934) Zanardi, P.; Rasetti, M. Noiseless Quantum Codes. *Phys. Rev. Lett.* **1997**, *79*, 3306–3309.

(935) Lidar, D. A.; Chuang, I. L.; Whaley, K. B. Decoherence-Free Subspaces for Quantum Computation. *Phys. Rev. Lett.* **1998**, *81*, 2594–2597.

(936) Zanardi, P. Stabilizing Quantum Information. *Phys. Rev. A: At., Mol., Opt. Phys.* **2000**, *63*, No. 012301.

(937) Mirrahimi, M.; Leghtas, Z.; Albert, V. V.; Touzard, S.; Schoelkopf, R. J.; Jiang, L.; Devoret, M. H. Dynamically Protected Cat-Qubits: A New Paradigm for Universal Quantum Computation. *New J. Phys.* **2014**, *16*, No. 045014.

(938) Zanardi, P.; Campos Venuti, L. Geometry, Robustness, and Emerging Unitarity in Dissipation-Projected Dynamics. *Phys. Rev. A: At., Mol., Opt. Phys.* **2015**, *91*, No. 052324.

(939) Albert, V. V.; Shu, C.; Krastanov, S.; Shen, C.; Liu, R.-B.; Yang, Z.-B.; Schoelkopf, R. J.; Mirrahimi, M.; Devoret, M. H.; Jiang, L. Holonomic Quantum Control with Continuous Variable Systems. *Phys. Rev. Lett.* **2016**, *116*, 140502.

(940) Dong, Y.; Zheng, Y.; Li, S.; Li, C.-C.; Chen, X.-D.; Guo, G.-C.; Sun, F.-W. Non-Markovianity-Assisted High-Fidelity Deutsch-Jozsa Algorithm in Diamond. *NPJ. Quantum Inf.* **2018**, *4*, 3.

(941) Deutsch, D.; Jozsa, R. Rapid Solution of Problems by Quantum Computation. *Proc. R. Soc. A: Math. Phys. Sci.* **1992**, 439, 553–558.

(942) Zhang, J.; Souza, A. M.; Brandao, F. D.; Suter, D. Protected Quantum Computing: Interleaving Gate Operations with Dynamical Decoupling Sequences. *Phys. Rev. Lett.* **2014**, *112*, No. 050502.

(943) Bennett, C. H.; Wiesner, S. J. Communication via One- and Two-Particle Operators on Einstein-Podolsky-Rosen States. *Phys. Rev. Lett.* **1992**, 69, 2881–2884.

(944) Laine, E.-M.; Breuer, H.-P.; Piilo, J.; Li, C.-F.; Guo, G.-C. Nonlocal Memory Effects in the Dynamics of Open Quantum Systems. *Phys. Rev. Lett.* **2012**, *108*, 210402.

(945) Ballester, D.; Romero, G.; García-Ripoll, J. J.; Deppe, F.; Solano, E. Quantum Simulation of the Ultrastrong-Coupling Dynamics in Circuit Quantum Electrodynamics. *Phys. Rev. X* **2012**, *2*, No. 021007.

(946) Lv, D.; An, S.; Liu, Ź.; Zhang, J.-N.; Pedernales, J. S.; Lamata, L.; Solano, E.; Kim, K. Quantum Simulation of the Quantum Rabi Model in a Trapped Ion. *Phys. Rev. X* **2018**, *8*, No. 021027.

(947) Macridin, A.; Spentzouris, P.; Amundson, J.; Harnik, R. Electron-Phonon Systems on a Universal Quantum Computer. *Phys. Rev. Lett.* **2018**, *121*, 110504.

(948) Turney, J. M.; Simmonett, A. C.; Parrish, R. M.; Hohenstein, E. G.; Evangelista, F. A.; Fermann, J. T.; Mintz, B. J.; Burns, L. A.; Wilke, J. J.; Abrams, M. L.; et al. Psi4: an Open-Source ab initio Electronic Structure Program. *Wiley Interdiscip. Rev. Comput. Mol. Sci.* **2012**, *2*, 556–565.

(949) Gómez Pueyo, A.; Marques, M. A. L.; Rubio, A.; Castro, A. Propagators for the Time-Dependent Kohn–Sham Equations: Multistep, Runge–Kutta, Exponential Runge–Kutta, and Commutator Free Magnus Methods. *J. Chem. Theory Comput.* **2018**, *14*, 3040–3052.

(950) McKay, D. C.; Alexander, T.; Bello, L.; Biercuk, M. J.; Bishop, L.; Chen, J.; Chow, J. M.; Córcoles, A. D.; Egger, D.; Filipp, S. et al. Qiskit Backend Specifications for OpenQASM and OpenPulse Experiments.2018; arXiv:1809.03452 [quant-ph]; https://arxiv.org/abs/1809.03452.